CORGENIX MEDICAL CORP/CO Form PREM14A September 24, 2014 Table of Contents

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

SCHEDULE 14A

Proxy Statement Pursuant to Section 14(a) of the Securities Exchange Act of 1934 (Amendment No.)

Filed by the Registrant X

Filed by a Party other than the Registrant O

Check the appropriate box:

x Preliminary Proxy Statement

o Confidential, for Use of the Commission Only (as permitted by Rule 14a-6(e)(2))

o Definitive Proxy Statement o Definitive Additional Materials o Soliciting Material under §240.14a-12

Corgenix Medical Corporation (Name of Registrant as Specified In Its Charter)

(Name of Person(s) Filing Proxy Statement, if other than the Registrant)

Payment of Filing Fee (Check the appropriate box):

No fee required.

x Fee computed on table below per Exchange Act Rules 14a-6(i)(1) and 0-11.

(1) Title of each class of securities to which transaction applies:

Common stock, par value \$0.001 per share, of Corgenix Medical

Corporation (the Common Stock)

(2) Aggregate number of securities to which transaction applies:

As of September 15, 2014, 54,228,834 shares of Common Stock,

3,325,000 shares of Common Stock is suable upon the exercise of options

with exercise price less than \$0.27 per share and warrants to buy

12,369,394 shares of Common Stock with exercise price less than \$0.27

per share

(3) Per unit price or other underlying value of transaction computed pursuant

to Exchange Act Rule 0-11 (set forth the amount on which the filing fee is

calculated and state how it was determined):

The maximum aggregate value was determined based upon the sum of: (A) 54,228,834 shares of Common Stock outstanding, multiplied by \$0.27 per share and (B) options to purchase 3,325,000 shares of Common Stock with exercise prices less than \$0.27 multiplied by \$0.14, which is the difference between \$0.27 and the weighted average exercise price of \$0.13 per share, and (C) warrants to buy 12,369,394 shares of Common Stock with exercise prices less than \$0.27 multiplied by \$0.09, which is the difference between \$0.27 and the weighted average exercise price of

\$0.18 per share

(4) Proposed maximum aggregate value of transaction:

\$16,220,531

(5) Total fee paid:

\$2,089.20, determined based upon multiplying .0001288 by the proposed maximum aggregate value of the transaction

o Fee paid previously with preliminary materials.

o

Check box if any part of the fee is offset as provided by Exchange Act Rule 0-11(a)(2) and identify the filing for which the offsetting fee was paid previously. Identify the previous filing by registration statement number, or the Form or Schedule and the date of its filing.

(1) Amount Previously Paid:

(2) Form, Schedule or Registration Statement No.:

(3) Filing Party:

(4) Date Filed:

Table of Contents

Corgenix Medical Corporation

11575 Main Street, Suite 400

Broomfield, CO 80020

(303) 457-4345

MERGER PROPOSED YOUR VOTE IS IMPORTANT	
Dear Shareholder:	
You are cordially invited to attend a special meeting of the shareholders of Corgenix Medical Corporation, a Nevada corporation, which will be held on , 2014, at , Mountain Time, .	эе
At the special meeting, we will ask you to consider and vote on a proposal to adopt an Agreement and Plan of Merger (the Merger Agreement that we entered into with Centennial Medical Holdings, Inc., whom we refer to as Buyer, and Centennial Integrated, Inc., a wholly owned subsidiary of Buyer whom we refer to as the Merger Sub, on August 27, 2014. Buyer and the Merger Sub are affiliates of Orgentec Diagnot (Orgentec), a specialty diagnostics company headquartered in Mainz, Germany. If shareholders representing at least a majority of the outstanding shares of our common stock, par value \$0.001 per share (Common Stock), entitled to vote thereon adopt the Merger Agreementhe merger is completed, we will become a wholly owned subsidiary of Buyer, and you will be entitled to receive \$0.27 in cash, without intereand less any applicable withholding taxes, for each share of our Common Stock that you own.	ostil nt an
After careful consideration, our board of directors, by the unanimous vote of all directors, approved the Merger Agreement and determined that the merger and the Merger Agreement are advisable and in the best interests of our company and our shareholders. Our board of directors unanimously recommends that you vote FOR the adoption of the Merger Agreement.	at
Your vote is very important. Adoption of the Merger Agreement requires the affirmative vote of the holders of a majority of the outstanding	

shares of our Common Stock entitled to vote thereon. Therefore, failure to vote will have the same effect as a vote against the adoption of the Merger Agreement.

In addition, the Securities and Exchange Commission has adopted rules that require us to seek a non-binding, advisory vote with respect to certain compensation arrangements for our named executive officers in connection with the merger. Our board of directors unanimously recommends that you vote FOR the merger-related compensation proposal described in the accompanying proxy statement.

The accompanying proxy statement provides a detailed description of the proposed merger, the Merger Agreement and related matters, and a copy of the Merger Agreement is included as Annex A to the proxy statement. We urge you to read all of these materials carefully.

Whether or not you are able to attend the special meeting in person, please submit your proxy via the internet address and telephone voting options available on the enclosed proxy card, or complete, sign and date the enclosed proxy card and return it in the provided envelope as soon as possible. If you have Internet access, we encourage you to record your vote via the Internet. This action will not limit your right to vote in person at the special meeting.

If you have any questions or need assistance voting your shares, please call our proxy solicitor, Georgeson Inc., at (877) 278-4751.

Thank you for your cooperation and your continued support of Corgenix Medical Corporation.

Very truly yours,

Douglass T. Simpson, President and Chief Executive Officer

This proxy statement is dated

CRIMINAL OFFENSE.

, 2014 and is first being mailed to shareholders on or about

NEITHER THE SECURITIES AND EXCHANGE COMMISSION NOR ANY STATE SECURITIES COMMISSION HAS APPROVED OR DISAPPROVED THE MERGER, PASSED UPON THE MERITS OR FAIRNESS OF THE MERGER AGREEMENT OR THE TRANSACTIONS CONTEMPLATED THEREBY, INCLUDING THE PROPOSED MERGER, OR PASSED UPON THE ADEQUACY OR ACCURACY OF THE INFORMATION CONTAINED IN THIS DOCUMENT. ANY REPRESENTATION TO THE CONTRARY IS A

, 2014.

Table of Contents

NOTICE OF SPECIAL MEETING OF SHAREHOLDERS

OF CORGENIX MEDICAL CORPORATION

, 2014

To be held on

TO THE SHAREHOLDERS OF					
CORGENIX MEDICAL CORPORATION:					
Notice is hereby given that a special meeting (the special Meeting) of shareholders of Corgenix Medical Corporation (Corgenix) will be held a the shareholders of Corgenix Medical Corporation (Corgenix) will be held a Mountain Time to consider and act upon the following matters:					
1. To adopt the Agreement and Plan of Merger, dated as of August 27, 2014, by and among Corgenix, Centennial Medical Holdings, Inc. (Buyer) and Centennial Integrated, Inc., a wholly owned subsidiary of Buyer (the Merger Sub), as such may be amended from time to time (the Merger Agreement), pursuant to which each holder of shares of Corgenix common stock, par value \$0.001 per share (the Common Stock), will be entitled to receive \$0.27 in cash, without interest and less any applicable withholding taxes, for each share of our Common Stock held by such holder.					
2. To consider and vote on a proposal to approve, by a non-binding, advisory vote, certain compensation arrangements for our named executive officers in connection with the merger, which we refer to as the golden parachute proposal.					
3. To approve a proposal to adjourn or postpone the special meeting, if necessary, to solicit additional proxies in favor of adoption of the Merger Agreement, which we refer to as the adjournment proposal.					
4. To transact such other business as may properly come before the special meeting or any adjournment or postponement thereof, including the consideration of any procedural matters incident to the conduct of the special meeting.					
A copy of the Merger Agreement is attached as Annex A to the accompanying proxy statement.					

Corgenix s Board of Directors has fixed the close of business on September 15, 2014 as the record date. Only holders of record of our Common Stock as of the close of business on September 15, 2014 are entitled to receive notice of, and to vote at, the Special Meeting and any adjournment or postponement of the Special Meeting. Adoption of the Merger Agreement requires the affirmative vote of the holders of a majority of the outstanding shares of our Common Stock entitled to vote thereon, and the approval of each of the golden parachute proposal and the adjournment proposal requires the affirmative vote of the holders of a majority of the shares present in person or by proxy and entitled to vote at the Special Meeting. The list of shareholders entitled to vote at the special meeting will be available for inspection at our principal executive offices at 11575 Main Street, Suite 400, Broomfield, Colorado, 80020 during ordinary business hours at least 10 days before the Special Meeting.

WHETHER OR NOT YOU ARE ABLE TO ATTEND THE SPECIAL MEETING IN PERSON, PLEASE SUBMIT YOUR PROXY VIA THE INTERNET ADDRESS OR BY THE TELEPHONE NUMBER AVAILABLE ON THE ENCLOSED PROXY CARD, OR COMPLETE, SIGN AND DATE THE ENCLOSED PROXY CARD AND RETURN IT IN THE ENVELOPE PROVIDED AS SOON AS POSSIBLE. IF YOU HAVE INTERNET ACCESS, WE ENCOURAGE YOU TO RECORD YOUR VOTE VIA THE INTERNET. THIS ACTION WILL NOT LIMIT YOUR RIGHT TO VOTE IN PERSON AT THE SPECIAL MEETING. IF YOU FAIL TO VOTE BY PROXY OR IN PERSON, IT WILL HAVE THE SAME EFFECT AS A VOTE AGAINST THE ADOPTION OF THE MERGER AGREEMENT. IF YOU RETURN A PROPERLY SIGNED PROXY CARD BUT DO NOT INDICATE HOW YOU WANT TO VOTE, YOUR PROXY WILL BE COUNTED AS A VOTE FOR ADOPTION OF THE MERGER AGREEMENT, FOR APPROVAL OF THE GOLDEN PARACHUTE PROPOSAL AND FOR THE ADJOURNMENT PROPOSAL.

Table of Contents

TABLE OF CONTENTS

	Page
SUMMARY TERM SHEET	1
<u>Parties to the Merger</u>	1
<u>The Merger</u>	2
The Special Meeting of Shareholders	2
Record Date; Stock Entitled to Vote	2
Vote Required for Approval	2
Our Board s Recommendation	3
Opinion of Corgenix s Financial Advisor	3
Conditions to the Merger	3
No Solicitation	5
<u>Termination of the Merger Agreement</u>	5
<u>Termination Fees</u>	6
<u>Expenses</u>	6
<u>Voting Agreement</u>	6
<u>Dissenter s Righ</u> ts	6
Material United State Federal Income Tax Consequences	7
<u>Treatment of Options and Warrants</u>	7
Interests of Our Directors and Executive Officers and Our Largest Shareholder in the Merger	7
Common Stock Ownership of Our Directors and Executive Officers	8
QUESTIONS AND ANSWERS ABOUT THE SPECIAL MEETING AND THE MERGER	9
CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS	13
THE SPECIAL MEETING OF THE SHAREHOLDERS	14
Date, Time and Place of the Special Meeting	14
Purpose of the Special Meeting	14
Our Board s Recommendation	14
Record Date; Stock Entitled to Vote	14
Quorum	14
Vote Required for Approval	15
Voting	15
Revocability of Proxies	15
Solicitation of Proxies	16
Proposal to Approve Adjournment of the Special Meeting	16
Questions and Additional Information	16
Availability of Documents	16
PROPOSAL ONE: THE MERGER	17
Parties to the Merger	17
Background of the Merger	18
Reasons for the Merger and Recommendation of our Board of Directors	27
Opinion of Corgenix s Financial Advisor	31
Certain Corgenix Forecasts	39
Financing of the Merger	40
Deregistration of Corgenix Common Stock	40
Interests of our Directors and Executive Officers in the Merger	40
Interests of our Largest Shareholder in the Merger	44
DECLII ATODY MATTERS	**
REGULATORY MATTERS	44

i

Table of Contents

MATERIAL UNITED STATED FEDERAL INCOME TAX CONSEQUENCES	45
THE MERGER AGREEMENT	49
Explanatory Note Regarding the Merger Agreement	49
The Merger	49
The Merger Consideration	50
Payment Procedures	50
Treatment of Options and Warrants	51
Representations and Warranties	52
Conduct of Our Business Pending the Merger	54
Preparation of Proxy Statement; Shareholders Meeting	56
<u>Public Statements</u>	57
Standard of Efforts; Required Approvals	57
Notification of Certain Matters	57
Confidentiality	57
No Solicitation	57
Indemnification and Insurance	59
Section 16 Matters	60
Financing Control of the Control of	60
Conditions to the Merger	60
Termination of the Merger Agreement	62
Termination Fees Even and the second	63 63
Expenses Remedies	63
Governing Law/Jurisdiction	63
Voting Agreement	63
voting Agreement	03
<u>DISSENTER S RIGHT</u> S	64
MARKET PRICE INFORMATION	65
SECURITY OWNERSHIP OF MANAGEMENT AND CERTAIN BENEFICIAL OWNERS	66
PROPOSAL TWO COMPENSATION ARRANGEMENTS FOR OUR NAMED EXECUTIVE OFFICERS IN CONNECTION WITH THE MERGER	68
PROPOSAL THREE ADJOURNMENT OF THE SPECIAL MEETING	69
FUTURE SHAREHOLDER PROPOSALS	70
WHERE YOU CAN FIND MORE INFORMATION	70
ANNEX A AGREEMENT AND PLAN OF MERGER BY AND AMONG CORGENIX MEDICAL CORPORATION, CENTENNIAL MEDICAL HOLDINGS, INC., AND CENTENNIAL INTEGRATED, INC. DATED AUGUST 27, 2014	A-i
ANNEX B VOTING AGREEMENT BY AND BETWEEN CENTENNIAL MEDICAL HOLDINGS, INC. AND DOUGLASS SIMPSON, WILLIAM CRITCHFIELD, ANN L. STEINBARGER, STEPHEN P. GOUZE AND ROBERT TUTAG	B-1
ANNEX C INVERNESS FAIRNESS OPINION DATED AUGUST 26, 2014	C-1
ANNEX D DISSENTER S RIGHTS STATUTE	D-1
ii	

Table of Contents

SUMMARY TERM SHEET

This summary highlights selected information from this proxy statement and may not contain all of the information that is important to you. Accordingly, we urge you to read carefully this entire proxy statement and the annexes to this proxy statement. We have included page references parenthetically to direct you to a more complete description of the topics in this summary.

In this proxy statement, the terms we, us, our, our company and Corgenix refer to Corgenix Medical Corporation, the term Buyer refers to Centennial Medical Holdings, Inc., the term Merger Sub refers to Centennial Integrated, Inc., a wholly owned subsidiary of Buyer, the term Orgentec refers to Orgentec Diagnostika, an affiliate of Buyer and the Merger Sub, and the term Merger Agreement refers to the Agreement and Plan of Merger, dated as of August 27, 2014, by and among Corgenix, Buyer and the Merger Sub, as it may be amended from time to time.

Parties to the Merger

Corgenix Medical Corporation 11575 Main Street, Suite 400

Broomfield, Colorado, 80020 Telephone: (303) 457-4345

We develop and manufacture specialized diagnostic kits for immunology disorders, vascular diseases (including the world s only non-blood-based test for aspirin effect), bone and joint disorders and a line of unique detection products for viral hemorrhagic disease. Our diagnostic products are commercialized for use in clinical laboratories throughout the world. We currently sell over 50 diagnostic products through a global distribution network and have significant experience in product submissions to the FDA and other worldwide regulatory authorities. Additionally, we contract, develop and manufacture products for key medical and life science companies in state-of-the-art facilities in Colorado.

Orgentec Diagnostika

333 W. Wacker, Suite 2800

Chicago, IL 60606

Telephone: (312) 506-2900

Orgentec is a global specialty diagnostics company. The company offers a portfolio of more than 300 tests, primarily enzyme-linked immunosorbent assays (ELISA), that help diagnose rheumatology, thrombosis and gastroenterology disorders, as well as infectious diseases. It also offers a fully automated instrument and associated test kits that enable laboratories to complete multiple assays and deliver faster results

than traditional ELISA tests. Orgentec has an established position in Europe and a growing presence in Asia, Latin America and the Middle East. It is headquartered in Mainz, Germany. Orgentec is an affiliate of Water Street Healthcare Partners, LLC (Water Street), a strategic investor focused exclusively on the healthcare industry.

Centennial Medical Holdings, Inc.
c/o Water Street Healthcare Partners, LLC
333 West Wacker Drive, Suite 2800
Chicago, IL 60606
Telephone: (312) 506-2900
Buyer is an affiliate of Orgentec and was formed for the purpose of acquiring and holding us. Buyer has not carried on any activities to date, except for activities incidental to its formation and activities undertaken in connection with the transactions contemplated by the Merger Agreement. Upon consummation of the proposed merger, we will become a wholly owned subsidiary of Buyer.
Centennial Integrated, Inc.
c/o Water Street Healthcare Partners, LLC
333 West Wacker Drive, Suite 2800

1

Table of Contents
Chicago, IL 60606
Telephone: (312) 506-2900
The Merger Sub is a direct, wholly owned subsidiary of Buyer and was formed solely for the purpose of facilitating Buyer s acquisition of us. The Merger Sub has not carried on any activities to date, except for activities incidental to its formation and activities undertaken in connection with the transactions contemplated by the Merger Agreement. Upon consummation of the proposed merger, the Merger Sub will merge with and into us and will cease to exist.
The Merger (page 49)
Upon the terms and subject to the conditions of the Merger Agreement, the Merger Sub will be merged with and into us, and each holder of shares of our common stock, par value \$0.001 per share (Common Stock), will be entitled to receive \$0.27 in cash, without interest and less any applicable withholding taxes, for each share of Common Stock held by such holder immediately prior to the merger (unless such holder has not voted in favor of the Merger Agreement and has properly exercised his or her statutory dissenter s rights with respect to the merger) (such consideration, the Merger Consideration). Following the merger, we will cease to be a publicly traded company and will instead become a wholly owned subsidiary of Buyer. You will not own any shares of the surviving corporation of the merger. The Merger Agreement is attached as Annex A to this proxy statement. Please read it carefully.
The Special Meeting of Shareholders (page 14)
The special meeting will be held on , 2014, at , Mountain Time, at the . At the special meeting, you will be asked to vote upon a proposal to adopt the Merger Agreement that we have entered into with Buyer and the Merger Sub, which we refer to as the merger proposal. You will also be asked to vote upon a non-binding proposal to approve the compensation that may become payable to certain of our named executive officers in connection with the completion of the merger, which we refer to as the golden parachute proposal, and a proposal to adjourn or postpone the special meeting, if necessary, to solicit additional proxies in favor of adoption of the Merger Agreement, which we refer to as the adjournment proposal. You may also be asked to vote upon such other matters as may properly come before the special meeting or any adjournment or postponement thereof.
Record Date; Stock Entitled to Vote (page 14)
Our board of directors has fixed the close of business on September 15, 2014, as the record date for determining shareholders entitled to receive notice of and to vote at the special meeting. On the record date, we had 54,228,834 outstanding shares of Common Stock held by approximately 80 shareholders of record. We have no other class of voting securities outstanding.

Shareholders of record on the record date will be entitled to one vote per share of our Common Stock on any matter that may properly come before the special meeting and any adjournment or postponement of that meeting.

Vote Required for Approval (page 15)

Pursuant to the requirements of Section 92A.120 of the Nevada Revised Statutes (NRS), the approval of the merger proposal requires the affirmative vote of the holders of a majority of the outstanding shares of our Common Stock entitled to vote thereon. Abstentions and broker non-votes, as described below, will have the same effect as a vote AGAINST the approval of the merger proposal.

The affirmative vote of the holders of a majority of the shares of our Common Stock present in person or by proxy and entitled to vote at the special meeting will be required to approve each of the golden parachute proposal and the adjournment proposal. Abstentions will have the same effect as a vote AGAINST the approval of these proposals. Broker non-votes will have no effect on the approval of these proposals.

Table of Contents

Brokers or other nominees who hold shares of our Common Stock in street name for customers who are the beneficial owners of such shares may not, in the absence of specific instructions from those customers, give a proxy to vote their customers shares with respect to the approval of the merger proposal, the golden parachute proposal or the adjournment proposal. Shares held by brokers or other nominees that are not voted for lack of instructions from their customers are referred to as broker non-votes. Broker non-votes will not be counted as votes cast or shares voting and will have the same effect as votes AGAINST the merger proposal. Assuming a quorum is present at the special meeting, broker non-votes will have no effect on the approval of the golden parachute proposal or the adjournment proposal.

On the record date, our directors and executive officers owned and were entitled to vote approximately 5.5% of our shares of Common Stock outstanding on that date, and such persons have entered into a voting agreement obligating them to vote to approve the Merger Agreement at our special meeting.

Our Board s Recommendation (page 14)

Our board of directors unanimously (i) determined that the merger and the Merger Agreement are advisable and in the best interests of our company and our shareholders, (ii) approved the Merger Agreement, (iii) resolved to recommend that shareholders vote in favor of the adoption of the Merger Agreement, and (iv) directed that such matter be submitted for the consideration of our shareholders at the special meeting. Accordingly, our board of directors unanimously recommends that our shareholders vote FOR the merger proposal, FOR the golden parachute proposal and FOR the adjournment proposal.

For the factors considered by our board of directors in reaching its decision to approve the Merger Agreement, see The Merger Reasons for the Merger and Recommendation of our Board of Directors beginning on page 27 of this proxy statement.

Opinion of Corgenix s Financial Advisor (page 31 and Annex C)

In connection with the merger, our financial advisor, Inverness Advisors, a division of KEMA Partners LLC, or Inverness, rendered to our board of directors its written opinion, that as of August 26, 2014, and based upon and subject to the various assumptions, procedures, factors, qualifications and limitations set forth in the written opinion, the consideration to be received by holders of shares of our Common Stock pursuant to the Merger Agreement was fair, from a financial point of view, to such holders. The full text of the written opinion of Inverness, dated as of August 26, 2014, is attached as Annex C to this proxy statement and is incorporated by reference into this proxy statement in its entirety. The opinion sets forth, among other things, the assumptions made, procedures followed, matters considered and limitations on the scope of the review undertaken by Inverness in rendering its opinion. The summary of the opinion of Inverness in this proxy statement is qualified in its entirety by reference to the full text of the opinion. Inverness s opinion is directed to our board of directors and addresses only the fairness, from a financial point of view, of the consideration to be received by holders of shares of our Common Stock pursuant to the Merger Agreement as of the date of the opinion. Inverness s opinion does not address any other aspects of the merger and does not constitute a recommendation to any holder of our Common Stock as to how to vote at any shareholders meeting held in connection with the merger or whether to take any other action with respect to the merger.

Conditions to the Merger (page 60)

Conditions to Each Party s Obligations. The respective obligations of each party to effect the merger are subject to the satisfaction or, to the extent permitted by law, waiver of each of the following conditions:

- the adoption of the Merger Agreement by the holders of a majority of the shares of our Common Stock outstanding on the record date for the special meeting on or before December 31, 2014;
- none of the parties are subject to any order or injunction that prohibits consummation of the transactions contemplated by the Merger Agreement;

3

Table of Contents

• all consents, authorizations, order and approvals with any governmental entity required will have been obtained; and
• no action will be pending or threatened by or before any court in which an unfavorable judgment would prevent any of the transactions contemplated by the Merger Agreement or cause any such transaction to be declared unlawful or rescinded or that would reasonal cause a company material adverse effect.
Conditions to Buyer s and the Merger Sub s Obligations. The obligations of Buyer and the Merger Sub to consummate the merger are subject to the satisfaction or, to the extent permitted by law, waiver of further conditions, including:
• we must have performed in all material respects all obligations required to be performed by us under the Merger Agreement at or prior to the closing date of the merger;
• our representations and warranties, except those related to authorization, validity and effect of agreements and our capitalization, must be true and correct as of the date of the Merger Agreement and as of the closing date of the merger (provided that, for this purpose, our representations and warranties will be deemed not qualified by any references therein to materiality), except to the extent that such representation or warranty relates to an earlier time (in which case as of such earlier time), provided that this condition will be deemed satisfied even if not true and correct unless the failure would have or would reasonably be likely to have a company material adverse effect;
• our representations and warranties relating to authorization, validity and effect of agreements and capitalization must be true and correct in all respects as of the date of the Merger Agreement and as of the closing date of the merger;
• we will not be in default of any obligation under any material contract, unless such default would not reasonably be expected to ha a company material adverse effect;
Buyer must have received a certificate as to our incorporation and good standing and the good standing of Corgenix, Inc.;
• there will have been no change in our financial condition or results of operations from that reflected in the balance sheet so as to result, or be reasonably likely to result, in a company material adverse effect;
• Buyer must have received certain consents and agreements from third parties and release of certain liens;

	no action will be pending where an unfavorable injunction, judgment, order or ruling would prevent consummation of the merger or s contemplated by the Merger Agreement;
•	dissenting shares will not constitute more than 10% of the shares issued and outstanding; and
• first, secon	Buyer must have received a certificate signed by one of our senior executive officers as to the satisfaction of the conditions in the d, third and sixth bullets above.
Conditions including:	to Our Obligations. Our obligation to consummate the merger is subject to the satisfaction or waiver of further conditions,
	Buyer and Merger Sub must have performed in all material respects all obligations required to be performed by us under the Merger at or prior to the closing date of the merger;
	the representations and warranties of Buyer and Merger Sub, except those related to authorization, validity and effect of agreements, the and correct as of the date of the Merger Agreement and as of the closing date of the merger (provided that, for this purpose, our ions and warranties will be deemed not qualified by any references therein to materiality), except to the extent that such ion
	4

Table of Contents

or warranty relates to an earlier time (in which case as of such earlier time), provided that this condition will be deemed satisfied even if not true and correct unless the failure would have or would reasonably be likely to have a Buyer material adverse effect;

- the representations and warranties of Buyer and Merger Sub relating to authorization, validity and effect of agreements must be true and correct in all respects as of the date of the Merger Agreement and as of the closing date of the merger;
- we will have received a certificate from the Secretary of State of Delaware with respect to the incorporation and good standing of Buyer and certificate from the Secretary of State of Nevada with respect to the incorporation and good standing of Merger Sub;
- we will have received all consents and approvals from third parties required in connection with the Merger Agreement and consummation of the transactions contemplated by it;
- we will have received certificates signed by the President or Vice President and Secretary of each of Buyer and Merger Sub, attaching a copy of the resolutions of the board of directors of each approving the Merger Agreement and the consummation of the transactions; and
- we must have received a certificate signed by one of Buyer s and Merger Sub s senior executive officers as to the satisfaction of the conditions in the first, second and third bullets above.

No Solicitation (page 57)

Immediately upon signing of the Merger Agreement, we agreed not to, and agreed not to authorize or permit any of our officers, directors or employees, or authorize any of our financial advisors, attorneys or other advisors or representatives to (i) solicit, initiate, or knowingly encourage or induce the submission or announcement of any takeover proposal, (ii) participate or engage in any discussions or negotiations with any person or group other than Buyer, the Merger Sub regarding any takeover proposal, or furnish any non-public information relating to us to any third party that we believe would be used for purposes of making or furthering any takeover proposal, (iii) amend or grant any waiver, amendment or release under any standstill or similar agreement with a third party; or (iv) enter into any agreement in principle, letter of intent, term sheet, acquisition agreement, merger agreement, option agreement, joint venture agreement, partnership agreement or other similar agreement relating to a takeover proposal.

Notwithstanding the restrictions described above, at any time before the adoption of the Merger Agreement by our shareholders, if we receive an unsolicited written takeover proposal from a third party that did not result from a breach of the provisions of the Merger Agreement regarding restrictions on solicitations, we may provide or furnish information to the third party making the takeover proposal pursuant to an executed confidentiality agreement so long as our board of directors determines in good faith, after consultation with our financial advisor and outside counsel, that the takeover proposal is, or could reasonably be expected to result in or lead to, a superior proposal or take any action that any court of competent jurisdiction order us to take.

We must promptly notify Buyer after receipt of any takeover proposal, identify the third party making the proposal and provide a summary of the material terms and conditions of such proposal.

Termination of the Merger Agreement (page 62)

The parties may terminate on mutual written consent. Any party may also terminate if the merger is not consummated by December 31, 2014, a governmental entity takes action that would make the merger illegal or prevents consummation, or the shareholders fail to approve the merger. Buyer and Merger Sub may terminate if our board of directors effects a change in recommendation, we breach or fail to perform in accordance with the Merger Agreement, and such failure is not cured, or we enter into an acquisition agreement in accordance with the terms of the Merger Agreement. We may terminate if we enter into an acquisition agreement with a third party in accordance

Т	ab	le	of	Cor	itents

with the terms of the Merger Agreement,	or Buyer or Merger	Sub breaches or fails	to perform in	accordance v	with the Merger	Agreement, and
such failure is not cured.						

Termination Fees (page 63)

The Merger Agreement provides that we would pay Buyer a termination fee of \$324,000 plus out of pocket expenses up to \$400,000, if:

- We or Buyer terminates the Merger Agreement based on a superior proposal termination;
- Buyer terminates the agreement for our breach of the Merger Agreement; or
- Buyer terminates the Merger Agreement on the basis that our board of directors has effected a change in recommendation.

Buyer has agreed to pay us a termination fee of \$648,000 plus out of pocket expenses up to \$400,000, if we terminate the Merger Agreement based on a breach of the Merger Agreement by the Buyer or Merger Sub.

We, Buyer and the Merger Sub are also entitled to seek and obtain an injunction, specific performance and other equitable relief to prevent breaches of the Merger Agreement to enforce specifically the terms and provisions of the Merger Agreement. The election to pursue an injunction, specific performance or other equitable relief will not restrict, impair or otherwise limit the parties from seeking to terminate the Merger Agreement and collect the termination fee or expenses described above.

Expenses (page 63)

Except to the extent described herein regarding Termination Fees, we and Buyer are required to pay our own expenses in connection with the Merger Agreement and consummation of the transactions contemplated thereby, whether or not the merger is consummated.

Voting Agreement (page 63 and Annex B)

Simultaneously with the execution of the Merger Agreement, all of our executive officers and directors who are also shareholders, holding approximately 5.5% of shares of our Common Stock issued and outstanding (not including options held by any executive officer or director or warrants held by certain other shareholders), each entered into a voting agreement with Buyer pursuant to which each executive officer and director agreed that he or she will vote his or her shares of Common Stock (i) in favor of approval and adoption of the Merger Agreement and (ii) against certain other acquisition proposals. The voting agreement terminates on the earlier of (a) the effective date of the Merger, (b) the termination of the Merger Agreement and (c) the date our board of directors effects a change in board recommendation pursuant to the Merger Agreement.

Dissenter s Rights (page 64 and Annex D)

Holders of our Common Stock are entitled to assert dissenter s rights under Nevada Revised Statutes 92A.300 to 92A.500, inclusive. In the event of certain extraordinary corporate transactions, such as a merger, shareholders may exercise dissenter s rights by demanding payment for their shares at a price equal to its fair value immediately before the corporate action. If we and the shareholder cannot agree on a fair value, we must then commence an appraisal process in state court to determine the fair value, which may be more than, equal to, or less than the Merger Consideration. To exercise dissenter s rights, a holder of our Common Stock must deliver to us a notice of the holder s intent to demand payment for his or her shares before the shareholder vote on the Merger Agreement is taken at the special meeting, must not vote any shares in favor of the proposal to adopt the Merger Agreement, and must strictly comply with all of the procedures required by Nevada law.

6

Table of Contents

A copy of the full text of NRS 92A.300 to 92A.500 is included as Annex D to this proxy statement. Failure to follow the procedures set forth in NRS 92A.300 to 92A.500 will result in the forfeiture of dissenter s rights.

Material United States Federal Income Tax Consequences (page 45)

If the merger is completed, the exchange of Common Stock by our shareholders for the Merger Consideration will generally be treated as a taxable transaction for United States federal income tax purposes under the Internal Revenue Code of 1986, as amended. Because of the complexities of the tax laws, we advise you to consult your personal tax advisors concerning the applicable United States federal, state, local, foreign and other tax consequences of the merger to you.

Treatment of Options and Warrants (page 51)

We issued options to purchase shares of our Common Stock under our 2011 Incentive Compensation Plan and 2007 Incentive Compensation Plan (which, together, we refer to as the stock plans). Our board of directors has adopted resolutions providing that each stock option that has not been exercised and has not expired or terminated and remains outstanding immediately prior to the effective time of the merger will be canceled, and the holder of any such option will be entitled to receive an amount in cash, without interest, equal to the product of (i) the total number of shares of our Common Stock issuable upon exercise of such option and (ii) the excess, if any, of the Merger Consideration over the per share exercise price under such option, subject to any required withholding taxes. No holder of an option that has an exercise price per share greater than \$0.27 is entitled to any payment with respect to such canceled option before or after the effective time of the merger. See Material United States Federal Income Tax Consequences beginning on page 45 of this proxy statement for a discussion of the tax consequences of the merger to option holders.

Our board of directors also adopted resolutions providing that each warrant to purchase shares of Common Stock that is outstanding immediately prior to the effective time of the merger will be canceled, and the holder of any such warrant will be entitled to receive an amount in cash, without interest, equal to the product of (i) the total number of shares of our Common Stock previously subject to such warrant and (ii) the excess, if any, of the Merger Consideration over the per share exercise price previously subject to such warrant, subject to any required withholding taxes. No holder of a warrant that has an exercise price per share that is greater than \$0.27 is entitled to any payment with respect to such canceled warrant before or after the effective time of the merger. See Material United States Federal Income Tax Consequences beginning on page 45 of this proxy statement for a discussion of the tax consequences of the merger to warrant holders.

Interests of Our Directors and Executive Officers and Our Largest Shareholder in the Merger (page 40 and 44)

In considering the recommendation of our board of directors with respect to the Merger Agreement, holders of shares of our Common Stock should be aware that our directors and executive officers have interests in the merger that may be different from, or in addition to, those of our shareholders generally. These interests may create potential conflicts of interest. Our board of directors was aware that these interests existed when it approved the Merger Agreement. These interests include:

accelerated vesting, as a result of the merger, of certain equity awards held by our executive officers;

severance arrangements for our President and Chief Executive Officer, Senior Vice President Operations and Finance and Chief Financial Officer and Senior Vice President Sales and Marketing pursuant to the terms of their management agreements; and
 indemnification of our directors and executive officers by the surviving corporation following the merger.
 These arrangements are further described under The Merger Interests of Our Directors and Executive Officers in the Merger beginning on page 40 of this proxy statement. In addition, our largest shareholder may have interests in the merger that are in addition to, or different from, the interests of our shareholders generally. These

7

Table of Contents

arrangements are further described under The Merger Interests of our Large Shareholder beginning on page 44 of this proxy statement.

Common Stock Ownership of Our Directors and Executive Officers (page 66)

As of September 15, 2014, our current directors and executive officers beneficially owned in the aggregate 4,912,183 shares of our Common Stock or approximately 8.87% of our total outstanding shares. The share ownership of our directors and executive officers is further described under Security Ownership of Management and Certain Beneficial Owners beginning on page 66 of this proxy statement.

8

Table of Contents

QUESTIONS AND ANSWERS ABOUT THE SPECIAL MEETING AND THE MERGER

The following questions and answers briefly address some commonly asked questions about the special meeting of shareholders and the merger. These questions and answers may not address all questions that may be important to you as a shareholder. You should carefully read this entire proxy statement, including each of the annexes.

this entire proxy statement, including each of the annexes.			
Q.	What is the proposed transaction?		
	We, Buyer and the Merger Sub have entered into a definitive agreement pursuant to which, and subject to the terms and user will acquire us by merging the Merger Sub with and into us, with our company as the surviving corporation of the se to be a publicly traded company and will become a wholly owned subsidiary of Buyer.		
Q. receive it?	If the merger is completed, what will I be entitled to receive for my shares of Common Stock, and when will I		
A. Common Stock that y Nevada law.	You will be entitled to receive \$0.27 in cash, without interest and less any applicable withholding taxes, for each share of you own. This does not apply to shares held by our shareholders, if any, who properly perfect dissenter s rights under		
certificates or transfer shareholder once that	es, Buyer will arrange for a letter of transmittal to be sent to each of our shareholders for use in surrendering stock ring uncertificated shares in exchange for the Merger Consideration. The Merger Consideration will be paid to a shareholder submits a properly completed letter of transmittal, his, her or its stock certificates and any other required the case of book-entry transfer of uncertificated shares of our Common Stock, an agent s message.		

Q. When is the merger expected to be completed?

- A. We expect the merger to be completed during the fourth quarter of 2014. However, the merger is subject to various closing conditions, including shareholder approval, and it is possible that the failure to timely meet these closing conditions or other factors outside our control could require us to complete the merger at a later time or not at all.
- Q. What will happen to my shares of Common Stock after the merger?

	Following the effectiveness of the merger, your shares of Common Stock will represent only the right to receive the on, and trading in our Common Stock on the OTC Bulletin Board® will cease. Price quotations for our Common Stock will le, and we will cease filing periodic reports under the Securities Exchange Act of 1934, as amended, or the Exchange Act.
Q.	What will I be asked to vote upon at the special meeting?
A.	You will be asked to vote on:
Merger Sub will be rand	the adoption of the Merger Agreement that we have entered into with Buyer and the Merger Sub, pursuant to which the merged with and into us and we will become a wholly owned subsidiary of Buyer, which we refer to as the merger proposal
• in connection with the	the approval, by a non-binding, advisory vote, of certain compensation arrangements for our named executive officers ne merger, which we refer to as the golden parachute proposal; and
	9

Table of Contents

Q.	What shareholder approval is required for the adjournment proposal?
contractual in nature advisory vote, if the	Approval of the golden parachute proposal is not a condition to the completion of the merger. The vote with respect to the posal is an advisory vote and will not be binding on us or Buyer. Further, the underlying plans and arrangements are and not, by their terms, subject to shareholder approval. Accordingly, regardless of the outcome of the non-binding Merger Agreement is adopted by the shareholders and the merger is completed, our named executive officers will receive a compensation to which they may be entitled.
Q:	What will happen if shareholders do not approve the golden parachute proposal at the special meeting?
	The holders of a majority of the shares of our Common Stock present in person or represented by proxy and entitled to eeting will be required to approve the non-binding advisory golden parachute proposal. Abstentions will have the same AINST the golden parachute proposal. Broker non-votes will have no effect on the approval of the golden parachute
Q.	What shareholder approval is required for the golden parachute proposal?
Stock at the close of on the record date, the	The holders of a majority of the outstanding shares of our Common Stock entitled to vote on September 15, 2014, which the special meeting of shareholders, must vote in favor of the merger proposal. Only holders of record of our Common business on the record date are entitled to receive notice of and to vote at the special meeting. As of the close of business are were 54,228,834 shares of our Common Stock outstanding, held by approximately 80 shareholders of record, and a special meeting. Abstentions and broker non-votes (as described below) will have the same effect as a vote AGAINST the ger Agreement.
Q.	What shareholder approval is required for the merger proposal?
	are able to attend the special meeting in person, please submit your proxy via the Internet address or by the telephone the enclosed proxy card, or complete, sign and date the enclosed proxy card and return it in the envelope provided as soon
• favor of adoption of	the approval of the adjournment or postponement, if necessary, of the special meeting to solicit additional proxies in the Merger Agreement, which we refer to as the adjournment proposal.

	The holders of a majority of the shares of our Common Stock present in person or represented by proxy and entitled to eeting will be required to approve the adjournment proposal. Abstentions will have the same effect as a vote AGAINST the l. Broker non-votes will have no effect on the approval of the adjournment proposal.		
Q.	Who is entitled to vote at the special meeting?		
A. the special meeting.	Holders of record of shares of our Common Stock as of the close of business on September 15, 2014 are entitled to vote at Such holders are entitled to one vote per share of our Common Stock held.		
Q.	Why is our board of directors recommending the merger?		
A. Our board of directors engaged in a lengthy process, with the assistance of Inverness and senior management, to explore the strategic alternatives available to us, including a sale of our company. After careful consideration and consultation with our senior management, legal counsel and financial advisor, our board of directors, by the unanimous vote of all directors, approved the Merger Agreement and determined that the merger and the Merger Agreement are advisable and in the best interests of our company and our shareholders. Our board of directors unanimously recommends that you adopt the Merger Agreement. For a more detailed explanation of the factors that our board of directors considered in determining whether to recommend the merger, see The Merger Reasons for the Merger and Recommendation of our Board of Directors beginning on page 27 of this proxy statement.			
Q.	What should I do now?		
	After carefully reading and considering the information contained in this proxy statement, please vote in one of the property, regardless of whether you plan to attend the special meeting: (i) by completing your proxy through the Internet at the accompanying proxy card, (ii) by		
	10		

7D 1	1			_			
Tal	าเ	e.	Ot	()	Ωn	ter	1fs

completing your proxy using the toll-free telephone number li	sted on the proxy card, or (iii) by completing, signing and dating the proxy card
and returning it in the enclosed postage-prepaid envelope. Yo	ou can also attend the special meeting and vote in person. Do <u>NOT</u> enclose or
return your stock certificate(s) with your proxy card.	

Q.	If my shares are held in street name by my broker, will my broker vote my shares for me?
shares with respect to procedures provided I specific instructions f the same effect as a v	No. Brokers or other nominees who hold shares of our Common Stock in street name for customers who are the hose shares may not, in the absence of specific instructions from those customers, give a proxy to vote their customers the approval of the merger proposal, the golden parachute proposal or the adjournment proposal. You should follow the by your broker regarding the voting of your shares. Shares of our Common Stock that are not voted by brokers for lack of from their customers are referred to as broker non-votes and will not be counted as votes cast or shares voting and will have one AGAINST the approval of the merger proposal. Assuming a quorum is present at the special meeting, broker to effect on the approval of the golden parachute proposal or the adjournment proposal.
Q.	What if I do not vote?
A. vote will have no effe obtain a quorum at th	If you fail to vote in person or by proxy, it will have the same effect as a vote AGAINST the merger proposal. Failure to ect on the golden parachute proposal or the adjournment proposal. Failure to vote could adversely affect our ability to e special meeting.
	ly signed proxy card but do not indicate how you want to vote, your proxy will be counted as a vote FOR the merger golden parachute proposal and FOR the adjournment proposal.
determining the prese	operly signed proxy and affirmatively elect to abstain from voting, your proxy will be counted as present for the purpose of nce of a quorum but will have the same effect as a vote AGAINST the merger proposal, AGAINST the golden parachute NST the adjournment proposal.

Q. When should I cast my vote?

- A. You should complete your proxy card through the Internet or by telephone or mail in your proxy card as soon as possible, but in any event before , 2014 so that your shares will be voted at the special meeting.
- Q. May I change my vote after I have mailed my signed proxy card or voted via the Internet or by telephone?

A.	Yes. You may change your vote and revoke your proxy at any time before the polls close at the special meeting. You can
do this in one of three	ways. First, you can send a written, dated notice to our Secretary stating that you would like to revoke your proxy.
Second, you can comp	elete, date and submit a new proxy card. Third, you can attend the meeting and vote in person. Your attendance alone will
not revoke your proxy	. If you have instructed a broker or other nominee to vote your shares, you must follow the directions received from your
broker to change those	e instructions. With respect to voting your proxy via the Internet or by telephone, you can revoke your proxy by voting
again, and only your la	ast action via the Internet or by telephone will be counted.

Q. May I vote in person?

A. Yes. You may attend the special meeting of shareholders and vote your shares of our Common Stock in person. If you hold shares in street name, you must provide a proxy executed by your broker or other

11

7D 1	1	-	_		
1 21	\mathbf{n}	nt.	((mte	ents

nominee in your favor in order to vote your shares at the special meeting. In accordance with our security procedures, all persons attending the special meeting will be required to present picture identification.

Q. Am I entitled to dissenter s or appraisal rights?

A. Under the NRS, holders of our Common Stock are entitled to assert dissenter s rights, but only if they comply with all requirements of Nevada law, which are summarized in this proxy statement. By exercising dissenter s rights, shareholders can demand payment for their shares at a price equal to its fair value immediately before the corporate action. If we and the shareholder cannot agree on a fair value we must then commence an appraisal process in state court to determine the fair value. This appraisal amount could be more than, the same as, or less than the amount that a shareholder will be entitled to receive under the Merger Agreement. Any holder of shares of our Common Stock intending to exercise dissenter s rights, among other things, must submit a written demand for appraisal to us prior to the vote on the adoption of the Merger Agreement and must not vote or otherwise submit a proxy in favor of the adoption of the Merger Agreement. Failure to follow exactly the procedures specified under Nevada law will result in the forfeiture of dissenter s rights. Because of the complexity of the Nevada law relating to dissenter s rights, if you are considering exercising your dissenter s rights, we encourage you to seek the advice of your own legal counsel. For more information, see Dissenter s Rights beginning on page 64 of this proxy statement. In addition, a copy of the full text of NRS 92A.300 to 92A.500 is attached as Annex D to this proxy statement.

Q. Will the merger be a taxable transaction to me?

A. The receipt of cash for shares of our Common Stock pursuant to the merger will generally be a taxable transaction for United States federal income tax purposes. In general, you will recognize gain or loss equal to the difference between the amount of cash you receive and the adjusted tax basis of your shares of our Common Stock. For a more detailed explanation of the tax consequences of the merger, see Material United States Federal Income Tax Consequences beginning on page 45 of this proxy statement. You should consult your tax advisor on how specific tax consequences of the merger apply to you.

Q. Should I send in my stock certificates now?

A. No. After the merger closes, Buyer will arrange for a letter of transmittal containing detailed instructions to be sent to each shareholder. The Merger Consideration will be paid to a shareholder once that shareholder submits a properly completed letter of transmittal accompanied by that shareholder s stock certificates and any other required documentation.

PLEASE DO NOT SEND YOUR STOCK CERTIFICATES NOW.

Q. What happens if the merger is not consummated?

If the Merger Agreement is not adopted by shareholders or if the merger is not completed for any other reason,

A.

	receive any payment for their shares in connection with the merger. Instead, we will remain an independent public amon stock will continue to be traded on the OTC Bulletin Board®.
Q.	What should I do if I have questions?
A. merger to our proxy s	You should direct any questions regarding extra copies of the proxy materials, the special meeting of shareholders or the olicitor, Georgeson Inc., at (877) 278-4751.
If your brokerage firn other nominee for add	n, bank, trust or other nominee holds your shares in street name, you should also call your brokerage firm, bank, trust or litional information.
	12

Table of Contents

AVAILABLE INFORMATION

We file reports, proxy materials and other information with the Securities and Exchange Commission (the Commission). These reports, proxy materials and other information concerning Corgenix can be inspected and copied at the Public Reference Section maintained by the Commission at Room 1024, Judiciary Plaza, 450 Fifth Street, N.W., Washington, D.C. 20549. Copies can be obtained by mail from the Commission at prescribed rates from the Public Reference Section of the Commission at its principal office in Washington D.C. The Commission also maintains a site on the World Wide Web (http://www.sec.gov) that contains reports, proxy and information statements and other information regarding registrants (including Corgenix) that file electronically with the Commission.

Important Notice Regarding the Availability of Proxy Material for the Shareholder Meeting to Be Held on , 2014:

The Proxy Statement for the Corgenix Medical Corporation Special Meeting of Shareholders and Annual Report on Form 10-K for the year ended June 30, 2014 are available at

CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

We caution you that this document contains disclosures that are forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995 about us and the merger. All statements regarding our expected future financial position, results of operations, cash flows, dividends, financing plans, business strategy, budget, projected costs or cost savings, capital expenditures, competitive positions, growth opportunities for existing products or products under development, plans and objectives of management for future operations, markets for stock, and our evaluation of strategic alternatives, including the merger, are forward-looking statements. In addition, forward-looking statements include statements in which we use words such as expect, believe, anticipate, intend, or similar expressions. These forward-looking statements are based upon information presently available to our management and are inherently subjective, uncertain and subject to change, due to any number of risks and uncertainties. Factors that could cause events not to occur as expressed in the forward-looking statements in this document include, but are not limited to, unanticipated delays; the occurrence of any event, change or other circumstances that could give rise to the termination of the Merger Agreement; the outcome of any litigation that has been or may be instituted with respect to the merger; and the inability to complete the merger due to the failure to obtain the approval of our shareholders of the adoption of the Merger Agreement or the failure to satisfy other closing conditions, as well as other risk factors detailed in our Annual Report on Form 10-K for the year ended June 30, 2014, filed with the SEC on September 10, 2014, under the captions Forward looking Statements and Risk Factors and otherwise in our reports and filings with the Securities and Exchange Commission. Many of these factors are beyond our ability to control or predict. You should not place undue reliance on any forward-looking statements, since those statements speak only as of the date that they are made. We assume no obligation to update, revise or correct any forward-looking statements after the date of this document or after the respective dates on which such statements otherwise are made, whether as a result of new information, future events or otherwise, except as otherwise may be required by law.

Table of Contents

Our Board s Recommendation

THE SPECIAL MEETING OF SHAREHOLDERS

rectors
me.
we refer
e officers
ption of
ling the

Our board of directors, by the unanimous vote of all directors, approved the Merger Agreement and determined that the merger and the Merger Agreement are advisable and in the best interests of our company and our shareholders. **Accordingly, our board of directors unanimously**

recommends that you vote FOR the merger proposal, FOR the golden parachute proposal and FOR the adjournment proposal.

Record Date; Stock Entitled to Vote

The holders of record of shares of our Common Stock as of the close of business on September 15, 2014, which is the record date for the special meeting, are entitled to receive notice of and to vote at the special meeting. On the record date, there were 54,228,834 shares of our Common Stock outstanding and entitled to vote held by approximately 80 shareholders of record. Each share of our Common Stock entitles the holder to one vote on all matters properly coming before the special meeting or any adjournment or postponement thereof.

Quorum

Our Bylaws and Nevada law require the presence, in person, or by duly authorized proxy, of the holders of a majority of the outstanding shares of our Common Stock entitled to vote at the special meeting to constitute a quorum. Both abstentions and broker non-votes (as that term is described under Voting beginning on page 15 of this proxy statement) will be counted as present for purposes of determining the existence of a quorum. If a quorum is not present at the special meeting, the special meeting may be adjourned by either the chairman of the meeting or by the holders of a majority of the shares of our Common Stock represented at the meeting, but no other business may be conducted. If a quorum is not present, we expect to adjourn the special meeting to solicit additional proxies and intend to vote any proxies we have received at the time of the special meeting in favor of an adjournment.

Table of Contents

Vote Required for Approval

Our Amended and Restated Articles of Incorporation (Articles), Bylaws and Nevada law require the affirmative vote of holders of a majority of the shares of our Common Stock outstanding on record date entitled to vote to approve the merger proposal. For the merger proposal, you may vote FOR, AGAINST or ABSTAIN. Abstentions and broker non-votes will have the same effect as a vote AGAINST the adoption of the Merg Agreement.

The affirmative vote of the holders of a majority of the shares of our Common Stock present in person or represented by proxy and entitled to vote at the special meeting will be required to approve the golden parachute proposal and the adjournment proposal. Abstentions will have the same effect as a vote AGAINST the golden parachute proposal and the adjournment proposal. Assuming a quorum is present at the special meeting, broker non-votes will have no effect on the approval of the golden parachute proposal or the adjournment proposal.

Voting

If you are a holder of record of shares of our Common Stock, you may vote your shares by attending the special meeting and voting your shares of Common Stock in person, or by one of the following three ways, regardless of whether you plan to attend the special meeting:

- by completing your proxy through the Internet, as listed on the accompanying proxy card;
- by completing your proxy using the toll-free telephone number, as listed on the proxy card; or
- by completing, signing and dating the proxy card and returning it in the enclosed postage-prepaid envelope.

All shares of our Common Stock represented by properly executed proxies received in time for the special meeting will be voted at the special meeting in the manner specified by the holder. If a written proxy card is signed by a shareholder and returned without instructions, the shares of Common Stock represented by the proxy will be voted FOR the merger proposal, FOR the golden parachute proposal, FOR the adjournment proposal and in accordance with the recommendations of our board of directors on any other matters properly brought before the special meeting for a vote.

Shareholders who have questions or requests for assistance in completing and submitting proxy cards should contact our proxy solicitor, Georgeson Inc., at (877) 278-4751.

Shareholders who hold their shares of our Common Stock in street name, meaning in the name of a bank, broker or other person who is the record holder, must either direct the record holder of their shares of our Common Stock how to vote their shares or obtain a proxy from the record holder to vote their shares at the special meeting. Brokers or other nominees may not, in the absence of specific instructions from their customers, give a proxy to vote their customers shares with respect to the merger proposal, the golden parachute proposal or the adjournment proposal. Shares held by brokers or other nominees that are not voted for lack of instructions from their customers are commonly referred to as broker non-votes. You should follow the procedures provided by your broker or nominee regarding the voting of your shares. Broker non-votes will not be counted as votes cast or shares voting and will have the same effect as votes AGAINST the adoption of the Merger Agreement. Assuming a quorum is present at the special meeting, broker non-votes will have no effect on the golden parachute proposal or the adjournment proposal.

Revocability of Proxies

You may change your vote at any time before the polls close at the special meeting. You can do this in one of three ways. First, you can send a written, dated notice to our Secretary at Corgenix Medical Corporation, 11575 Main Street, Suite 400, Broomfield, Colorado 80020, stating that you would like to revoke your proxy. Second, you can complete, date and submit a new proxy card with a later date. Third, you can attend the special meeting and vote in person. Your attendance alone will not revoke your proxy. If you have instructed a broker or nominee to

Table of Contents

vote your shares, you must follow directions received from your broker or nominee to change those instructions. If you have voted your proxy via the Internet or by telephone, you can revoke your proxy by voting again and only your last action via the Internet or by telephone will be counted.

Solicitation of Proxies

This proxy solicitation is being made and paid for by Corgenix on behalf of our board of directors. In addition, we have retained Georgeson Inc. to assist in the solicitation for a fee of approximately \$20,000. In addition to solicitation by mail, our directors, officers and employees may solicit proxies in person or by e-mail, telephone, facsimile or other means of communication. Our directors, officers and employees will not receive any additional compensation for their services, but we will reimburse them for their reasonable out-of-pocket expenses, if any. We will reimburse banks, brokers, nominees, custodians and fiduciaries for their reasonable expenses in forwarding copies of this proxy statement to the beneficial owners of shares of our Common Stock and in obtaining voting instructions from those owners. We will pay all expenses of filing, printing and mailing this proxy statement.

Proposal to Approve Adjournment of the Special Meeting

We are submitting a proposal for consideration at the special meeting to authorize the named proxies to approve one or more adjournments or postponements of the special meeting if there are not sufficient votes to adopt the Merger Agreement at the time of the special meeting, which we refer to as the adjournment proposal. Even though a quorum may be present at the special meeting, it is possible that we may not have received sufficient votes to adopt the Merger Agreement by the time of the special meeting. In that event, we would determine to adjourn or postpone the special meeting in order to solicit additional proxies. The adjournment proposal relates only to an adjournment of the special meeting for purposes of soliciting additional proxies to obtain the requisite shareholder approval to adopt the Merger Agreement. Any other adjournment of the special meeting (e.g., an adjournment required because of the absence of a quorum) would be voted upon pursuant to the discretionary authority granted by the proxy.

The approval of the adjournment proposal would require the affirmative vote of the holders of a majority of the shares of our Common Stock present in person or represented by proxy and entitled to vote at the special meeting. Assuming that a quorum is present, broker non-votes would have no effect on the approval of the adjournment proposal.

Our board of directors recommends that you vote FOR the adjournment proposal so that proxies may be used for that purpose, should it become necessary. Properly executed proxies will be voted FOR the adjournment proposal, unless otherwise noted on the proxies. If the special meeting is adjourned or postponed, we are not required to give notice of the time and place of the adjourned meeting unless our board of directors fixes a new record date for the special meeting.

Questions and Additional Information

If you have more questions about the merger or how to submit your proxy, or if you need additional copies of this proxy statement or the enclosed proxy card or voting instructions, please contact our proxy solicitor, Georgeson Inc., at (877) 278-4751.

Availability of Documents

The reports, opinions or appraisals referenced in this proxy statement will be made available for inspection and copying at our principal executive offices during our regular business hours by any interested holder of our Common Stock. In addition, our list of shareholders entitled to vote at the special meeting will be available for inspection at our principal executive offices at least 10 days before the special meeting.

m	. 1		c	\sim			
Tα	hl	e	Ωt	Cc	n	tei	าts

PROPOSAL ONE THE MERGER

Parties to the Merger

Corgenix Medical Corporation 11575 Main Street, Suite 400

Broomfield, Colorado, 80020 Telephone: (303) 457-4345

We develop and manufacture specialized diagnostic kits for immunology disorders, vascular diseases (including the world s only non-blood-based test for aspirin effect), bone and joint disorders and a line of unique detection products for viral hemorrhagic disease. Our diagnostic products are commercialized for use in clinical laboratories throughout the world. We currently sell over 50 diagnostic products through a global distribution network and have significant experience in product submissions to the FDA and other worldwide regulatory authorities. Additionally, we contract, develop and manufacture products for key medical and life science companies in state-of-the-art facilities in Colorado.

Orgentec Diagnostika

333 W. Wacker, Suite 2800

Chicago, IL 60606

Telephone: (312) 506-2900

Orgentec is a global specialty diagnostics company. The company offers a portfolio of more than 300 tests, primarily enzyme-linked immunosorbent assays (ELISA), that help diagnose rheumatology, thrombosis and gastroenterology disorders, as well as infectious diseases. It also offers a fully automated instrument and associated test kits that enable laboratories to complete multiple assays and deliver faster results than traditional ELISA tests. Orgentec has an established position in Europe and a growing presence in Asia, Latin America and the Middle East. It is headquartered in Mainz, Germany. Orgentec is an affiliate of Water Street Healthcare Partners, LLC (Water Street), a strategic investor focused exclusively on the healthcare industry.

Centennial Medical Holdings, Inc.

c/o Water Street Healthcare Partners, LLC

333 West Wacker Drive, Suite 2800

Chicago, IL 60606

Telephone: (312) 506-2900

Buyer is an affiliate of Orgentec and was formed for the purpose of acquiring and holding us. Buyer has not carried on any activities to date, except for activities incidental to its formation and activities undertaken in connection with the transactions contemplated by the Merger Agreement. Upon consummation of the proposed merger, we will become a wholly owned subsidiary of Buyer.

Centennial Integrated, Inc.

c/o Water Street Healthcare Partners, LLC

333 West Wacker Drive, Suite 2800

Chicago, IL 60606

Telephone: (312) 506-2900

The Merger Sub is a direct, wholly owned subsidiary of Buyer and was formed solely for the purpose of facilitating Buyer s acquisition of us. The Merger Sub has not carried on any activities to date, except for activities incidental to its formation and activities undertaken in connection with the transactions contemplated by the Merger Agreement. Upon consummation of the proposed merger, the Merger Sub will merge with and into us and will cease to exist.

Table of Contents

Background of the Merger

Our board of directors and senior management periodically review and assess our long-term strategy and objectives, as well as developments in the markets in which we operate. Over the past several years, we considered a diverse range of strategic alternatives with a view to increasing shareholder value, including potential opportunities for business combinations, acquisitions and other strategic alternatives. Our reviews include assessment of the overall prospects of Corgenix and its key products, both proprietary and contracted, landscape of our industry and management s financial projections and Corgenix s performance versus those projections. From August 2012 to April 2013, our board of directors held five board meetings. In addition to other matters to come before the board, these meetings included reviews of periodic financial performance versus prior year and budget, and analysis of risks and opportunities. During these reviews, the board regularly expressed its concerns that our financial performance consistently fell short of established financial and operational performance targets, in the past few years and recently. Along with the members of our board, Mr. Critchfield (our Senior Vice President and Chief Financial Officer) regularly attended these board meetings and the meetings described below.

On June 24 and 25, 2013, our board of directors held a meeting in Colorado to review management s most recent fiscal 2013 financial performance versus our budgeted projections. Management estimated fiscal 2013 revenue would show an increase over the prior year but would be considerably less than budget for the year, with net income estimated to slightly exceed both the prior year and the budget. Senior management then presented to our board a budget for fiscal 2014.

There was consensus that even if the budgeted targets were achieved, senior management and our board were not confident that current and projected performance trends would likely result in a material increase in our stock price or in the market valuation of Corgenix. The board decided to continue these discussions in subsequent meetings.

On August 27, 2013, our board of directors held a telephonic meeting at which the directors continued the discussion of Corgenix s progress and financial performance, and the appropriate timing for actively investigating and considering various strategic alternatives for Corgenix. The board asked Mr. Ludvigson, the member of the board of directors nominated by ELITech, to lead an informal review of strategic alternatives together with our senior management. As a follow up to previous discussions conducted by our board over the previous year regarding the future of Corgenix, our board authorized Mr. Ludvigson to compile a range of potential types of transactions that we might pursue, if the board were to make a decision in the future to initiate a formal process. Our board further agreed that the process was merely exploratory in nature, intended to generate information for the board, and was not an authorization to pursue any transaction. Around this same time, a number of shareholders, including our largest shareholder, and directors expressed their discontent with the share price.

In September and October 2013, Mr. Ludvigson, Mr. Simpson (our President and Chief Executive Officer) and Mr. Critchfield together held numerous conference calls to outline a process for exploring strategic alternatives and compiling a list of firms that could potentially provide Corgenix with advisory and/or investment banking services. During this period, after numerous calls with potential advisors, the team narrowed the list to three advisory candidates (including Inverness Advisors, or Inverness) and invited all three to make in-person presentations; however one of those was unable to do so due to scheduling conflicts. We refer to the second advisor presenting as Advisory Candidate A.

On September 30, 2013, we reported financial results for the fiscal year ended June 30, 2013, announcing revenue growth of 9.7% over the prior year, and net income of \$278,536. In the two prior fiscal years, we had incurred net losses of \$623,391 in fiscal 2012 and \$415,553 in fiscal 2011.

On October 11, 2013, senior management held a meeting in our Broomfield, Colorado headquarters with Advisory Candidate A discussing Advisory Candidate A s capabilities and initial thoughts regarding possible strategic alternatives that might be considered by a company such as ours. Four days later, on October 15, 2013, we received a draft M&A Advisory Engagement letter from Advisory Candidate A. The draft letter was distributed to our board for review and consideration.

Table of Contents

On October 16, 2013, Messrs. Simpson and Critchfield met with Inverness in Colorado to discuss their capabilities and initial ideas regarding possible strategic alternatives that might be considered by a company such as ours.

On October 22, 2013, prior to any board decision regarding engaging an advisor, Advisory Candidate A visited our Broomfield headquarters accompanied by representatives of one of their clients that was reviewing its own strategic alternatives. We held an all-day discussion regarding the interest of their client in a potential transaction with Corgenix. Following the meeting, Messrs. Simpson and Critchfield briefed our board on the discussions held with Advisory Candidate A and their client.

On October 23 and 24, 2013, our board of directors held a meeting in Colorado at which Mr. Ludvigson reported on his informal review of strategic alternatives, including preliminary, informal discussions with the candidate advisory firms. Mr. Critchfield attended this meeting. Mr. Ludvigson stated that he and senior management had narrowed the list of potential investment banks/financial advisory firm candidates to two. Mr. Ludvigson reaffirmed to our board that he emphasized to all potential advisory firms that we were only exploring alternatives so that if the board elected to proceed, it would do so only after knowing what types of deals are available in the market, and whether it was likely that Corgenix could pursue one. At that meeting, our board agreed it would be prudent for the board to receive presentations from the two remaining firms in order to interact directly with them and refine their thoughts as to the range of possible strategic alternatives we might pursue.

Following the report from Mr. Ludvigson, the directors agreed that it was appropriate to advance the exploration effort, and appointed a Strategic Subcommittee consisting of Mr. Gouze (Chairman of the Subcommittee) and Mr. Ludvigson to be responsible for interacting with both potential investment bankers and senior management. Mr. Ludvigson was responsible for organizing the planned face-to-face meetings with the candidate firms including scheduling and pre-meeting preparation of detailed agendas including specific questions our board wished to have answered by the candidate firms. Senior management presented an overview of the October 22 meeting with Advisory Candidate A and their client, and the board decided further discussion with that client would be premature and that we should continue the process in an orderly manner.

On November 14, 2013, we reported financial results for the fiscal quarter ended September 30, 2013, announcing revenue growth of 2.0% over the same quarter in the prior year with a decline in net income from \$198,114 to \$84,047.

On November 25, 2013, our board of directors held a meeting in Colorado at which the Strategic Subcommittee provided a full report of activities following the previous board meeting. As a follow up to the board decision at the previous board meeting, Mr. Ludvigson had provided the two firms with a list of questions presented to him by board members to ensure each candidate addressed them in their presentations.

At this meeting, our board of directors received a detailed presentation by Advisory Candidate A, at which they discussed their capabilities and initial thoughts regarding possible strategic alternatives that might be considered by Corgenix. Following the departure of Advisory Candidate A representatives, the board held an extensive review of our current state of affairs, including our current business plan and senior management s latest long range plan (LRP) and a discussion of the risks associated with achieving the LRP and whether or not our LRP demonstrated sufficient progress towards the goal of maximizing long-term shareholder value, or if pursuing a transformative, strategic transaction would be the best way to maximize shareholder value. The board determined that the discussion would continue following the scheduled presentation by the second finalist firm (Inverness) at the next board meeting.

On December 16, 2013, our board of directors held a meeting in Colorado. Representatives of Inverness and representatives of our outside legal counsel, Husch Blackwell LLP, attended this meeting. The directors received a presentation from Inverness. Following the Inverness presentation the board discussed the two presentations including strengths, weaknesses, historical performance, principal credentials and quality of the presentation of each firm, along with the estimated fees and expenses if either were to be engaged by us. Our board discussed with Husch Blackwell the pros and cons of a going private transaction, by deregistering our Common Stock under the Securities Exchange Act of 1934, thereby relieving us of burdensome public reporting obligations.

Table of Contents

The board also engaged in a discussion about its alternatives, including whether to: (1) test the waters (as described below); (2) commence a sale process with the specific goal of engaging in a sale/auction process; or (3) stay the course by continuing to pursue our strategic plans.

The board decided to initiate a test the waters approach on an anonymous basis that is to conduct a preliminary market check to explore whether any third parties would have an interest in discussing a strategic transaction, what types of transactions might be of interest to the market, and what range of potential values might be expected. Our Board continued the discussion on Corgenix s LRP, including expectations going forward and assessing whether or not the LRP demonstrated sufficient progress to maximize shareholder value in the long term. The Board also considered that our senior management had not, for the past several years, accurately projected future results and had not achieved the financial targets and forecasted financial performance projected at the beginning of each fiscal year.

On December 17, 2013, immediately following the annual shareholder meeting at our Broomfield headquarters for the fiscal year ended June 30, 2013, our board of directors continued its discussion on strategic alternatives, with a thorough review of the current state of company affairs and plan in light of the updated LRP, the additional strategic alternatives presented by counsel on the previous day, and various aspects of the investment banker presentations. The board determined that based on the presentations received, Inverness provided the best opportunity for a more timely and successful execution of the board strategy, and senior management was instructed to attempt to finalize a definitive engagement letter with Inverness.

On December 20, 2013, following several conference calls with senior management, Inverness submitted a draft engagement letter for consideration.

On January 7, 2014, our board of directors held a telephonic meeting to continue discussion of strategic alternatives and the board s fiduciary responsibilities to shareholders. A representative of Husch Blackwell attended this meeting. Husch Blackwell reviewed with the board its fiduciary responsibilities under these circumstances. The directors agreed to begin a process to explore whether a transaction to infuse new technology or other resources into Corgenix, or whether a sale of Corgenix or a merger transaction, might be available on terms favorable to and in the best interests of Corgenix and our shareholders. Our board reviewed the draft engagement letter with Inverness prepared by senior management and counsel, and agreed to engage Inverness for the limited purpose of assisting the board in its exploration of whether or not to pursue any number of potential strategic transactions. The board discussed how the engagement letter should be clarified to better reflect the board s current goals. Senior management was instructed to communicate to Inverness in writing that Inverness was to:

- (1) explore whether a strategic transaction might be possible;
- (2) advise the board as to the potential variations of such a strategic transaction (perhaps a sale of Corgenix or certain assets, infusion of new technology or other resources by way of licensing or joint venture arrangement, or other approach yet to be identified); and
- (3) proceed to a market check or any other process only if and when the board elected to move from the exploratory phase to active pursuit of a transaction. The board approved the retention of Inverness pursuant to the engagement agreement in substantially the form presented to the board.

On January 10, 2014, we signed an engagement letter with Inverness. Four days later, Mr. Simpson and Mr. Ludvigson met with Inverness in San Francisco to review the process to date and discuss issues related to operating on an anonymous basis.

On February 13, 2014, we reported financial results for the fiscal quarter ended December 31, 2013, announcing quarterly revenue growth of 9.4% and six-month revenue growth of 5.5%. Net income for the six-month period increased 5.5% and increased 254.4% for the quarter compared to the prior periods.

On February 24 - 25, 2014, our board of directors met to review recent activity including a report of the meetings with Inverness. Representatives of Inverness and a representative of Husch Blackwell attended this meeting. Inverness presented a report to the board, reviewing the process used to identify potential

Table of Contents

targets and related matters. Inverness stated that the exploratory process and communications with parties began on February 20, 2014 and that 24 parties had been contacted to date. Inverness indicated that the exploratory process was ongoing.

On March 11, 2014, we filed an 8-K followed by a press release on March 12, 2014 disclosing the decision of the board to explore strategic alternatives to enhance value for shareholders, which could include, among others, possible joint ventures, strategic partnerships or alliances, a sale of Corgenix or other possible transactions, and that Inverness had been retained as financial advisor to Corgenix. On March 12, 2014, the stock price increased immediately following this announcement to close at \$0.27 per share, a 20% increase over the prior trading day s close, which was \$0.225 per share.

On March 12, 2014, our board of directors held a telephonic meeting at the request of senior management. Representatives of Inverness and a representative of Husch Blackwell attended this meeting. Mr. Simpson reported to our board that, after discussion with legal counsel and due to the fact that some Corgenix employees had received an anonymous email message regarding the board s decision to explore strategic alternatives, we had filed a Form 8-K the previous day. Inverness then updated our board on their exploratory process to date including each target company s status (such as whether an NDA was currently in place, a management presentation had taken place or been scheduled, and initial feedback regarding interest in further discussions). Inverness expressed the view that our recent filing on Form 8-K allowed Inverness to engage in a more open effort to complete the preliminary market check, and they anticipated receiving unsolicited inquiries, which might enhance the field of potential transaction candidates.

Throughout March, April and May, we and Inverness worked with potential targets, including both financial and strategic entities. During this period, in addition to the public announcement by us regarding the board s decision to explore strategic alternatives, over 100 parties were contacted, either telephonically, by email or by both methods of communication related to Inverness market check. Of those contacted, 24 expressed an interest and executed non-disclosure and confidentiality agreements (NDAs) and began their due diligence process. This included conference calls covering non-confidential information and exchanges of confidential information after execution of NDAs. Of all the parties we communicated with, six of them advanced to submission of non-binding letters of indication of interest. These six companies are referred to as Targets A through E, and Orgentec, an affiliate of Water Street.

In March and April, 2014, we signed NDAs with Targets A though E, and began an extensive series of conference calls, in person meetings and exchanges of confidential information, enabling each of the targets to evaluate Corgenix for the purpose of proposing a strategic alternative. Conference call topics included presentation of financial information, strategies, capabilities, products, markets, competition and potential. Only Target B did not visit Corgenix.

On April 14-15, 2014, at a meeting of our board of directors in Colorado, Inverness provided an update on the status of ongoing discussions. For example, Inverness reported that they had received five unsolicited inquiries from potential transaction candidates who had seen the 8-K disclosure. Inverness stated that through their efforts in conjunction with senior management, over 80 companies had been contacted with many progressing to conference calls, in-person meetings, and data room access. These 80-plus companies represented the vast majority of the over 100 who would eventually be contacted during the market check. The interested parties included both domestic and international as well as financial and strategic parties. Inverness completed their presentation by stating that it was their conclusion that there was sufficient interest expressed by both domestic and international parties, as well as financial and strategic parties, currently engaged in the process with us to conclude the preliminary market analysis, referred to as Phase I.

After an extensive discussion regarding the process to date and results of Phase I, our board determined unanimously that the Phase I process had been successful and that it was appropriate to advance the process to Phase II to solicit preliminary non-binding indications of interest.

However, Inverness was, to the extent possible, to also attempt to re-engage with target companies that had previously expressed interest, but had declined further discussions. Inverness designed a process to solicit indications of interest and instructed recipients to respond by May 9, 2014.

Table of Contents

Between mid-April and early May, we and Inverness responded to diligence requests and held numerous conference calls, in-person meetings and facility tours with the target companies.

In mid-to-late April, Inverness sent 16 process letters and in early May another 3 process letters to targets that were identified as remaining interested in pursuing a possible transaction pursuant to the board s instructions. The process letters specified that indication of interest letters were to be returned no later than May 9, 2014.

On April 30, 2014, Mr. Ludvigson learned that Water Street was pursuing opportunities in the diagnostics space and suggested that Inverness contact principals at Water Street. Inverness contacted Water Street the following day and set up a conference call with Corgenix senior management for the following week.

On May 6, 2014, we signed an NDA with Water Street, on behalf of Orgentec.

On May 9, 2014, we held a conference call with representatives of Water Street which covered a broad range of general business, financial and operations topics. Following the conference call, Water Street, on behalf of Orgentec, contacted Inverness to indicate their interest in a potential transaction. Water Street stated that it would not be in a position to submit a non-binding indication of interest on May 9, but could be in a position to do so by May 14.

By May 9, 2014, we had received indications of interest from five targets:

- On May 2, 2014, we received an initial indication of interest from Target A, a U.S.-based, privately held company, for an all-stock transaction. Target A stated that the purported value of the stock received by our shareholders would be approximately \$13 million or \$0.23 per share.
- On May 9, 2014, we received initial indications of interest from:
- Target B, a foreign-based, privately held company, for an all-cash transaction of approximately \$12.4 million or \$0.22 per share. Target B stated it based its offer on the summary financial information and EBITDA projections of 2014.
- Target C, a U.S.-based, privately held company, for either an all-stock or a combination of cash and stock transaction, for a purported range of price of \$0.27 to \$0.34 per share. Final determination of valuation and structure to be determined following the satisfactory completion of their due diligence;

- Target D, a foreign-based, publicly listed company, for an all-cash transaction of approximately \$0.32 per share; and
- Target E, a U.S.-based, micro-cap publicly listed company, for an all-stock, at-market transaction. The exact exchange rate would be determined by using the volume weighted average closing price during the 20 trading days immediately preceding the execution of an agreement and would be subject to fluctuations until such time. At the time of our board s consideration, our board of directors viewed a valuation as high as \$0.35 per share, as a benchmark indication of Target E s proposal on that date, while noting that the actual valuation upon consummation may differ materially.

On May 13, 2014, our board of directors held a telephonic meeting to review the five Indications of Interest (IOI) letters which had been received by Inverness, along with follow-up communication with several of the targets following receipt of the IOI letters. Representatives of Inverness and a representative of Husch Blackwell attended this meeting. Our board discussed each preliminary IOI in detail, including ranges of potential values presented by the targets, identities of the presenting parties, proposed deal structures and proposed forms and amounts of consideration (cash, stock, or a combination). Inverness assessed each IOI with our board. Our board spent extensive time assessing the total range of proposed values and the structure of each IOI. Based on the content of the IOI letters, the range of values and potential structures presented and the expectations that further IOIs would

Table of Contents

be submitted (Water Street, on behalf of Orgentec), our board concluded that the IOI letters received from Target A and Target B were not within the range of values that the board felt would demonstrate sufficient value to our shareholders and with regard to Target A, questioned whether the value of the company was verifiable, and directed Inverness to contact those parties to inform them that absent a material and compelling increase in their offers, we would terminate discussions with them. Our board also concluded that the IOI letters from Targets C, D and E were sufficient to warrant further discussions, coupled with a request that they submit revised offers with the goal of receiving best and final indications of interest for further consideration within 45-60 days. The board directed Inverness to notify it of any further IOIs submitted. Husch Blackwell reminded the board of its fiduciary responsibilities at this phase of the process.

On May 14, 2014, we received an initial indication of interest from Water Street, on behalf of Orgentec, for an all-cash transaction, of approximately \$0.25 to \$0.27 per share, which was provided to our board for review. Water Street, on behalf of Orgentec, referenced that its valuation reflected a 21% to 30% premium to the average daily share price over the 90 day period prior to our announcement that we were exploring strategic alternatives.

On May 15, 2014, our board of directors concluded that the proposed terms contained in the Orgentec IOI were sufficient to warrant further instructions along with Targets C, D and E.

On May 15, 2014, we reported financial results for the fiscal quarter ended March 31, 2014, announcing a slight decline in quarterly revenue of 1.8% and net income of 2.5% and an increase in revenue of 3.1% and net income of 4.0% for the nine-month period ending March 31, 2014.

On May 15-16, 2014, Inverness contacted Targets C, D, E and Water Street, on behalf of Orgentec, and invited them to participate in the next phase of the process and noted that a letter explaining the timing of and procedures governing the next phase of the process will be provided within the next week.

On May 16, 2014, Inverness contacted Target B to notify them that our board determined that its IOI was not of sufficient value, and that absent a material and compelling increase, Corgenix would terminate discussions with them. Three days later in a conference call, Target B informed Inverness that it would not be in a position to improve their offer and was therefore withdrawing from further consideration.

On May 19, 2014, Inverness contacted Target A to notify it that our board determined that its IOI was not sufficient and that absent a material and compelling increase in their offer in the value of the consideration it was offering, Corgenix would terminate discussions with it. The following day, Target A informed Inverness it would not be in a position to improve its offer and was therefore withdrawing from further consideration.

On May 20, 2014, Targets C, D, E and Water Street, on behalf of Orgentec, were directed to submit a final acquisition proposal and mark-up of the draft definitive agreement (which would be provided at a later date) by July 10, 2014.

Throughout the remainder of May and June, we continued the conference call discussions, exchange of documents and in-person meetings with Targets C, D and E and Water Street, on behalf of Orgentec.

On June 18, 2014, our board of directors met telephonically to receive a process update. Representatives of Inverness and a representative of Husch Blackwell attended this meeting. Inverness reported that Targets A and B had withdrawn, while the remaining four had been actively conducting due diligence. Inverness discussed each remaining target company s status, key areas of focus, and a summary of completed and scheduled on site and telephonic meetings with each. Inverness summarized indications of interest by the remaining four Target companies, reviewing with our board how current non-binding offers from each target company translated into fully diluted per share consideration. Our board discussed a revised timeline including plans to supply each potential bidder with a draft merger agreement prepared by Husch Blackwell within the next week, with instructions that each submit a proposed mark-up of that agreement along with their best and final offer no later than July 10. Our board discussed and approved the terms of the draft merger agreement with Husch Blackwell.

Table of Contents

On June 18, 2014, Inverness received a call from Target C stating that while it was still evaluating Corgenix, due to a lack of strategic fit, there was reasonable chance it would be withdrawing from the process after further consideration and preliminary due diligence.

On June 24, 2014, Inverness received a call from Target D stating that, due to a lack of strategic fit, it had elected to withdraw from the process.

On June 25, 2014, the draft definitive agreement was made available through our data room to the remaining parties.

On June 26, 2014, we issued a press release announcing receipt of an NIH grant to continue development of our rapid tests for Ebola.

On June 26, 2014, our board of directors held a conference call meeting with Inverness to review all activities. Mr. Critchfield, representatives of Inverness, and a representative of Husch Blackwell attended this meeting. Inverness reported that one target had withdrawn from consideration and another was likely to withdraw (Targets C and D), leaving two remaining, who would continue their due diligence efforts. Inverness reviewed with our board their discussions with the withdrawing targets, and conveyed its advice regarding the potential of convincing either to reconsider, and an overview of current areas of due diligence being undertaken by the two remaining targets. Our board concluded that we should continue advancing the current process with the remaining targets, but that it was prudent to investigate opportunities to bring back into the discussion selected parties who had been active in the process but that had withdrawn.

On July 9, 2014, the financial advisors of Target C contacted Inverness to withdraw their client from further consideration.

On July 10, 2014, we received an offer from Target E, a transaction consisting of (i) an upfront, at-market, stock-for-stock exchange, which would be subject to stock price movements by both parties up to execution, and (ii) a two-way contingent payment of shares payable in 2018 based on certain performance metrics.

On July 10, 2014, we received an offer from Water Street, on behalf of Orgentec, to acquire Corgenix in an all-cash transaction of \$0.26 per share, without financing conditions, but including conditions related to due diligence, voting agreements with our largest shareholder, Wescor, Inc., an affiliate of ELITech, and our senior management and directors, and the transition of certain arrangements between us and ELITech.

On July 9, 2014, the day before receipt of the offers, our stock closed at \$0.27 per share.

On July 11, 2014, Inverness conducted follow up discussions with the financial advisors to Target E to gain additional information regarding Target E $\,$ s offer.

On July 14, 2014, our board of directors held a meeting at company headquarters to review the two remaining offers received, and to reconsider the alternative of remaining independent and not pursuing any transaction based on the offers received. Representatives of Inverness and a representative of Husch Blackwell attended this meeting. Husch Blackwell advised our board on factors it may consider in comparing offers and factors it may consider on whether it should reject or pursue a transaction. Our board thoroughly reviewed the material terms of each offer, focusing on the fact that one offer (Orgentec) was an all-cash offer without any financing contingency, and the other (Target E) was a stock-for-stock exchange. The board expressed various concerns regarding the offer from Target E, including conversion of Corgenix shares into shares of Target E as well as questions regarding the financial condition and future prospects of Target E and the uncertain nature of the value, if any, of the two-way contingent payment of shares in 2018 based on certain performance metrics.

Senior management presented its own assessment of the technical, scientific and financial aspects of the combined company should such a transaction with Target E be consummated. The assessment included a presentation by Dr. Pitts, our Director of R&D, who reviewed Target E s product, technology, platform and scientific and clinical strategies. Mr. Critchfield, our CFO, made a presentation of his analysis of Target E s current financial condition, recent financial results, micro-cap market value and a preliminary pro forma on the

Table of Contents

combined company. Inverness also discussed the degree of risk to the shareholders associated with such a transaction based on our senior management s review. The board determined that exchanging Corgenix shares for a minority position in Target E as well as the uncertain nature of the value, if any, of the two-way contingent payment of shares in 2018 was not in the best interests of our shareholders, and further discussion with them would not be prudent unless they could significantly increase the cash component of their offer and increase the certainty of value related to the contingent payment contained in their offer. The board believed that the benefits of an all-cash offer from Orgentec outweighed the high degree of financial risk of Target E and its offer and other risks of such a combination with Target E.

Our board of directors instructed Inverness to contact Target E to determine whether it could significantly increase the cash component of its offer and increase the certainty of value related to the contingent payment contained in their offer. Our board also instructed Inverness to contact Water Street, on behalf of Orgentec, to clarify certain matters raised in their offer letter, including reducing their proposed termination fee of 3% of the aggregate consideration, and seeking to increase in the value offered. Our board then considered the perceived benefits of staying the course meaning termination of all discussions with possible acquisition candidates including the board s thoughtful consideration of the likelihood that the stock price would return to levels at or below those that existed before the March 11, 2014 announcement of the exploration of strategic alternatives, risk of not executing successfully on operational plans or not achieving stated financial goals based on past shortfalls, limited liquidity and the lack of products or performance history, all as compared to accepting an all cash purchase by Orgentec.

On July 14 and 15, 2014, Inverness had several conference calls with Water Street, on behalf of Orgentec, seeking both an increase in the per share consideration and a decrease in its termination fee. Orgentec agreed to reduce the proposed termination fee to 2% of the aggregate consideration and to increase their offer to \$0.27 per share, but emphasized that \$0.27 per share was its best and final offer.

On July 16, 2014, Inverness held a conference call with financial advisors to Target E, explained that our board viewed its offer as insufficient as presently constructed, and inquired about Target its ability to significantly increase the cash component of its offer. The following day, Target E stated that it was not in position to comply with the board s directive and could not materially improve its offer and withdrew from further consideration.

On July 16, 2014, our board of directors held a telephonic meeting to receive a detailed report of recent discussions with Target E and with Water Street. Representatives of Inverness and representatives of Husch Blackwell attended this meeting. Our board conducted a through discussion of the various aspects of the two remaining offers. Inverness indicated to our board that while it had not yet reached any conclusions regarding fairness of the possible transaction with Orgentec, and could not do so until it had completed its analysis and its internal procedures for finalizing such conclusions, it appeared likely that a complete analysis of Corgenix and its business and the price offered by Orgentec could be expected to result in a conclusion that the price fell within the range of prices that would be considered fair to the holders of Corgenix Common Stock from a financial point of view. Our board concluded unanimously that \$0.27 was the highest price available in the market. Our board once again summarized the reasons for rejecting the Target E offer, as noted above. Our board then unanimously agreed that it was appropriate and advisable to move forward with Orgentec at the proposed offer of \$0.27 per share, instructed Husch Blackwell to prepare and negotiate a limited exclusivity agreement with Water Street, on behalf of Orgentec, which could be extended based on the achievement of certain milestones in the process and directed Inverness to notify Water Street of the board s position.

On July 18, 2014, we held a conference call to coordinate on process and timing with representatives of Water Street and Much Shelist, P.C., outside counsel to Orgentec. Representatives of our management team, Inverness and Husch Blackwell participated in this call.

On July 18, 2014, our board of directors by unanimous written consent authorized Inverness to prepare a fairness opinion to be considered by our board in connection with its final determination on whether to approve a definitive agreement with affiliates of Orgentec.

Over the next several days, we held conference calls with representatives of Orgentec and consultants to Orgentec related to key customers, key contracts and financial due diligence.

Table of Contents

On July 25, 2014, we signed an Exclusivity Letter with Water Street, on behalf of Orgentec, agreeing to an exclusivity period expiring August 22, 2014, which was later extended to August 27, 2014. From July 25, 2014 to August 27, 2014, senior management and representatives of Husch Blackwell responded to many due diligence inquiries from Water Street, on behalf of Orgentec, and negotiated the final terms of the Merger Agreement. The key issues that were resolved during this time were the amount of the termination fees of \$324,000, the addition of a reverse termination fee in favor of Corgenix in the amount of \$648,000, the cap on out-of-pocket expense reimbursement of \$400,000, the terms of the voting agreement that Orgentec requested at a minimum be executed by all executive officers and directors who were shareholders, and the determination that while the CEO and CFO would not be retained as employees post-closing, all other employees were expected to retain their positions at the headquarters in Broomfield, Colorado.

On July 30, 2014, we issued a press release announcing that our Ebola Test Research publication had earned a divisional best abstract award at the meeting of the American Association for Clinical Chemistry. This was during the time period when there was a significant amount of global interest on the Ebola outbreak including several U.S. healthcare workers contracting the disease resulting in evacuation to the U.S. There was significant interest in all companies with an Ebola strategy, including Corgenix. Our stock price increased \$0.08 to \$0.39 per share on July 30.

Between July 30 and August 11, we received frequent inquiries from the media and from investors regarding our Ebola development project. It was the opinion of senior management that the stories on the internet regarding the status of our program were being significantly overstated in some cases, and in others contained erroneous assertions. The closing stock price rose as high as \$0.46 per share on August 4.

On August 11, 2014, in an attempt to address widespread and often false information regarding our Ebola program, we issued a press release updating and clarifying our activities with respect to an Ebola rapid test project funded by the NIH. The press release stated that we had not yet finalized development of the Ebola products, and that we do not currently have an Ebola test, nor will we likely have one in the new term. The closing stock price on that day declined \$0.05 to \$0.40 per share, and over the following 12 trading days prices ranged from \$0.34 to \$0.38.

On August 21, 2014, we signed an amendment to the Exclusivity Letter with Water Street, on behalf of Orgentec, extending the term for five days to August 27, 2014.

On August 25, we were notified by Water Street that the transaction had been approved by its investment committee, with approval to execute the agreement after the market closed on August 27, 2014.

On August 26, 2014, our board of directors held a telephonic meeting. Representatives of Inverness and representatives of Husch Blackwell attended this meeting. Husch Blackwell once again summarized the board s fiduciary responsibilities as well as the material terms of the execution version of the Merger Agreement and voting agreement, including the lack of a financing condition, the fiduciary out in the event of a superior proposal or certain intervening events and the amounts of the termination fees.

Inverness provided to the board Inverness s detailed financial analysis of the transaction, and Inverness rendered its oral opinion (which was subsequently confirmed in writing) to Corgenix s board of directors that, as of such date and based upon and subject to the qualifications, limitations and assumptions stated in its opinion, the Merger Consideration, which consists of \$0.27 in cash per share of Common Stock to be received by the holders of Common Stock is fair, from a financial point of view, to the holders of Common Stock. Inverness s opinion is more fully described in Opinion of Corgenix s Financial Advisor and the full text of the written opinion of Inverness is attached as Annex C hereto. In

addition, Inverness noted the historical trading prices of Corgenix Common Stock over the last three years, which had always been below \$0.27 prior to the March 11, 2014 announcement that we were exploring strategic alternatives, and the lack of liquidity in our stock (80% of the trading days over the last three years had less than \$10,000 of value traded).

Senior management presented their rationale for concluding that staying the course was not the better alternative under all of the circumstances, while at the same time noting the various opportunities that Corgenix currently has. However, the risks related to competition, regulatory matters, operational and distributions issues and our limited financial and human resources outweighed the opportunities including the recent contract with Eli Lilly and the opportunities that the Ebola outbreak might present in the future of a diagnostic test being successfully developed. Management and our board stated their belief that the commercial opportunity for an Ebola test is much more limited than the market appears to believe in light of how early we are in the development process, the risk of competition and the operational risks involved. Husch Blackwell summarized a number of factors that weighed in favor of and against the transaction. Husch Blackwell also advised our board that while the

Table of Contents

trading price of our stock can be a factor in valuations, it is not always indicative of fair value for a number of reasons, including when a micro-cap stock like ours is thinly traded, is not followed by many, if any, analysts has a small public float and is subject to short-term spikes based on announcements like ours in March that we were exploring strategic alternatives and the speculative trading following the Ebola outbreak. Our board concluded that the historical trading price before the March 11 announcement was a more reliable indicator of fair value than the current trading price and our board further believed that our largest shareholder was very supportive of the proposed transaction. After taking into account all of the information presented and discussing it thoroughly, our board approved the Merger Agreement as being in the best interest of our shareholders.

After the market closed on August 27, 2014, we signed the Merger Agreement with Buyer and Merger Sub, affiliates of Orgentec.

Before the market opened on August 28, 2014, we issued a press release and filed a Form 8-K with the SEC announcing the signing of the Merger Agreement. The stock price on that date declined \$0.08 to \$0.26 per share.

Buyer and ELITech are currently discussing the future distribution relationship between ELITech and Corgenix after the merger. The First Amended and Restated Joint Product Development Agreement, by and between us, Financiere ELITech SAS, and Wescor, Inc., dated July 28, 2011, has expired by its terms, and Buyer and ELITech are currently negotiating an agreement to clarify certain post-termination obligations pursuant to such agreement.

Reasons for the Merger and Recommendation of our Board of Directors

- Best Price. After a robust and lengthy auction process and extended discussions and negotiation with Water Street, on behalf of Orgentec, our board of directors came to the conclusion that the Merger Consideration of \$0.27 per share in cash to be received by our shareholders represented the highest price Buyer was willing to pay.
- Absence of Reasonably Available Superior Acquisition Proposals. Our board of directors considered the fact that, following its March 12, 2014 press release announcing its retention of Inverness to advise our board and announcing its exploration of strategic alternatives, including a possible sale of our company, we engaged in a thorough auction process to identify parties who might have an interest in acquiring us. Our board of directors considered the fact that a total of 113 prospective strategic and financial buyers were contacted for potential indications of interest, a total of 24 potential buyers entered into confidentiality agreements with us, and 6 potential buyers, including Orgentec, eventually submitted bids. Of the six potential buyers submitting bids, two were rejected by our board and two withdrew. Besides Orgentec, the other potential buyer submitted an all equity proposal which contained various contingencies. Following this process, our board of directors concluded that the merger was more favorable to our shareholders than any other acquisition proposal reasonably available.
- Opinion of Inverness to the Board of Directors. Our board of directors considered Inverness financial analysis and presentation delivered to our board of directors on August 26, 2014 and the written opinion, dated August 26, 2014, of Inverness to the effect that, as of such date, the Merger Consideration to be paid to holders of our Common Stock (other than shareholders who properly demand dissented under the NRS, Buyer or Merger Sub, or the affiliates of such holder) was fair, from a financial point of view, to such holders. The opinion is more fully described under Opinion of Corgenix s Financial Advisor beginning on page 31 of this proxy statement.

• All-Cash Consideration. Our board of directors considered that the form of consideration to be paid to holders of our Common Stock in the merger is cash, which will provide certainty of value and liquidity to our shareholders, while avoiding long-term business risk. The all-cash merger will also enable our shareholders to realize liquidity of their entire ownership interests in the context of market conditions of micro-cap public companies with limited trading volumes that offer little liquidity.

27

Table of Contents

•	Limited Existing Liquidity.	Our board of directors	considered our market	capitalization and the	e relatively low ve	olume of trading in
our	Common Stock, which limit the ab	oility of our shareholder	s to realize the quoted v	alue of their shares.		

- Uncertainty of Future Market Price. Our board of directors considered the financial condition, results of operations, competitive position and prospects of our business, as well as current economic, financial market and stock market conditions, historical trading values of our Common Stock in the open market, and the volatility in the trading price of our Common Stock. Based on all of this information, our board of directors believed that the trading price of our Common Stock would not sustain a value greater than the Merger Consideration of \$0.27 per share.
- Comparability of Other Alternatives. Our board of directors, with the assistance of our senior management and Inverness, evaluated a broad range of potential strategic alternatives available to us other than the sale of our entire company; considered that the stock price would likely drift back to pre-announcement levels; and factored into its decision our limited liquidity, relatively small number of products, financial and human resources and track record to determine that the other alternatives would not adequately address our long-term strategic challenges. These facts, in the context of the robust and lengthy market check conducted as part of this strategic undertaking, led our board to conclude unanimously that none of the reasonably available strategic alternatives to a sale of the company at the price obtained under the Merger Agreement were viable and the merger represented the best alternative to maximize shareholder value and was in the best interests of our shareholders.
- Risks of Remaining an Independent Public Company. Our board of directors also considered the risks, uncertainties and costs of continuing to operate as an independent public company, including our current and historical financial condition, results of operations, business and prospects, as well as our financial plan and prospects by continuing to pursue our existing business plan. Also, our board of directors considered the risks inherent in market conditions and achieving our growth targets, the substantial uncertainty regarding the long-term success of our financial plan or prospects, and the expense and regulatory risks of remaining a public company in relation to our relatively small size and market capitalization and lack of financial and human resources. Our board of directors reviewed our financial forecasts and analyses prepared and furnished to our board by senior management for the fiscal years ending June 30, 2014 through 2018 (the Forecasts). Our board of directors noted that the Forecasts reflected a level of financial performance that was materially stronger than our historic financial performance and that the Forecasts were subject to risks and uncertainties that could cause our actual financial performance to differ materially from the results projected in the Forecasts. Our board of directors noted that the shareholders would continue to be subject to the risks and uncertainties of our financial plan, and prospects, all in an uncertain economic environment. In light of the foregoing, our board of directors concluded that the merger was more favorable to our shareholders than the possibility of continuing to operate as an independent public company.
- *Likelihood of Consummation*. Our board of directors considered the likelihood that the merger will be completed on a timely basis and with minimal disruption on the terms set forth in the Merger Agreement based on, among other factors:
- the scope of the conditions to completion;
- our ability, pursuant to the Merger Agreement, to seek specific performance of Buyer and Merger Sub s obligations under the Merger Agreement;

the belief of our board of directors that the acquisition would not be subject to significant regulatory impediments; and
 the absence of a financing condition to Buyer s obligation to consummate the transaction.
 Terms of the Merger Agreement. Our board of directors considered the terms and conditions of the Merger Agreement and other transaction documents, including:

Table of Contents

• the parties representations, warranties and covenants;
• our ability, under specified conditions, to furnish information to, and engage in discussions and negotiations with, other potential buyers in response to a bona fide unsolicited written takeover proposal that we conclude in good faith is, or could reasonably be expected to result in or lead to, a superior proposal, as more fully described in The Merger Agreement No Solicitation beginning on page 57 of this proxy statement;
• our ability, under specified conditions, to terminate the Merger Agreement in order to enter into an alternative acquisition agreement with respect to a superior proposal, provided that we pay Buyer a termination fee of \$324,000 plus out of pocket expenses up to \$400,000, as more fully described in The Merger Agreement Termination of the Merger Agreement beginning on page 62 of this proxy statement, and The Merger Agreement Termination Fees beginning on page 63 of this proxy statement;
our ability, under specified conditions, to change the recommendation of our board of directors with respect to the merger in response to a superior proposal or a material development or change in material circumstances occurring or arising after the date of the Merger Agreement, the existence and material consequences of which were not known by our board of directors on or before the date we entered into the Merger Agreement, on the understanding that Buyer would then have the right to terminate the Merger Agreement and collect the termination fee described above and in The Merger Agreement No Solicitation Change in Recommendation beginning on page 58 of this prostatement, The Merger Agreement Termination of the Merger Agreement beginning on page 62 of this proxy statement and The Merger Agreement Termination Fees beginning on page 63 of this proxy statement;
• the requirement that, under specified conditions, if Buyer or the Merger Sub materially breaches its obligations such that the conditions to closing would not be satisfied, Buyer must pay us a reverse termination fee of \$648,000 plus out of pocket expenses up to \$400,000, without our having to establish any damages, as more fully described in The Merger Agreement Termination Fees beginning on page 63 of this proxy statement;
• the limited number and nature of the conditions to Buyer s and the Merger Sub s obligations to consummate the merger; and
• that the Merger Agreement provides us with adequate operational flexibility to conduct our business generally in the ordinary course between the signing of the Merger Agreement and the completion of the merger.
See The Merger Agreement beginning on page 49 of this proxy statement, for a more complete description of the terms and conditions of the Merger Agreement.

Reasonable Termination Fee. Our board of directors considered the termination fees and reimbursement of out of pocket expenses,

including the views of our financial and legal advisors as to the customary nature of the presence and size of termination fees in transactions

proxy

similar to the merger. Our board of directors concluded that a termination fee of this size for the transactions contemplated by the Merger Agreement, representing approximately 2% of the equity value of our company implied by the Merger Consideration plus out of pocket expenses, was not likely to preclude or unduly deter a third party from making, or inhibit our board of directors from evaluating and negotiating, and, if appropriate, our terminating the Merger Agreement to enter into a transaction that is, a superior proposal, as more fully described under The Merger Agreement No Solicitation Change in Recommendation beginning on page 58 of this proxy statement and The Merger Agreement Termination of the Merger Agreement beginning on page 62 of this proxy statement.

Our board of directors also considered potential risks or negative factors relating to the merger, including the following:

Table of Contents

• Historical Market Prices. Our board of directors considered the Merger Consideration relative to the current market prices of our Common Stock, including the fact that the offer of \$0.27 per share represented an approximate 21% discount below the closing price of our Common Stock on August 27, 2014 (the last trading day prior to the public announcement of the merger) and an approximate 27% discount below the closing price of our Common Stock on August 25, 2014 (the last trading day prior to the date of the board meeting to approve the Merger Agreement). While our Common Stock had generally traded at prices greater than \$0.27 per share since the announcement of our decision to explore strategic alternatives, our Common Stock did not close at a price above the Merger Consideration price of \$0.27 per share on a single day over the five year period from March 2009 to March 11, 2014, and the trading range of our Common Stock for the same period was between \$0.06 and \$0.25 per share.

Our board of directors noted that the announcement that we intended to explore strategic alternatives appears to have led to speculation concerning our company. Our board also noted that inaccurate information regarding our Ebola program, and how global events might impact our future financial performance, was circulating publicly, and believed that this may have led to speculative trading that caused the quoted price of our Common Stock, which has very low trading volume, to increase. In the case of Ebola, our board and senior management felt compelled to publish additional clarification to correct errors in the press, online blogs and inquiries we had received. Our stock has low daily trading volume, a small public float, and is covered by only one research analyst. These factors, coupled with what appears to be speculative trading by a limited number of investors, may have contributed to the recently elevated market price, which our board and senior management do not believe is indicative of fair value.

- Risk of Non-Completion. Our board of directors considered the risk that the merger might not be completed. Our board considered that we would incur significant transaction and opportunity costs attempting to consummate the merger, including the distraction of our management and employees, the potential disruption of our relationships with customers, suppliers and others with whom we do business, and possible employee attrition.
- Non-Solicitation Covenant. Our board of directors considered that the Merger Agreement precludes us from actively soliciting takeover proposals and restricts our ability to provide information to, or engage in discussions or negotiations with, third parties regarding any takeover proposal except under specified conditions. Our board considered that these provisions could discourage the emergence or consummation of a takeover proposal that would be more favorable to our shareholders than the merger. Our board also considered the risk that the termination fee that we would have to pay in order to accept a superior proposal could discourage third parties from making a takeover proposal that would be more favorable to our shareholders than the merger. However, our board of directors also took into account the public and active solicitation of bids that occurred before the execution of the Merger Agreement, its belief that the non-solicitation provisions were reasonable in the circumstances and its belief that they were not likely to preclude or unduly deter a third party from making a takeover proposal.
- Restrictions on Our Business. Our board of directors considered the fact that the Merger Agreement restricts our ability to engage in certain activities between the date of the Merger Agreement and the effective time of the merger, and that these restrictions could prevent us from taking advantage of attractive business opportunities outside the ordinary course of business, such as potential acquisitions, which would be advisable if we were to remain an independent company. See The Merger Agreement Conduct of Our Business Pending the Merger beginning on page 54 of this proxy statement.
- Taxable Transaction. Our board of directors considered that an all-cash transaction would be taxable to our shareholders for United States federal income tax purposes. See Material United States Federal Income Tax Consequences beginning on page 45 of this proxy statement.

• Cash-out Transaction. Our board of directors considered that the nature of the merger as a cash transaction will prevent our shareholders from participating in our future earnings or growth and any potential appreciation in the value of our shares, including any value that could be achieved if we were to engage in

Table of Contents

future strategic or other transactions or if our business were to continue to improve despite the risks and uncertainties we currently face.

• Availability of Dissenter s Rights. Our board of directors considered the availability of statutory dissenter s rights under Nevada law in the cash-out merger for shareholders who do not vote their shares in favor of adoption of the Merger Agreement (and who otherwise comply with the statutory requirements of Nevada law), and who believe that exercising such rights would yield them a greater per share amount than the Merger Consideration, which could delay the transaction as the Buyer s obligations to consummate the merger are conditioned on dissenting shares not exceeding ten percent (10%) of the outstanding shares. See Dissenter s Rights beginning on page 64 of this proxy statement.

Our board of directors concluded that the potentially negative factors associated with the proposed merger were outweighed by the opportunity for our shareholders to monetize their investment in us for the cash Merger Consideration of \$0.27 per share. Our board of directors believed that the proposed merger would maximize the immediate value of the shareholders—shares and eliminate the unavoidable risks and uncertainties affecting our future prospects. Accordingly, our board of directors determined that the merger is fair to, and in the best interests of, our company and our shareholders, and approved and declared advisable the Merger Agreement, the merger and the other transactions contemplated by the Merger Agreement.

The foregoing discussion of the information and factors considered by the board of directors is not intended to be exhaustive, but includes the material factors considered by the board of directors. In view of the variety of factors considered in connection with its evaluation of the merger, the board of directors did not find it practicable to, and did not, quantify or otherwise assign relative weights to the specific factors considered in reaching its determination and recommendation. In addition, individual directors may have given different weights to different factors. The board of directors did not undertake to make any specific determination as to whether any factor, or any particular aspect of any factor, supported or did not support its ultimate determination. The board of directors based its recommendation on the totality of the information presented.

Our board of directors recommends that you vote FOR approval of the merger proposal.

In considering the recommendation of our board of directors with respect to the merger proposal, you should be aware that our directors and executive officers have interests in the merger that may be different from, or in addition to, those of our shareholders generally. The board of directors was aware of and considered these interests, among other matters, in evaluating and negotiating the Merger Agreement and the merger, and in recommending that the Merger Agreement be adopted by our shareholders. See Interests of Our Directors and Executive Officers in the Merger beginning on page 40 of this proxy statement.

Opinion of Corgenix s Financial Advisor

Corgenix engaged Inverness, to act as its financial advisor with respect to the acquisition of Corgenix by Buyer. On August 26, 2014, Inverness rendered its oral opinion (which was subsequently confirmed in writing) to Corgenix s board of directors that, as of such date and based upon and subject to the qualifications, limitations and assumptions stated in its opinion, the Merger Consideration, which consists of \$0.27 in cash per share of Corgenix Common Stock, par value \$0.001 per share, to be received by the holders of Common Stock is fair, from a financial point of view, to the holders of Common Stock.

The full text of Inverness written opinion, dated as of August 26, 2014, is attached as Annex C to this Proxy Statement. Inverness written opinion sets forth, among other things, the assumptions made, procedures followed, factors considered and limitations upon the review undertaken by Inverness in rendering its opinion. You are encouraged to read the opinion carefully in its entirety. The following is a summary of Inverness opinion and the methodology that Inverness used to render its opinion. This summary is qualified in its entirety by reference to the full text of the opinion.

Inverness opinion, the issuance of which was approved by Inverness Fairness Opinion Review Committee, is addressed to the Corgenix board of directors, addresses only the fairness, from a financial point of

Table of Contents

view, of the consideration to be received by the holders of Common Stock and does not constitute a recommendation to any shareholder of Corgenix as to how such shareholder should vote with respect to the proposed transaction or any other matter. The terms of the proposed ess in

transaction were determined through arm s-length negotiations between Corgenix and Buyer and were unanimously approved by the Corgenix board of directors. Inverness was not requested to address, and its opinion does not in any manner address, Corgenix s underlying business decision to proceed with or effect the proposed transaction or the likelihood of consummation of the proposed transaction. In addition, Inverne expressed no opinion on, and it does not in any manner address, the fairness of the amount or the nature of any compensation to any officers, directors or employees of any parties to the proposed transaction, or any class of such persons, relative to the Merger Consideration to be paid it the proposed transaction or otherwise. No limitations were imposed by the Corgenix board of directors upon Inverness with respect to the investigations made or procedures followed by it in rendering its opinion.
In arriving at its opinion, Inverness, among other things:
• reviewed a draft of the Merger Agreement, dated August 25, 2014 (the Draft Agreement), including the financial terms and conditions set forth therein;
• reviewed Corgenix s audited financial statements for the fiscal year ended June 30, 2013, Corgenix s unaudited financial statement for the nine months ended March 31, 2014 and a preliminary draft of Corgenix s audited financial statements for the fiscal year ended June 30, 2014;
 reviewed certain other business, operating and financial data of Corgenix, prepared and furnished to us by Corgenix s management, including the Forecasts;
 held discussions with the senior management team of Corgenix concerning the business, past and current operations, financial condition and future prospects of Corgenix, and certain other matters Inverness believed necessary or appropriate to its inquiry;
 compared the financial performance of Corgenix and trading activity of Common Stock with that of certain other publicly-traded companies that Inverness deemed relevant;

- compared the financial terms of the merger with the financial terms, to the extent publicly available, of other transactions that Inverness deemed relevant;
- reviewed historical stock prices and trading volumes of Corgenix Common Stock;

- reviewed Corgenix s draft annual report on Form 10-K for the fiscal year ended June 30, 2014, Corgenix s annual report on Form 10-K for the fiscal year ended June 30, 2013, and Corgenix s quarterly report on Form 10-Q for the fiscal quarter ended March 31, 2014;
- · reviewed certain other publicly-available business, operating and financial information of Corgenix; and
- made such other studies and inquiries, reviewed such other data, and considered such other factors as Inverness has deemed, in its sole judgment, to be necessary, appropriate or relevant to render the opinion set forth herein.

In arriving at its opinion, Inverness assumed and relied upon the accuracy and completeness of the financial and other information supplied or otherwise made available to Inverness by Corgenix and all publicly-available financial and other information regarding Corgenix reviewed by Inverness without any independent verification of such information (and Inverness did not assume responsibility or liability for any independent verification of such information) and Inverness further relied upon the assurances of the management of Corgenix that all such information was complete and accurate in all material respects and that they were unaware of any facts or

Table of Contents

circumstances that would make such information incomplete or misleading in any material respect. With respect to financial projections, upon the advice of Corgenix, Inverness assumed that such projections were reasonably prepared on a basis reflecting the best currently-available estimates and judgments of the management of Corgenix as to the future financial performance of Corgenix and that Corgenix would perform substantially in accordance with such projections. Inverness assumed no responsibility for and Inverness expressed no view as to any such projections or estimates or the assumptions on which they are based. In arriving at Inverness opinion, Inverness did not conduct a physical inspection of the properties and facilities of Corgenix and has not made or obtained any valuations or appraisals of the assets or liabilities of Corgenix. Inverness opinion necessarily was based upon market, economic and other conditions as they existed on, and could be evaluated as of, August 26, 2014. Inverness assumed no responsibility for updating or revising its opinion based on events or circumstances that may occur after August 26, 2014. Inverness expressed no opinion as to the prices at which shares of Corgenix Common Stock would trade following the announcement of the proposed transaction.

Additionally, Inverness assumed the accuracy of the representations and warranties contained in the Merger Agreement and all agreements related thereto and that the final form of the Merger Agreement reviewed by Inverness would not vary from the Draft Agreement in any regard that is material to Inverness analysis. Inverness also assumed, with Corgenix s consent, that in the course of obtaining necessary regulatory and third party approvals and consents for the proposed transaction, no modification, delay, limitation, restriction or condition will be imposed that will have an adverse effect on Corgenix or the contemplated benefits of the proposed transaction and that the proposed transaction will be consummated in accordance with the terms of the Merger Agreement, without waiver, modification or amendment of any material term, condition or agreement therein.

Inverness did not evaluate the solvency of Corgenix under any state, federal or other laws relating to bankruptcy, insolvency or similar matters, and has undertaken no independent analysis of any pending or threatened litigation, regulatory action, possible unasserted claims or other contingent liabilities, to which Corgenix is a party or may be subject. At the direction of Corgenix and with its consent, Inverness opinion makes no assumption concerning, and therefore does not consider, the possible assertion of claims, outcomes or damages arising out of any such matters. Inverness did not express any opinion as to any tax or other consequences that might result from the proposed transaction, nor did Inverness opinion address any legal, tax, regulatory or accounting matters, as to which Inverness understood that Corgenix had obtained such advice as it deemed necessary from qualified professionals.

In connection with rendering its opinion, Inverness performed certain financial, comparative and other analyses as summarized below. In arriving at its opinion, Inverness did not ascribe a specific range of values to the shares of Common Stock but rather made its determination as to fairness, from a financial point of view, to the holders of Common Stock of the consideration to be received by such holders in the proposed transaction on the basis of various financial and comparative analyses. The preparation of a fairness opinion is a complex process and involves various determinations as to the most appropriate and relevant methods of financial and comparative analyses and the application of those methods to the particular circumstances. Therefore, a fairness opinion is not readily susceptible to summary description.

In arriving at its opinion, Inverness did not attribute any particular weight to any single analysis or factor considered by it but rather made qualitative judgments as to the significance and relevance of each analysis and factor relative to all other analyses and factors performed and considered by it and in the context of the circumstances of the particular transaction. Accordingly, Inverness believes that its analyses must be considered as a whole, as considering any portion of such analyses and factors, without considering all analyses and factors as a whole, could create a misleading or incomplete view of the process underlying its opinion.

The following is a summary of the material financial analyses used by Inverness in preparing its opinion to the Board. Certain financial analyses summarized below include information presented in tabular format. In order to fully understand the financial analyses used by Inverness, the tables must be read together with the text of each summary, as the tables alone do not constitute a complete description of the financial analyses. In performing its analyses, Inverness considered industry performance, regulatory, general business, economic, market and financial conditions

and other matters, many of which are beyond the control of Corgenix or any other parties to the proposed transaction. None of Corgenix, Buyer, Merger Sub, Inverness or any other person assumes responsibility if future

Table of Contents

results are materially different from those discussed. Any estimates contained in these analyses are not necessarily indicative of actual values or
predictive of future results or values, which may be significantly more or less favorable than as set forth below. In addition, analyses relating to
the value of the businesses do not purport to be appraisals or reflect the prices at which the businesses may actually be sold.

Selected Publicly-Traded Companies Analysis

Heska Corporation

Inverness reviewed and compared certain publicly-available financial and operating information of selected publicly-traded companies that
participate in the diagnostic and laboratory testing industries and that Inverness deemed to be relevant to analyzing the historical and projected
financial and operating information of Corgenix. The selected companies were:

•	BG Medicine, Inc.
•	Biomerica Inc.
•	Bio-Rad Laboratories, Inc.
•	Bio-Reference Laboratories Inc.
•	Chembio Diagnostics, Inc.
•	diaDexus, Inc.
•	Enzo Biochem Inc.
•	ERBA Diagnostics, Inc.

•	Laboratory Corporation of America Holdings
•	Miraca Holdings Inc.
•	Neogenomics Inc.
•	Omega Diagnostics Group PLC
•	Quest Diagnostics Inc.
•	Transgenomic Inc.
company	calculated and compared various financial multiples and ratios of Corgenix and the selected companies. As part of its selected analysis, Inverness calculated and analyzed each company s ratio of its enterprise value (EV) to (a) its revenue for calendar year 2014 (CV2014E Revenue.) (b) its estimated revenue for calendar year 2014 (CV2014E Revenue.)

Inverness calculated and compared various financial multiples and ratios of Corgenix and the selected companies. As part of its selected company analysis, Inverness calculated and analyzed each company s ratio of its enterprise value (EV) to (a) its revenue for the last twelve months (LTM Revenue), (b) its estimated revenue for calendar year 2014 (CY2014E Revenue), (c) its estimated revenue for calendar year 2015 (CY2015E Revenue), (d) its estimated earnings before interest taxes, depreciation and amortization (EBITDA) for the last twelve months (LTM EBITDA), (e) its estimated EBITDA for calendar year 2014 (CY2014E EBITDA) and (f) its estimated EBITDA for calendar year 2015 (CY2015E EBITDA). Additionally, Inverness calculated and analyzed each company s ratio of its stock price to (x) its earnings per share for the last twelve months (LTM P/E), (y) its estimated earnings per share for calendar year 2014 (CY2014E P/E), and (z) its estimated earnings per share for calendar year 2015 (CY2015E P/E). In each case, estimates are based on Wall Street research consensus estimates obtained from commercially-available sources. All of the calculations were performed and based on publicly-available financial data and closing prices as of August 22, 2014. Based upon these calculations, Inverness selected ranges of multiples for Corgenix and applied such range to the relevant Corgenix financial statistic to calculate ranges of implied prices per share of Common Stock. The following summarizes the result of these calculations:

Table of Contents

Multiple	Range of Multiples Utilized in the Analysis(1)	Implied Multiple of Proposed Transaction	Range of Implied Price Per Share for Proposed Transaction
EV / LTM Revenue	1.1x - 2.1x	1.3x	\$0.25 - \$0.40
EV / CY2014E Revenue	1.1x - 2.4x	1.2x	\$0.25 - \$0.48
EV / CY2015E Revenue	1.0x - 2.3x	1.0x	\$0.26 - \$0.52
EV / LTM EBITDA	7.9x - 13.7x	17.1x	\$0.16 - \$0.23
EV / CY2014E EBITDA	7.7x - 9.7x	14.1x	\$0.17 - \$0.21
EV / CY2015E EBITDA	6.7x - 9.2x	8.4x	\$0.23 - \$0.29
LTM P/E	15.8x - 30.1x	41.1x	\$0.11 - \$0.20
CY2014E P/E	16.0x - 27.2x	26.6x	\$0.17 - \$0.27
CY2015E P/E	13.3x - 23.7x	12.3x	\$0.29 - \$0.49

⁽¹⁾ Range reflects 1st quartile to 3rd quartile of the selected companies.

The selected companies listed above are all participants in the diagnostic and laboratory testing industries; however, Inverness did not analyze every company that could be deemed to be a participant in the same industry as Corgenix. None of the selected companies are identical to Corgenix, and, due to the inherent differences between the business, operations and prospects of Corgenix and those of the selected companies, Inverness analysis involved complex considerations and judgments concerning differences between the business, financial and operating characteristics and prospects of Corgenix and the selected companies that could affect the public trading values of each in order to provide a context in which to consider the results of the quantitative analysis.

Selected Merger & Acquisition Transaction Analysis

Inverness reviewed and compared the purchase prices and financial multiples paid in selected publicly-announced transactions. In connection with this analysis, Inverness reviewed the following transactions involving target companies that participate in the diagnostic and laboratory testing industries that were announced between August 22, 2011 and August 22, 2014 and that Inverness, based on its experience with merger and acquisition transactions, deemed relevant:

Target Enterprise Value Less Than \$150 Million

Date Announced	Target	Buyer
03/20/14	DiaSpect Medical AB	EKF Diagnostics Holdings plc
03/20/14	Selah Genomics Inc	EKF Diagnostics Holdings plc
01/07/14	Alverix, Inc.	Becton, Dickinson and
		Company
10/22/13	PLUS Diagnostics, Inc.	Miraca Life Sciences, Inc.
07/30/13	Immco Diagnostics, Inc.	Trinity Biotech plc
05/06/13	Biohelix Corporation	Quidel Corp.
01/03/13	Hologic, Inc. LIFECODES® Business	Immucor Inc.
12/16/12	MorphoSys s AbD Serotec Research Reagent & Diagnostic Business	Bio-Rad Loboratories, Inc.
09/17/12	Complete Genomics, Inc.	BGI
08/06/12	Cardiocore Lab, Inc.	CardioNet, Inc.

Table of Contents

07/09/12	GenturaDx-USA, Inc.	Luminex Corporation
03/01/12	Fiomi Diagnostics AB	Trinity Biotech plc
02/13/12	SeraCare Life Sciences, Inc.	Linden LLC
12/02/11	Verum Diagnostica GmbH	Roche Holdings AG
10/13/11	Claros Diagnostics, Inc.	Opko Health, Inc.

Target Enterprise Value Greater Than \$150 Million

Date Announced	Target	Buyer
06/17/14	ProteinSimple	Techne Corp.
04/07/14	Iquum, Inc.	Roche Molecular
		Systems, Inc.
02/04/14	Crescendo Bioscience, Inc.	Myriad Genetics Inc.
01/22/14	Solstas Lab Partners Group, LLC	Quest Diagnostics Inc.
01/16/14	Ortho-Clinical Diagnostics, Inc.	The Carlyle Group LP
11/11/13	Novartis Vaccines & Diagnostics, Inc., Blood Transfusion	Grifols, S.A.
	Diagnostics Unit	
09/03/13	BioFire Diagnostics, Inc.	bioMérieux SA
04/15/13	Life Technologies Corporation	Thermo Fisher Scientific, Inc.
02/24/13	HemoCue AB	Radiometer Medical ApS
01/07/13	Verinata Health, Inc.	Illumina Inc.
09/17/12	IRIS International Inc.	Beckman Coulter, Inc./
07/16/12	One Lambda, Inc.	Thermo Fisher Scientific, Inc.
06/04/12	MEDTOX Scientific Inc.	Laboratory Corp. of America
		Holdings
05/17/12	Dako A/S	Agilent Technologies Inc.
04/30/12	Gen-Probe Inc.	Hologic Inc.
10/06/11	Caris Life Sciences, Inc., Anatomic Pathology Business	Miraca Holdings Inc.

Inverness calculated ranges of multiples of LTM Revenue and LTM EBITDA and applied such ranges to Corgenix s LTM Revenue and LTM EBITDA, respectively, to calculate a range of implied prices per share of Corgenix Common Stock. The following tables set forth the results of such calculations:

M M Is	Range of Multiples	Implied Multiple of	Range of Implied Price Per		
Multiple	Utilized in the Analysis(1)	Proposed Transaction	Share for Proposed Transaction		
EV / LTM Revenue	1.3x 4.8x	1.3x	\$0.28 - \$0.82		
EV / LTM EBITDA	13.4x 25.1x	17.1x	\$0.23 - \$0.36		

Table of Contents

Target Enterprise Value Less Than \$150 Million

(1) Range reflects 1st quartile to 3rd quartile of the selected transactions.

Target Enterprise Value Greater Than \$150 Million

Multiple	Range of Multiples Utilized in the Analysis(1)	Implied Multiple of Proposed Transaction	Range of Implied Price Per Share for Proposed Transaction		
EV / LTM Revenue	2.8x 5.8x	1.3x	\$0.50 - \$0.98		
EV / LTM EBITDA	12.0x 21.0x	17.1x	\$0.21 - \$0.32		

(1) Range reflects 1st quartile to 3rd quartile of the selected transactions.

None of the transactions analyzed in the selected merger and acquisition transaction analysis are identical to the proposed transaction. The reasons for and the circumstances surrounding each of the selected transactions analyzed were diverse and there are inherent differences in the business, operations, financial conditions and prospects of Corgenix and the companies included in the selected transaction analysis.

Accordingly, Inverness analysis involved complex considerations and judgments concerning differences between the characteristics of the selected transactions and the proposed transaction which could affect the acquisition values observed in the selected transactions and the value of Corgenix, in order to provide a context in which to consider the results of the quantitative analysis.

Discounted Cash Flow Analysis

In order to estimate the present value of Common Stock, Inverness performed a discounted cash flow analysis of Corgenix. A discounted cash flow analysis is a traditional valuation methodology used to derive a valuation of an asset by calculating the present value of estimated future cash flows of such asset. Present value refers to the current value of future cash flows or amounts and is obtained by discounting those future cash flows or amounts by a range of discount rates that take into account macroeconomic assumptions and estimates of risk, the opportunity cost of capital, expected returns and other appropriate factors.

Inverness used the Forecasts, as provided by Corgenix s management, to perform two discounted cash flow analyses: (i) one based on the terminal multiple method and (ii) the second based on the perpetuity growth method. In conducting these analyses, Inverness assumed that Corgenix would perform in accordance with the Forecasts.

Terminal Multiple Method

To calculate the implied range of prices per share of Common Stock using the terminal multiple method, Inverness first estimated the terminal value of Corgenix s projected cash flows by applying a range of terminal multiples Inverness deemed relevant to Corgenix s estimated fiscal year 2018 EBITDA, which multiples ranged from 7.0x to 9.0x and were derived in Inverness professional judgment, based on its experience with merger and acquisition transactions. Inverness calculated projected unlevered free cash flow from fiscal year 2015 through fiscal year 2018 using the Forecasts and discounted these cash flows and the terminal value to present values as of July 1, 2014 using discount rates of 15.0% to 20.0%, based on Corgenix s weighted average cost of capital. To calculate the estimated enterprise value of Corgenix, Inverness added the present value of (i) the unlevered free cash flow from fiscal year 2015 through fiscal year 2018 to (ii) the terminal value. Using the estimated enterprise value of Corgenix, Inverness then added Corgenix s net cash to calculate a range of implied equity values. These implied equity values were then divided by fully-diluted shares outstanding using the treasury stock method to calculate the range of implied prices per share of Common Stock as \$0.38 to \$0.51.

Table of Contents

Perpetuity Growth Method

To calculate the implied range of prices per share of Common Stock using the perpetuity growth method, Inverness first estimated the terminal value of Corgenix s projected cash flows by applying a range of perpetuity growth rates Inverness deemed relevant to Corgenix s estimated fiscal year 2018 free cash flow, which growth rates ranged from 2.0% to 3.0% and were derived in Inverness professional judgment, based on its experience with merger and acquisition transactions. Inverness calculated projected unlevered free cash flow from fiscal year 2015 through fiscal year 2018 using the Forecasts and discounted these cash flows and the terminal value to present values as of July 1, 2014 using discount rates of 15.0% to 20.0%, based on Corgenix s weighted average cost of capital. To calculate the estimated enterprise value of Corgenix, Inverness added the present value of (i) the unlevered free cash flow from fiscal year 2015 through fiscal year 2018 to (ii) the terminal value. Using the estimated enterprise value of Corgenix, Inverness then added Corgenix s net cash to calculate a range of implied equity values. These implied equity values were then divided by fully-diluted shares outstanding using the treasury stock method to calculate the range of implied prices per share of Corgenix Common Stock as \$0.25 to \$0.35.

Premiums Paid Analysis

Inverness reviewed and compared implied premiums paid in selected other transactions that Inverness, based on its experience with merger and acquisition transactions, deemed relevant. These selected transactions included eleven transactions that closed between August 22, 2011 and August 22, 2014 involving the all-cash acquisition of North America-based public healthcare companies, each with a total transaction value between \$5 million and \$50 million. For each of the selected transactions, Inverness compared, based on publicly-available information, the implied premiums paid in such transactions represented by the per share acquisition price as compared to the target company s closing share price for various periods prior to the announcement of the execution of the definitive agreement with respect to the relevant transaction.

Set forth below is Inverness analysis of the range of the implied premiums paid in the selected transactions to the stock price one month prior to the announcement of the execution of the definitive agreement with respect to the relevant transaction, and the application of this range of premiums to the price of Corgenix Common Stock as of one month prior to March 11, 2014, the date of filing of Corgenix s current report on Form 8-K announcing the exploration of strategic alternatives:

Range of Premiums Paid(1)	Premium of Proposed Transaction	Range of Implied Per Share Price for the Proposed Transaction
(11.7%) 78.6%	27.2%	\$0.19 - \$0.38 / Share

(1) Range reflects 1st quartile to 3rd quartile of the selected transactions.

Inverness noted that the Merger Consideration per share to be received by holders of shares of the Corgenix Common Stock pursuant to the Merger Agreement was \$0.27 per share.

The premiums paid analysis discussed above is based on the historical performance of Common Stock. Inverness noted that Corgenix Common Stock experienced increased trading volumes and price volatility starting in March of 2014, which coincided with the filing of Corgenix s

Form 8-K disclosing the Corgenix board of director s decision to explore strategic alternatives. The premiums paid analysis was presented to the Board to provide additional information and to add context to the other analyses performed by Inverness, as described above. The valuation derived from the premiums paid analysis is not necessarily indicative of Corgenix s present or future value or results.

General

Inverness, as part of its investment banking services, is regularly engaged in the valuation of businesses, assets and securities in connection with mergers, acquisitions, private placements and valuations for corporate and other purposes. Corgenix s board of directors selected Inverness because of its familiarity with Corgenix s industry

Table of Contents

and its qualifications, reputation and experience in the valuation of businesses and securities in connection with mergers and acquisitions generally, as well as experience in transactions comparable to the proposed transaction.

Inverness is acting as financial advisor to Corgenix in connection with the proposed transaction and will receive a fee of \$546,000 for its services, \$60,000 of which consisted of a retainer that has been paid, \$200,000 of which was paid upon delivery of the Inverness fairness opinion and the balance of which is contingent upon the consummation of the proposed transaction. In addition, Corgenix has agreed to reimburse Inverness expenses in connection with the proposed transaction and indemnify Inverness for certain liabilities that may arise out of Inverness engagement by Corgenix and the rendering of Inverness opinion. Except for the services described above, Inverness has not performed investment banking or financial services for Corgenix or Buyer in the past two years. Inverness may perform such services in the future and expects to receive customary fees for such services.

Inverness and its affiliates engage in a wide range of investment banking and financial advisory services. In the ordinary course of its business, Inverness and its affiliates may actively trade and effect transactions in the equity, debt and/or other securities (and any derivatives thereof) and financial instruments (including loans and other obligations) of Corgenix, Buyer and their respective affiliates for its own account and for the accounts of its customers and, accordingly, may at any time hold long or short positions and investments in such securities and financial instruments.

Certain Corgenix Forecasts

We do not, as a matter of course, publicly disclose financial forecasts as to future financial performance, earnings or other results as set forth below and are especially cautious of making financial forecasts due to the unpredictability of the underlying assumptions and estimates. However, in connection with the evaluation of a possible transaction, we made available to Orgentec and its advisors, other potential targets and their advisors, our board and Inverness, as applicable, the Forecasts that were prepared by our senior management and discussed with our board and not for public disclosure.

A summary of the Forecasts is included below solely to give shareholders access to the information that was made available to Orgentec and its advisors, other potential targets and their advisors, our board and Inverness, as described below, and is not included in this proxy statement in order to influence you to make any investment decision with respect to the merger, or to influence your decision whether to vote for or against the proposal to adopt the Merger Agreement. The inclusion of this information should not be regarded as an indication that our board, its advisors or any other person considered, or now considers, it to be material or to be a reliable prediction of actual future results. Our senior management s internal financial forecasts, upon which the Forecasts were based, are subjective in many respects. There can be no assurance that the Forecasts will be realized or that actual results will not be significantly higher or lower than forecasted. The Forecasts cover multiple years and such information by its nature becomes subject to greater uncertainty with each successive year. Senior management has not, for the past several years, accurately projected future results and has not achieved the financial targets and forecasted financial performance that they had projected at the beginning of each fiscal year. As a result, the inclusion of the Forecasts in this proxy statement should not be relied on as necessarily predictive of actual future events. The Forecasts and summary information should be evaluated, if at all, in conjunction with the historical financial statements and other information regarding us contained in our public filings with the SEC.

The Forecasts were based on numerous variables, estimates and assumptions that are inherently uncertain and may be beyond our control, and which may prove to have been, or may no longer be, accurate. Important factors that may affect actual results and cause actual results to differ materially from the Forecasts include, but are not limited to, risks and uncertainties relating to our business (including our ability to achieve strategic goals, objectives and targets over the applicable periods), industry performance, the regulatory environment, general business and

economic conditions, market and financial conditions, new product development, competition, various risks set forth in our reports filed with the SEC, and other factors described or referenced under Cautionary Note Regarding Forward-Looking Statements beginning on page 13 of this proxy statement. In addition, the Forecasts also reflect assumptions that are subject to change and are susceptible to multiple interpretations and periodic revisions based on actual results, revised prospects for our business, changes in general business or economic conditions, or any other transaction or event that has occurred or that may occur and that was not anticipated at the time the Forecasts were prepared. The Forecasts do not take into account any circumstances, transactions or events occurring after the date they were prepared. Accordingly, there can be no assurance that the Forecasts will be realized or that our future financial results will not materially vary from the Forecasts.

Table of Contents

No one has made or makes any representation to any shareholder or anyone else regarding, nor assumes any responsibility for the validity, reasonableness, accuracy or completeness of, the information included in the Forecasts set forth below. You are cautioned not to rely on the Forecasts. Some or all of the assumptions which have been made regarding, among other things, the timing of certain occurrences or impacts, may have changed since the date such Forecasts were made. We have not updated and do not intend to update or otherwise revise the Forecasts, even in the short term, to reflect circumstances existing after the date when they were made or to reflect the occurrence of future events, including the merger contemplated by the Merger Agreement. Further, the Forecasts do not take into account the effect of any failure of the merger to occur and should not be viewed as accurate or continuing in that context. We have made no representation to Buyer or Merger Sub or any other person in the Merger Agreement or otherwise, concerning these Forecasts.

The following is a summary of those Forecasts: The Forecasts reflected (i) revenues of approximately \$12.7 million for fiscal year 2015, \$14.8 million for fiscal year 2016, \$17.4 million for fiscal year 2017 and \$20.8 million for fiscal year 2018; (ii) EBITDA (earnings before interest, taxes, depreciation and amortization) of approximately \$1.2 million for fiscal year 2015, \$2.2 million for fiscal year 2016, \$3.4 million for fiscal year 2017, and \$5.0 million for fiscal year 2018; (iii) net income of approximately \$0.8 million for fiscal year 2015, \$1.8 million for fiscal year 2016, \$3.0 million for fiscal year 2017 and \$3.1 million for fiscal year 2018; and (iv) unlevered free cash flow (EBIT (earnings before interest and taxes), less taxes, capital expenditures and any increase in working capital, plus depreciation, amortization and any decrease in working capital) of approximately \$0.5 million for fiscal year 2015, \$1.5 million for fiscal year 2016, \$2.4 million for fiscal year 2017 and \$2.4 million for fiscal year 2018. Our senior management projected that no taxes would be paid by Corgenix through fiscal year 2017 due to its operating history. All calendar year projections reflect our senior management s assumption that the average of the fiscal year projections provides a reasonable basis for the calendar year projections. These Forecasts reflect a level of financial performance that is materially stronger than we have historically achieved.

Financing of the Merger

The merger is not conditioned on Buyer s ability to obtain financing. Buyer has various sources available to it to fund the consideration to be paid to our shareholders under the Merger Agreement.

Deregistration of Corgenix Common Stock

If the merger is completed, our Common Stock will be removed from listing on the OTC Bulletin Board® and deregistered under the Securities Exchange Act of 1934, as amended, and we will no longer file periodic reports with the SEC.

Interests of Our Directors and Executive Officers in the Merger

In considering the recommendation of our board of directors with respect to the Merger Agreement, holders of shares of our Common Stock should be aware that our directors and executive officers have interests in the merger that may be different from, or in addition to, those of our shareholders generally. These interests may create potential conflicts of interest. Our board of directors was aware that these interests existed when it approved the merger and the Merger Agreement. The material interests are summarized below.

Common Stock, Stock Options and Warrants to be Cashed Out in the Merger

The following table provides the cash proceeds that each of our current directors and named executive officers are expected to receive in the closing of the merger from:

- the conversion into the Merger Consideration of the shares of our Common Stock that they already own free of any restrictions; and
- the cash-out of their in-the-money options to purchase shares of our Common Stock (i.e., options with an exercise price less than \$0.27 per share) that are already vested, as well as the in-the-money options that

40

Table of Contents

will vest in accordance with the Merger Agreement on or before the closing date, in each case in an amount equal to the Merger Consideration less the applicable exercise price of such options.

Information regarding the beneficial ownership of vested and unrestricted Common Stock is based on information provided by or on behalf of the individuals named in the table. For purposes of this table, we assume that the merger will occur on the date of the special meeting. The amounts in the table do not reflect any taxes that will be incurred upon the conversion of shares into the Merger Consideration or the cash-out of options.

Value of Equity Securities (estimated as of the date of the special meeting)

Name		Value Common stock (\$)	alue of Vested and Unrestricted Corgenix Equity Stock Subtotal options (\$) (\$)				Value of Corgenix Equity that will Vest Upon the Merger Stock options (\$)		Total cash Payment for Equity Securities (\$)	
Named Executive Officers		, , ,		•		, , ,		• '''		• •
Douglass T. Simpson	\$	117,676	\$	100,800	\$	218,476	Ф	50,400	¢	268,846
William H. Critchfield	\$	98.967	\$	85,500	\$	184,467		36,000		220,467
	\$	49,371	\$	46,800	\$	96,171		28,800		124,971
Ann L. Steinbarger	Ф	49,371	Ф	40,800	Ф	90,171	Ф	20,000	Ф	124,971
Non-employee Directors			_				_			
Robert Tutag	\$	259,470	\$	54,000	\$	313,470	\$	0	\$	313,470
Brandon J. Price	\$	0	\$	13,500	\$	13,500	\$	0	\$	13,500
Stephen P. Gouze	\$	284,205	\$	54,000	\$	334,205	\$	0	\$	334,205
David Ludvigson	\$	0	\$	43,200	\$	43,200	\$	0	\$	43,200
Dennis Walczewski	\$	0	\$	54,000	\$	54,000	\$	0	\$	54,000
Dennis Fusco	\$	0	\$	21,600	\$	21,600	\$	0	\$	21,600
All current directors and										
executive officers as a										
group (8 persons)	\$	809,659	\$	464,400	\$	1,274,059	\$	115,200	\$	1,369,259

Information About Golden Parachute Compensation

We have certain plans and agreements that provide for amounts to be paid to our named executive officers or for the accelerated vesting of equity awards held by our named executive officers in connection with a change in control. The merger will constitute a change in control under each of these plans and agreements. We describe each of these plans and agreements below and quantify the amounts that will be payable to our named executive officers and the value of any equity awards held by our named executive officers for which vesting will be accelerated in connection with the merger, assuming the merger is completed and assuming that the employment of such named executive officers with us is terminated upon the consummation of the merger.

Equity Incentive Plans. In the event of a change in control, our 2011 Incentive Compensation Plan provides that our board of directors, or compensation committee, has the power to prescribe the terms for any outstanding awards under such plan. Under the terms of the Merger

Agreement, all stock options, whether vested or not, including options held by our named executive officers, that are outstanding and unexercised immediately prior to the effective time of the merger will be canceled and automatically converted into the right to receive an amount in cash, without interest, equal to the product of (i) the total number of shares of Common Stock otherwise issuable upon exercise of such option and (ii) the Merger Consideration less the exercise price per share of the stock option and less applicable withholding taxes.

Management Agreements.

Management Agreement with Mr. Simpson, President/CEO. In the event of a change of control of our company, such as this merger, and if Mr. Simpson is terminated within twenty four (24) months of the merger without cause or terminated within eighteen (18) of the merger for good reason, Mr. Simpson is entitled to receive

Table of Contents

an amount equal to 24 months of his then current monthly compensation. In addition, in the event of a change of control of our company, all awards due to Mr. Simpson under any stock incentive plan are automatically accelerated and fully exercisable with respect to the total number of shares subject to such stock option and may be exercised for any or all of those shares as fully-vested shares of Common Stock as of the date of such occurrence, without regard to the conditions expressed in the agreements relating to such stock option. For a description of how stock options are to be treated under the Merger Agreement see Equity Incentive Plans above. If Mr. Simpson elects to continue group health insurance coverage under COBRA, following a termination upon a change in control (as defined in the Management Agreement), we must pay the difference between (a) the monthly group health insurance premium Mr. Simpson was paying immediately prior to such termination and (b) the total monthly COBRA premium in relation to the continuation of Mr. Simpson s group health insurance coverage only. Payments will be made for a period of twenty-four (24) months.

Mr. Simpson is bound by a covenant not to compete during his employment and for a period of 18 months thereafter. Mr. Simpson is also bound by a confidentiality requirement, wherein he is required to keep our information and trade secrets confidential during his employment and thereafter.

Management Agreement with Mr. Critchfield, Senior Vice President Finance and Administration and CFQ. In the event of a change of control of our company, such as this merger, and if Mr. Critchfield is terminated within twenty four (24) months of the merger without cause or terminated within eighteen (18) of the merger for good reason, Mr. Critchfield is entitled to receive an amount equal to 24 months of his then current monthly compensation. In addition, in the event of a change of control of our company, all awards due to Mr. Critchfield under any stock incentive plan shall automatically accelerate and be fully exercisable with respect to the total number of shares subject to such stock option and may be exercised for any or all of those shares as fully-vested shares of Common Stock as of the date of such occurrence, without regard to the conditions expressed in the agreements relating to such stock option. For a description of how stock options are to be treated under the Merger Agreement see Equity Incentive Plans above. If Mr. Critchfield elects to continue group health insurance coverage under COBRA, following a termination upon a change in control (as defined in the Management Agreement), we must pay the difference between (a) the monthly group health insurance premium Mr. Critchfield was paying immediately prior to such termination and (b) the total monthly COBRA premium in relation to the continuation of Mr. Critchfield s group health insurance coverage only. Payments will be made for a period of twenty-four (24) months.

Mr. Critchfield is bound by a covenant not to compete during his employment and for a period of 18 months thereafter. Mr. Critchfield is also bound by a confidentiality requirement, wherein he is required to keep our information and trade secrets confidential during his employment and thereafter.

Management Agreement with Ms. Steinbarger, Senior Vice President Sales and Marketing. In the event of a change of control of our company, such as this merger, and if Ms. Steinbarger is terminated within twenty four (24) months of the merger without cause or terminated within eighteen (18) of the merger for good reason, Ms. Steinbarger is entitled to receive an amount equal to 24 months of his then current monthly compensation. In addition, should Ms. Steinbarger be terminated for any reason other than for cause, or in the event of a change of control of our company, all awards due to Ms. Steinbarger under any stock incentive plan shall automatically accelerate and be fully exercisable with respect to the total number of shares subject to such stock option and may be exercised for any or all of those shares as fully-vested shares of Common Stock as of the date of such occurrence, without regard to the conditions expressed in the agreements relating to such stock option For a description of how stock options are to be treated under the Merger Agreement see Equity Incentive Plans above. If Ms. Steinbarger elects to continue group health insurance coverage under COBRA, following a termination upon a change in control (as defined in the Management Agreement), we must pay the difference between (a) the monthly group health insurance premium Ms. Steinbarger was paying immediately prior to such termination and (b) the total monthly COBRA premium in relation to the continuation of Ms. Steinbarger s group health insurance coverage only. Payments will be made for a period of twenty-four (24) months.

Ms. Steinbarger is bound by a covenant not to compete during her employment and for a period of 18 months thereafter. Ms. Steinbarger is also bound by a confidentiality requirement, wherein she is required to keep our information and trade secrets confidential during her employment and thereafter.

Table of Contents

Aggregate Amounts of Potential Compensation. The following table provides information regarding the aggregate dollar value of the various elements of compensation that could be received by the named executive officers that is based on or otherwise relates to the merger. The descriptions and quantifications of the payments in the table are intended to comply with Item 402(t) of Regulation S-K promulgated by the SEC, which requires disclosure of information about compensation and benefits that each of the named executive officers will or may receive in connection with the merger. This compensation is referred to as golden parachute compensation by applicable SEC disclosure rules and is subject to a non-binding, advisory vote of our shareholders, as described below in the section entitled Compensation Arrangements for our Named Executive Officers in Connection with the Merger beginning on page 68 of this proxy statement, In preparing the table, we have made the following assumptions:

- the merger will close on the date of our special meeting; and
- Mr. Simpson, Mr. Critchfield and Ms. Steinbarger s employment will be terminated by us without cause or for good reason (as that term is used in his or her management agreement), effective as of the effective time of the merger.

In addition to the above assumptions, the costs of COBRA premiums for continued medical insurance coverage for our executive officers and their dependents and the calculation of potential tax reimbursements to our named executive officers are based on estimates that we believe to be reasonable. Any changes in these estimates would affect the amounts shown in the following table. Although Buyer has expressed its present intention to terminate Mr. Simpson s and Mr. Critchfield s employment in connection with the merger, it has no obligation to do so, and any delay in the termination of Mr. Simpson s or Mr. Critchfield s employment could affect the amount of any golden parachute compensation each would receive. Buyer has expressed no intention to terminate Ms. Steinbarger.

Golden Parachute Compensation

Name	Cash (\$)	Pension/ NQDC (\$)(1)		Perquisites/ Benefit	ts (\$)	Tax Reimbursement (S	S)	Other (\$)	Total (\$)
Douglass T.										
Simpson	\$ 480,000	\$	0	\$ 3	3,163	\$	0	\$	0 \$	567,563
William H.										
Critchfield	\$ 438,000	\$	0	\$ 3	3,163	\$	0	\$	0 \$	507,163
Ann L. Steinbarger	\$ 364,000	\$	0	\$ 3	4,742	\$	0	\$	0 \$	427,542

(1) We do not have any pension plans or nonqualified deferred compensation benefit arrangements.

Director and Officer Indemnification and Insurance

For a period of six years after the effective time of the merger, Buyer and the surviving corporation are required to fulfill and honor in all respects all of our obligations to the same extent required under Nevada law, our Articles and Bylaws and under any indemnification or other similar agreements between us and our current and former directors, officers and employees (whom we refer to as indemnified parties) in effect

on the date of the Merger Agreement.

Prior to the effective time of the merger, we or the surviving corporation will purchase one or more six-year tail policies on our directors and officers liability insurance policies in effect on the date of the Merger Agreement or comparable coverage in respect of actions or omissions occurring prior to the effective time of the merger. The cost in obtaining these policies may not exceed 300% of our current annual premium for such insurance. The surviving corporation will (and Buyer will cause the surviving corporation to) maintain such tail policies in full force and effect and continue to honor their respective obligations under the policies for a period of six years after the effective time of the merger.

If Buyer, the surviving corporation or any of their respective successors or assigns consolidates with or merges into any other person and is not the continuing or surviving entity of such consolidation or merger, or transfers or conveys all or substantially all of its properties and assets to any person, then the Merger Agreement requires that proper provision be made so that the successors and assigns of Buyer or the surviving corporation will

Table of Contents

assume all of the applicable obligations described above. The indemnified parties are intended third-party beneficiaries of the indemnification and insurance provisions in the Merger Agreement.

Interests of our Largest Shareholder in the Merger

In considering the recommendation of our board of directors with respect to the Merger Agreement and the merger, you should be aware that our largest shareholder has interests in the merger that are in addition to, or different from, the interests of our shareholders generally and that create potential conflicts of interest. These interests are described below. Buyer and ELITech are currently discussing the future distribution relationship between ELITech and us. The First Amended and Restated Joint Product Development Agreement, by and between us, Financiere ELITech SAS, and Wescor, Inc., dated July 28, 2011, has expired by its terms, and Buyer and ELITech are currently negotiating an agreement to clarify certain post-termination obligations pursuant to such agreement.

REGULATORY MATTERS

Mergers and acquisitions that may have an impact in the United States are subject to review by the Department of Justice and the Federal Trade Commission to determine whether they comply with applicable antitrust laws. Under the HSR Act, mergers and acquisitions that meet certain jurisdictional thresholds may not be completed until the expiration of a waiting period that follows the filing of notification forms by both parties to the transaction with the Department of Justice and the Federal Trade Commission. We and Buyer have determined that the merger does not meet the jurisdictional thresholds of the HSR Act. As a result, neither Buyer nor we are required to file notification forms with the Department of Justice or the Federal Trade Commission, and no waiting period will apply to the merger as a consequence of the HSR Act. See The Merger Agreement Conditions to the Merger beginning on page 60 of this proxy statement.

LITIGATION RELATED TO THE MERGER

We are aware of one purported class action complaint that has been filed in the Second Judicial District Court of the State of Nevada in connection with the merger: Rauenzhan v. Corgenix, et al., No. CV14-01907, filed on September 4, 2014. The complaint names as defendants us, each member of our board of directors, Buyer and Merger Sub. The complaint alleges that the board of directors breached its fiduciary duties and that we, Buyer, the Merger Sub aided and abetted those purported breaches, in connection with the proposed merger. The complaint challenges the Merger Consideration as inadequate, and makes a variety of other allegations, including the following:

- given the recent trading price of our Common Stock and potential future growth, the value of our Common Stock is greater than the consideration offered to shareholders in the proposed merger;
- the proposed merger is the result of a flawed process marred by conflicts of interest of our board and senior management; and

• the no solicitation and termination fee provisions of the Merger Agreement preclude us from soliciting, and otherwise restrict our ability to consider, competing offers.

The plaintiff in this case seeks an order certifying a proposed class of our shareholders, certifying the plaintiff as the class representatives, granting injunctive relief against the consummation of the merger, or, if the merger is consummated, rescinding the merger and awarding damages, directing the defendants to account for all damages caused by them and all profits or special benefits obtained by them as a result of their alleged breaches of fiduciary duties and an award of costs, expenses and reasonable attorneys fees, and accountants and experts fees.

We notified our carriers of our applicable insurance policies (the D&O Insurance) regarding these claims. The D&O Insurance provides coverage for defense costs incurred in connection with the claim, but is subject to a substantial deductible, depending on the claim. The D&O Insurance may not cover any amount of any judgment or settlement representing the amount by which the Merger Consideration or price is increased.

The defendants believe the claims asserted are without merit. Buyer has repeatedly indicated that the Merger Consideration is the highest amount Buyer is willing to pay to acquire our company and that it is not willing to increase the amount of the Merger Consideration.

Table of Contents

MATERIAL UNITED STATES FEDERAL INCOME TAX CONSEQUENCES

The following summary is a general discussion of the material United States federal income tax consequences to our shareholders whose shares of our Common Stock are converted into cash pursuant to the merger. This summary is based upon the provisions of the Internal Revenue Code of 1986, as amended, or the Code, applicable United States Treasury Regulations, judicial authority, and administrative rulings and practice, all as in effect on the date of this proxy statement. All of these authorities are subject to change, possibly on a retroactive basis. No ruling from the Internal Revenue Service has been or will be sought with respect to any aspect of the merger.

This summary is for the general information of our shareholders only and does not purport to be a complete analysis of all potential tax effects of the merger. It addresses only shareholders who hold our Common Stock as a capital asset within the meaning of Section 1221 of the Code (generally, property held for investment), and does not address any tax consequences to holders of stock options or holders of restricted stock. The summary does not address any state, local or foreign income tax consequences of the merger, nor does it address any non-income tax laws.

the sharehold	y does not address all aspects of United States federal income taxation that may be relevant to a particular shareholder in light of er s personal investment circumstances. The summary also does not address all aspects of United States federal income taxation elevant to particular shareholders that are subject to special treatment under United States federal income tax laws, including, ation:
• de	ealers, brokers and traders in securities;
• tax	x-exempt entities;
• fin	nancial institutions, mutual funds, regulated investment companies, real estate investment trusts or insurance companies;
• pa partnerships;	urtnerships (or entities or arrangements treated as partnerships for United States federal income tax purposes) and investors in such
• co	ontrolled foreign corporations or passive foreign investment companies;
• n	non-U.S. holders (as defined below) that own, or have owned, actually or constructively, more than 5% of our Common Stock;
	areholders who acquired their shares of our Common Stock in connection with stock option or stock purchase plans or in other

compensation arrangements;

•	shareholders who receive cash pursuant to the exercise of dissenter s rights;
• straddle o	shareholders who hold their shares of our Common Stock as part of an integrated investment such as a hedge or as part of a hedging or other risk reduction strategy;
•	shareholders who own, directly or indirectly, an interest in Buyer;
•	certain United States expatriates or former long-term residents of the United States;
•	shareholders subject to the alternative minimum tax provisions of the Code;
• Code; or	shareholders who hold Common Stock that constitutes qualified small business stock within the meaning of Section 1202 of the
•	holders who have a functional currency other than the United States dollar.
	ership (or an entity or arrangement treated as a partnership for United States federal income tax purposes) holds shares of our Common tax treatment of a partner generally will depend on the status of
	45

Table of Contents

the partner and the activities of the partnership.	Partners of partnerships holding our Common Stock should consult their own tax advisors
regarding the tax consequences of the merger.	

For purposes of this summary, U.S. holder refers to a beneficial owner of our Common Stock that is, for United States federal income tax purposes:

- an individual citizen or resident of the United States;
- a corporation, or other entity treated as a corporation for United States federal income tax purposes, created or organized in or under the laws of the United States or any state of the United States or the District of Columbia;
- a trust (1) the administration of which is subject to the primary supervision of a United States court and one or more United States persons have the authority to control all substantial decisions of the trust or (2) that has a valid election in effect under applicable United States Treasury Regulations to be treated as a United States person; or
- an estate that is subject to United States federal income tax on its income regardless of its source.

A non-U.S. holder is a beneficial owner of our Common Stock that is an individual, corporation, trust or estate that, in each case, is not a U.S. holder.

Shareholders are advised to consult their own tax advisors regarding the United States federal income tax consequences of the merger in light of their personal circumstances, as well as any tax consequences arising under any federal non-income tax laws and under any state, local and foreign tax laws.

U.S. Holders

The receipt of cash in exchange for shares of our Common Stock pursuant to the merger will be a taxable transaction to U.S. holders. Generally, this means that each U.S. holder will recognize capital gain or loss equal to the difference between the amount of cash received by such U.S. holder in the merger and the U.S. holder is adjusted tax basis in its shares of our Common Stock. Gain or loss will be determined separately for each block of shares (i.e., shares acquired at the same cost in a single transaction) owned by a U.S. holder. If the shares of our Common Stock have been held for more than one year at the effective time of the merger, the gain or loss will be long-term capital gain or loss, and will be short-term capital gain or loss if the shares have been held for one year or less. For non-corporate taxpayers, long-term capital gains generally are taxable at a reduced rate and short-term capital gains are taxed at ordinary income tax rates. The deductibility of capital losses is subject to certain limitations.

Certain individuals, estates and trusts must pay a 3.8% tax on net investment income for taxable years beginning after December 31, 2012. Net investment income is defined to include capital gains. Thus, a U.S. holder s capital gain resulting from the merger will likely be subject to the 3.8% tax on net investment income.

Information returns will be filed with the Internal Revenue Service in connection with payments to a U.S. holder pursuant to the merger, unless the U.S. holder is an exempt recipient. Under the United States federal income tax backup withholding rules, the exchange agent to be appointed by Buyer for the purpose of paying the Merger Consideration generally will be required to withhold 28% of all payments to which a U.S. holder is entitled in the merger, unless the U.S. holder (i) is a corporation or comes within other exempt categories and demonstrates this fact in a manner satisfactory to the exchange agent or (ii) provides its correct tax identification number (social security number, in the case of an individual, or employer identification number in the case of other shareholders), certifies under penalties of perjury that the number is correct, certifies as to no loss of exemption from backup withholding and otherwise complies with the applicable requirements of the backup withholding rules. To avoid such backup withholding, each U.S. holder and, if applicable, each other payee, should complete, sign and return to the exchange agent the substitute Form W-9 that such U.S. holder will receive with the letter of transmittal following completion of the merger. Backup withholding is not an additional tax. Any amounts withheld under the backup withholding rules will be credited against a U.S. holder s United States federal income tax liability, and the

7D 1	1		-	_			
Tal	าเ	e.	Ot	()	Ωn	ter	1fs

U.S. holder may obtain a refund of any amounts withheld in excess of its United States federal income tax liability, provided that the required information is timely furnished to the Internal Revenue Service.

Non-U.S. Holders

Any gain realized on the receipt of cash pursuant to the merger by a non-U.S. holder generally will not be subject to United States federal income tax unless:

- the gain is effectively connected with the conduct by the non-U.S. holder of a trade or business in the United States (and, if an applicable income tax treaty so provides, is also attributable to the non-U.S. holder s permanent establishment in the United States), in which case the non-U.S. holder generally will be taxed at graduated United States federal income tax rates applicable to United States persons (as defined under the Code) and, if the non-U.S. holder is a foreign corporation, an additional branch profits tax may apply to its effectively connected earnings and profits for the taxable year at the rate of 30% (or such lower rate as may be specified by an applicable income tax treaty); or
- the non-U.S. holder is a nonresident alien individual who is present in the United States for 183 days or more in the taxable year of the merger and certain other conditions are met, in which case the non-U.S. holder will be subject to a 30% tax on gain realized in the merger (unless an exception is provided under an applicable treaty), which may be offset by U.S. source capital losses of the non-U.S. holder recognized during the year, if any.

A non-U.S. holder will be subject to information reporting and, in certain circumstances, backup withholding (at a rate of 28%) may apply to the cash received pursuant to the merger, unless the non-U.S. holder certifies under penalties of perjury that it is a non-U.S. holder. To avoid backup withholding, non-U.S. holders generally must submit a completed and signed Form W-8BEN, Certificate of Foreign Status of Beneficial Owner for United States Tax Withholding, or other applicable Form W-8. Backup withholding is not an additional tax. Any amounts withheld under the backup withholding rules will be credited against a non-U.S. holder s United States federal income tax liability, and the non-U.S. holder may obtain a refund of any amounts withheld in excess of its United States federal income tax liability, provided that the required information is timely furnished to the Internal Revenue Service.

Option Holders

Generally, for U.S. federal income tax purposes, upon the cancellation of an option, issued under our 2011 Incentive Compensation Plan or our 2007 Incentive Compensation Plan, to purchase our Common Stock, the option holder will recognize as ordinary income all of the consideration attributable to the cancellation of the option. Such ordinary income will be taxed at ordinary income tax rates and will be treated as income received as compensation for services performed by the holder. Such ordinary income may be subject to withholding taxes and employment taxes and any other required employer withholding, if attributable to the holder s employment with us, or self-employment taxes, if attributable to services performed by the holder as an independent contractor. The consideration attributable to the cancellation of the option will generally be subject to information reporting, and payments to a non-employee option holder may be subject to backup withholding. Generally, the holder of such an option will not recognize loss for U.S. federal income tax purposes with respect to the cancellation of such option.

Warrant Holders

Generally, for U.S. federal income tax purposes, the holder of a warrant to purchase shares of our Common Stock will realize capital gain or loss equal to the excess of the consideration received on the cancellation of the warrant over the holder s adjusted tax basis in the warrant. Generally, if the holder has held the warrant for more than one year, such capital gain or loss will be long-term capital gain or loss. See *U.S. Holders* and *Non-U.S. Holders* for a more detailed discussion of the U.S. federal income tax taxation of capital gain realized by U.S. holders and non-U.S. holders, respectively.

THE FOREGOING DISCUSSION OF MATERIAL UNITED STATES FEDERAL INCOME TAX CONSEQUENCES IS NOT INTENDED TO BE, AND SHOULD NOT BE CONSTRUED AS, LEGAL OR TAX

47

Table of Contents

ADVICE TO ANY HOLDER OF SHARES OF OUR COMMON STOCK. EACH OF OUR SHAREHOLDERS IS STRONGLY URGED TO CONSULT HIS, HER OR ITS TAX ADVISOR TO DETERMINE THE PARTICULAR TAX CONSEQUENCES (INCLUDING THE APPLICATION AND EFFECT OF ANY STATE, LOCAL OR FOREIGN INCOME AND OTHER TAX LAWS) TO SUCH SHAREHOLDER OF THE RECEIPT OF CASH IN EXCHANGE FOR SHARES OF OUR COMMON STOCK PURSUANT TO THE MERGER.

Table of Contents

THE MERGER AGREEMENT

The following is a summary of the material terms and conditions of the Merger Agreement. The description in this section and elsewhere in this proxy statement is qualified in its entirety by reference to the complete text of the Agreement and Plan of Merger, dated as of August 27, 2014 a copy of which is attached as Annex A and is incorporated by reference into this proxy statement. This summary does not purport to be complete and may not contain all of the information about the Merger Agreement that is important to you. We encourage you to read the Merger Agreement carefully and in its entirety because it is the legal document that governs this merger.

Explanatory Note Regarding the Merger Agreement

The Merger Agreement and this summary of its terms and provisions have been included to provide you with information regarding the terms and provisions of the Merger Agreement. Factual disclosures about us, Buyer or the Merger Sub contained in this proxy statement or in our public reports filed with the SEC, which are available without charge at www.sec.gov, may supplement, update or modify the factual disclosures about us, Buyer or the Merger Sub contained in the Merger Agreement. In your review of the representations and warranties contained in the Merger Agreement and described in this summary, it is important to bear in mind that the representations and warranties have been negotiated with the principal purpose of establishing the circumstances in which a party to the Merger Agreement may have the right not to close the merger if the representations and warranties of the other party prove to be untrue due to a change in circumstance or otherwise, and allocates risk between the parties to the Merger Agreement, rather than establishing matters as facts. The representations and warranties may also be subject to a contractual standard of materiality different from those generally applicable to shareholders and in some cases have been qualified in a number of important respects, including through the use of negotiated exceptions for certain matters disclosed by each party to the other parties.

The Merger

The Merger Agreement provides that, in accordance with the Merger Agreement and the Nevada Revised Statutes, as amended, which we refer to as the NRS, at the effective time of the merger, the Merger Sub will be merged with and into us and, as a result of the merger, the separate existence of the Merger Sub will cease and we will continue as the surviving corporation and become a wholly owned subsidiary of Buyer. At the effective time of the merger, all of our and the Merger Sub s properties, rights, privileges, powers and franchises will vest in us as the surviving corporation and all of our and the Merger Sub s debts, liabilities and duties will become our debts, liabilities and duties as the surviving corporation.

The articles of incorporation and the bylaws of the Merger Sub, as in effect immediately prior to the effective time of the merger, will be the articles of incorporation and bylaws of the surviving corporation until amended as provided by in such documents or under applicable law. The directors of the Merger Sub immediately prior to the effective time of the merger will be the initial directors of the surviving corporation following the effective time of the merger until the earliest of their resignation, removal or death or their respective successors are duly elected or appointed and qualified, as the case may be. The officers of the surviving corporation immediately prior to the effective time of the merger will be the officers of the surviving corporation following the effective time of the merger until the earliest of their resignation, removal or death or their respective successors are duly elected or appointed and qualified, as the case may be.

The closing of the merger will occur within five business days following the satisfaction or, to the extent permitted by law, waiver of the closing conditions stated in the Merger Agreement and described under Conditions to the Merger beginning on page 60 of this proxy statement (other than those conditions that by their nature are to be satisfied at the closing, but subject to the fulfillment or, to the extent permitted by law, waiver of such conditions) or as otherwise agreed by the parties.

At the closing of the merger, the parties will cause the merger to be consummated by filing articles of merger with the Secretary of State of the State of Nevada. The merger will be effective at the time the certificate of merger is duly filed with the Secretary of State of the State of Nevada (or at such later time as is agreed upon by the parties to the Merger Agreement and specified in the certificate of merger), which we refer to as the effective time of the merger. It is currently anticipated that the effective time of the merger will occur in the fourth calendar quarter

Table of Contents

of 2014. However, the parties cannot predict the exact timing of the closing of the merger or whether the merger will be completed at an earlier or later time, as agreed by the parties, or at all.

The Merger Consideration

At the effective time of the merger, by virtue of the merger and without any action on the part of Buyer, the Merger Sub, us or any holder of shares of our Common Stock, each share of our Common Stock issued and outstanding immediately prior to the effective time of the merger will be canceled and converted automatically into the right to receive \$0.27 in cash, without interest, which we refer to as the Merger Consideration, upon surrender of the shares in accordance with the Merger Agreement, other than:

- shares held by shareholders who have not voted to adopt the Merger Agreement and who have properly assert dissenter s rights for shares in accordance with Sections 92A.300 through 92A.500 of the NRS, which shares will be canceled and converted into the right to receive payment of the fair value of such shares, unless such dissenter s rights are subsequently waived, withdrawn or lost, at which point such shares will represent the right to receive the Merger Consideration; and
- shares held by us as treasury stock or owned by Buyer, or the Merger Sub immediately prior to the effective time of the merger, which shares will be automatically cancelled and cease to exist without payment.

At the effective time of the merger, each share of Common Stock of the Merger Sub outstanding immediately prior to the effective time of the merger will be converted into and become one fully paid and non-assessable share of Common Stock, no par value per share, of the surviving corporation.

Payment Procedures

Prior to the effective time of the merger, Buyer will appoint an exchange agent reasonably acceptable to us to act as the agent, which shall provide for the payment of the Merger Consideration. As of the effective time of the merger, Buyer is required to deposit, or to take all steps necessary to enable and cause the surviving corporation to deposit, with the exchange agent the cash necessary to pay for the shares of our Common Stock converted into the right to receive Merger Consideration.

Promptly after the effective time, Buyer will mail, or cause the exchange agent to mail, to each holder of shares of our Common Stock that were converted into the right to receive the Merger Consideration, a letter of transmittal and instructions for use in surrendering stock certificates or transferring uncertificated shares in exchange for the Merger Consideration. **No shareholder should surrender any stock certificates until the shareholder receives the letter of transmittal and other materials for such surrender.** Upon surrender of a stock certificate representing shares of our Common Stock for cancellation to the exchange agent, together with a duly executed letter of transmittal and such other documents as may reasonably be required by the exchange agent, or receipt of an agent s message by the exchange agent (or such other evidence, if any, of transfer as the exchange agent may reasonably request), in the case of a book-entry transfer of uncertificated shares of our Common Stock, the

holder of such certificate or uncertificated shares will be entitled to receive the Merger Consideration into which the shares of our Common Stock previously represented by such stock certificate or uncertificated shares have been converted pursuant to the Merger Agreement. Each of Buyer, the Merger Sub, the surviving corporation, the exchange agent and us will be entitled to deduct and withhold from the Merger Consideration such amounts as may be required to be deducted or withheld under applicable tax laws. The cash paid to you upon surrender of your shares of our Common Stock will be issued in full satisfaction of all rights relating to your shares of our Common Stock.

If any portion of the Merger Consideration is to be paid to a person other than the person in whose name the surrendered stock certificate or the transferred uncertificated share is registered, it will be a condition to such delivery that such stock certificate will be properly endorsed or will otherwise be in proper form for transfer or such uncertificated share will be properly transferred and that the person requesting such payment will pay any transfer or other tax required as a result of such payment to a person other than the registered holder of such stock certificate or uncertificated share or establish to the satisfaction of Buyer that such taxes have been paid or are not applicable.

Table of Contents

After the effective time of the merger, there will be no further registration of transfers of shares of our Common Stock. If, after the effective time of the merger, stock certificates or uncertificated shares of our Common Stock are presented to the surviving corporation or the exchange agent, they will be canceled and exchanged for the Merger Consideration, except as otherwise provided by law.

If your stock certificate has been lost, stolen or destroyed, the exchange agent will deliver to you the applicable Merger Consideration for the shares represented by that certificate if:

- you make an affidavit (in a form customarily used by the exchange agent) claiming such certificate has been lost, stolen or destroyed; and
- if required by Buyer, you post a bond in such reasonable amount as Buyer may direct as indemnity against any claim that may be made against it with respect to that certificate.

You should not send your certificates now and should send them only pursuant to instructions set forth in the letter of transmittal to be mailed to shareholders promptly after the completion of the merger. In all cases, the Merger Consideration will be paid only in accordance with the procedures set forth in the Merger Agreement and such letters of transmittal.

The Merger Agreement provides that six months after the effective time of the merger, the exchange agent will deliver to Buyer upon its demand any funds made available to the exchange agent that have not been distributed to the former holders of shares of our Common Stock, and that any holders of stock certificates or uncertificated shares who have not complied with the above-described procedures to receive payment of the Merger Consideration during such twelve-month period may thereafter look only to Buyer for payment of the Merger Consideration, without interest, to which they are entitled. None of Buyer, the surviving corporation or the exchange agent will be liable for any amount delivered to a public official pursuant to any applicable abandoned property, escheat or similar law. If any stock certificates or uncertificated shares representing shares of our Common Stock prior to the effective time of the merger have not been surrendered by three years after the effective or such earlier date when the amounts would otherwise escheat to or become property of any governmental entity, then any Merger Consideration in respect of such stock certificates or uncertificated shares will become the property of Buyer free and clear of all claims or interests of any holders of such stock certificates or uncertificated shares to the extent permitted by law.

Treatment of Options and Warrants

We have issued options to purchase shares of our Common Stock under our 2011 Incentive Compensation Plan and our 2007 Incentive Compensation Plan (which, together, we refer to as the stock plans). Our board of directors has adopted resolutions providing that each stock option which has not been exercised and has not expired or terminated and remains outstanding immediately prior to the effective time of the merger will be canceled, and the holder of any such option will be entitled to receive an amount in cash, without interest, equal to the product of (i) the total number of shares of our Common Stock issuable upon exercise of such option and (ii) the excess, if any, of the Merger Consideration over the per share exercise price under such option, subject to any required withholding taxes. No holder of an option that has an exercise price per share that is greater than \$0.27 is entitled to any payment with respect to such canceled option before or after the effective time of the merger.

Our board of directors has also adopted resolutions providing that each warrant to purchase shares of Common Stock that is outstanding immediately prior to the effective time of the merger will be canceled, and the holder of any such warrant will be entitled to receive an amount in cash, without interest, equal to the product of (i) the total number of shares of our Common Stock previously subject to such warrant and (ii) the excess, if any, of the Merger Consideration over the per share exercise price previously subject to such warrant, subject to any required withholding taxes. No holder of a warrant that has an exercise price per share that is greater than \$0.27 is entitled to any payment with respect to such canceled warrant before or after the effective time of the merger.

51

Table of Contents

Representations and Warranties

The Merger Agreement contains representations and warranties of the parties to the Merger Agreement. The assertions embodied in those representations and warranties were made solely for purposes of the Merger Agreement and may be subject to important qualifications and limitations agreed to by the parties in connection with negotiating the terms of the Merger Agreement. Moreover, these representations and warranties have been qualified by certain disclosures that we made to Buyer and the Merger Sub in connection with the negotiation of the Merger Agreement, which disclosures are not reflected in the Merger Agreement. Furthermore, some of those representations and warranties may not be accurate or complete as of any particular date because they are subject to a contractual standard of materiality or material adverse effect different from that generally applicable to public disclosures to shareholders. The representations and warranties were used for the purpose of allocating risk between the parties to the Merger Agreement rather than establishing matters of fact. For the foregoing reasons, you should not rely on the representations and warranties contained in the Merger Agreement as statements of factual information. The representations and warranties in the Merger Agreement and the description of them in this proxy statement should be read in conjunction with the other information contained in our reports, statements and filings that we publicly file with the SEC. This description of the representations and warranties is included to provide our shareholders with additional information regarding the terms of the Merger Agreement.

In the Merger Agreement, we made representations and warranties to Buyer and the Merger Sub with respect to, among other things:

- corporate matters related to us, including, but not limited to, our valid existence, good standing, power and authority;
- our capitalization, including in particular the number of outstanding shares of our Common Stock and preferred stock and the number of shares of Common Stock issuable upon the exercise of stock options and warrants;
- our power and authority to enter into the Merger Agreement and to complete the transactions contemplated by the Merger Agreement and the enforceability of the Merger Agreement against us;
- the absence of violations of, conflicts with, creation of liens upon, or defaults under our governing documents, law or certain agreements as a result of entering into the Merger Agreement and the consummation of the merger;
- the required consents and approvals of governmental authorities in connection with the consummation of the merger and the other transactions contemplated by the Merger Agreement;
- our SEC filings since August 27, 2011, including financial statements contained in such SEC filings;

•	the absence of undisclosed material liabilities;
•	internal controls and compliance with the Sarbanes-Oxley Act of 2002;
	our conduct of business since March 31, 2014 in the ordinary course of business consistent with past practice, except as contemplated reger Agreement, and the absence since March 31, 2014 of any event, condition, change, occurrence or development that, individually ggregate, has had or would reasonably be expected to have a material adverse effect on us;
•	the absence of certain litigation;
•	the information included in this proxy statement;
•	broker s and finder s fees and expenses;
	52

Table of Contents

•	employee benefit plans, ERISA matters and certain related matters;
• proxy state	the receipt of the fairness opinion described in The Merger Opinion of Corgenix's Financial Advisor beginning on page 31 of thiement by our board of directors from Inverness;
•	tax matters;
•	environmental matters;
•	compliance with laws and permits, including compliance with the Food, Drug and Cosmetic Act;
•	intellectual property matters;
•	labor matters;
•	matters related to our insurance policies;
•	matters with respect to our material contracts;
•	title to properties and the absence of encumbrances;
•	related party transaction matters; and
•	the inapplicability of state anti-takeover statutes to the merger.

Many of the representations and warranties in the Merger Agreement made by us are qualified by a materiality or a company material adverse effect standard (that is, they will not be deemed to be untrue or incorrect unless their failure to be true or correct, individually or in the aggregate, would, as the case may be, be material or have a company material adverse effect on us). For purposes of the Merger Agreement, a company material adverse effect means a material adverse effect on (i) our business, financial condition, assets, liabilities or results of operations, or (ii) our ability to perform our obligations under the Merger Agreement or to consummate the merger and the other transactions contemplated by the Merger Agreement. However, the definition of company material adverse effect excludes any effect attributable to:

any other geograp	nange generally affecting the economy, financial markets or political, economic or regulatory conditions in the United States or othic region in which we conduct business (except to the extent that we are affected in a disproportionate manner as compared to participating in the same industry as ours);
	al financial, credit or capital market conditions, including interest rates or exchange rates, or any changes to the same business ent that we are affected in a disproportionate manner as compared to other companies participating in the same industry as
	nange attributable to the negotiation, execution, announcement, pendency or pursuit of the merger and related transactions, gation, any cancellation of or delays in customer orders, and any disruption in supplier, distributor, partner or similar
	king of any action by us as required by the Merger Agreement or expressly approved or permitted in writing by Buyer, or the vaction by us prohibited by the Merger Agreement;
	f war (whether or not declared), the commencement, continuation or escalation of a war, acts of armed hostility, sabotage or international or national calamity or any material worsening of such conditions threatened or existing as of the date of the nt;
• any hu	arricane, earthquake, flood, or other natural disasters or acts of God;
	53

to

Table of Contents

	nanges in law (except to the extent that we are affected in a disproportionate manner as compared to other companies participating adustry as ours);
	nanges in GAAP (except to the extent that we are affected in a disproportionate manner as compared to other companies in the same industry as ours);
	by failure by us to meet public or internal revenue, earnings or other projections (but not any event, condition, change, occurrence or of a state of circumstances that may have caused or contributed to the failure to meet such projections); and
	celine in the price of our Common Stock (but not any event, condition, change, occurrence or development of a state of s that may have caused or contributed to the change in price);
In the Merger	Agreement, Buyer and the Merger Sub made representations and warranties to us with respect to, among other things:
• co	orporate matters related to them, including, but not limited to, their valid existence, good standing, power and authority;
• the	eir formation, business purpose and conduct of operations;
	eir authority to enter into the Merger Agreement and to complete the transactions contemplated by the Merger Agreement and the of the Merger Agreement against each of them;
	e absence of violations of, conflicts with, creation of liens upon, or defaults under their governing documents, law or certain s a result of entering into the Merger Agreement and the consummation of the merger;
	e required consents and approvals of governmental authorities in connection with the consummation of the merger and the other contemplated by the Merger Agreement;
• br	oker s and finder s fees and expenses;

the absence of certain litigation;
 the availability of the funds necessary to consummate the merger and the other transactions contemplated by the Merger Agreement; and
 the solvency of the Buyer after the effective time of the merger.

The representations and warranties contained in the Merger Agreement and in any certificate or other writing delivered pursuant to the Merger Agreement will not survive the effective time of the merger. This limit does not apply to any covenant or agreement of the parties which by its

terms contemplates performance after the effective time of the merger. The confidentiality provisions will survive termination of the Merger

Conduct of Our Business Pending the Merger

Agreement.

The Merger Agreement contains covenants related to the conduct of our business from the date of the Merger Agreement until the effective time of the merger. These covenants are complicated and not easily summarized. You are urged to read carefully and in its entirety Section 5.1 of the Merger Agreement entitled Conduct of Business Prior to the Effective Time and Section 3.8, which is referenced therein. Subject to certain exceptions in the Merger Agreement and disclosure schedules delivered by us in connection with the Merger Agreement, between the date of the Merger Agreement and the effective time of the merger, unless Buyer gives its prior written consent (which cannot be unreasonably withheld, conditioned or delayed), we have agreed to (i) conduct our business in the ordinary and course of business and (ii) use our reasonable efforts to maintain and

Table of Contents

sell, assign, lease, transfer or otherwise dispose of, or agree to sell, assign, lease, transfer or otherwise dispose of, any of our tangible

assets other than sales of inventory in the ordinary course of business;

other than in the ordinary course of business, sell, assign, license, transfer, or otherwise dispose of, or agree to sell, assign, license, transfer or otherwise dispose of, any of its patents, inventions, shop rights, know-how, trade secrets, confidential information, registered or unregistered trademarks, service marks, logos, corporate names, trade names, and other trademark rights, trade dress, or other designations or combinations of such designations that are distinctive of our goods or services and that are used by us in a manner that identifies our goods or services and distinguishes them from the goods or services of others, works of authorship and any registered or unregistered copyright therein other intangible assets, and all registrations for, and applications for registration of, any of the foregoing or disclose any of the foregoing (other than to Buyer and Merger Sub);	
• enter into any transaction, contract, or commitment other than in the ordinary course of business;	
• make any capital expenditures or any commitment therefor in excess of \$50,000 in the aggregate;	
adopt or make any change in any executive compensation plan, bonus plan, incentive compensation plan, deferred compensation agreement, or other employee benefit plan or arrangement;	
55	

Table of Contents

pursuant to ar	ter into any employment or consulting agreement or arrangement, or, except for normal bonuses or wage or salary increases and consistent with existing plans or programs and consistent with past practices, grant or pay any bonus, or make or grant any or salary increase or any specific increase in the wages or salary of any employee;
	ncel or compromise any debt or claim except for adjustments made in the ordinary course of business that, in the aggregate, are no vaived or released any rights that are material;
• ter	rminate, amend, or modify any material agreement or instrument;
	ter into any transaction with any shareholder, officer, director, or key employee or any affiliate of any such person other than the ages and salaries and other benefits under employee benefit plans in existence as of the date of the Merger Agreement;
	ake any loans or advances to, guaranties for the benefit of, or investments in, any person (other than customary wage or travel ccordance with past practices which are less than \$10,000 in the aggregate);
• mo	erge or consolidate with, or acquire all or substantially all of the assets, capital stock, or business of any other person;
	ke any action (including the incurrence of any liabilities of any nature, whether or not accrued, contingent or otherwise) that would expected to have, individually or in the aggregate, a Company Material Adverse Effect; or
• ag	ree or commit to do any of the actions described in the foregoing.
Preparation	of Proxy Statement; Shareholders Meeting
As soon as re-	asonably practicable after the date of the Merger Agreement, we will:
	cooperation with Buyer and the Merger Sub, prepare and file with the SEC a preliminary proxy statement and respond to any

Unless the Merger Agreement is terminated as described below under Termination of the M	Ierger Agreement beginning on page 62 of this proxy
statement, as promptly as practicable following the clearance of the definitive proxy statement	by the SEC, we will:
 duly call, give notice of, convene and hold a special meeting of our shareholders (with purpose of seeking the affirmative vote for adoption of the Merger Agreement by the holders of outstanding on the record date; 	
mail to our shareholders the definitive proxy statement and all other proxy materials	s for the special meeting; and
 use our reasonable efforts to solicit proxies from our shareholders in favor of the adevote of our shareholders required by law to effect the merger. 	option of the Merger Agreement and secure the
56	

Tabl	le d	of (ากท	tents
1 au	ı v	лι	اللال	wiits

Public Statements

Unless our board of directors has effected a change in recommendation, we, Buyer and the Merger Sub will not issue any public announcement or statement or other disclosure with respect to the Merger Agreement or the merger without the prior written consent of the other party (which consent will not be unreasonably withheld or delayed), except as may be required by law or any rule of the SEC so long as the disclosing party in which case the party required to make the release or announcement will consult the other prior to issuing, and allow reasonable time with the opportunity to review and comment on such prior to issuance.

Standard of Efforts; Required Approvals

We, Buyer and the Merger Sub agreed to use our reasonable efforts to take, or cause to be taken, all action, and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or appropriate to consummate and make effective, the merger and the other transactions contemplated by the Merger Agreement, including:

- determining what filings, if any, are required and which consents are required; and
- timely obtaining all consents, approvals, authorizations and actions or non-actions required for or in connection with the consummation of the merger and the other transactions contemplated by the Merger Agreement.

Notification of Certain Matters

We will give prompt notice to Buyer and the Merger Sub of any written notice received from any person alleging that the consent of such person is required in connection with the merger or the other transactions contemplated by the Merger Agreement, any notice from any governmental authority in connection with the merger or the other transactions contemplated by the Merger Agreement, any actions pending that relate to the merger or the other transactions contemplated by the Merger Agreement, any event, change or effect which may cause the failure of the conditions to close the Merger Agreement not be satisfied.

Confidentiality

We entered into a confidentiality agreement dated May 6, 2014, with Water Street, on behalf of Orgentec, which we refer to as the confidentiality agreement. Under the terms of the confidentiality agreement, Water Street agreed that, subject to certain exceptions, any information regarding us or any of our affiliates or any of our respective assets or businesses furnished to Water Street by us or our representatives would be used by Water Street and its representatives solely for the purpose of evaluating a possible transaction between Water Street and its affiliates and us and would be kept confidential except as provided in the confidentiality agreement. The parties also agreed in the

Merger Agreement to keep certain information regarding the other party confidential until the closing. In the event of a termination of the Merger Agreement, the obligations of confidentiality will continue for a period of five years.

TAT T	0 1			
No	20	ıcı	tatı	on

We agreed not to, and agreed not to authorize or permit any of our officers, directors or employees, or authorize any of our financial advisors, attorneys or other advisors or representatives to:

- solicit, initiate or knowingly take any action or encourage the submission of any takeover proposal or making of any proposal that could reasonably expect to lead to a takeover proposal;
- participate or engage in any discussions or negotiations with any person or group other than Buyer, the Merger Sub regarding any takeover proposal, or furnish any non-public information relating to us to any third party that we believe would be used for purposes of making or furthering any takeover proposal;

57

Table of Contents

• amend or grant any waiver, amendment or release under any standstill or similar agreement with a third party; or
• enter into any agreement in principle, letter of intent, term sheet, acquisition agreement, merger agreement option agreement, joint venture agreement, partnership agreement or other similar agreement relating to a takeover proposal.
Notwithstanding the restrictions described above, at any time before the adoption of the Merger Agreement by our shareholders, if we receive an unsolicited written takeover proposal from a third party that did not result from a breach of the provisions of the Merger Agreement regarding restrictions on solicitations, we may provide or furnish information to the third party making the takeover proposal pursuant to an executed confidentiality agreement so long as our board of directors determines in good faith, after consultation with our financial advisor and outside counsel, that the takeover proposal is, or could reasonably be expected to result in or lead to, a superior proposal or take any action that any court of competent jurisdiction order us to take.
We must promptly notify Buyer after receipt of any takeover proposal, identify the third party making the proposal and provide a summary of the material terms and conditions of such proposal.
Change in Recommendation
Subject to the provisions described below, our board of directors unanimously agreed to recommend that our shareholders vote in favor of the adoption of the Merger Agreement at the special meeting, which we refer to as the board recommendation. Subject to the provisions described below, the Merger Agreement provides that, after the date of the Merger Agreement, neither our board of directors nor any committee of our board of directors will:
• withhold, change, amend, withdraw, or qualify (or modify in a manner adverse to Buyer) the board recommendation;
• adopt, approve or recommend or otherwise declare advisable the adoption of any takeover proposal;
• publicly propose to adopt, approve, or recommend any takeover proposal or publicly take a neutral position or no position with respect to a takeover proposal;
• submit any takeover proposal to the vote of our shareholders; or

• resolve or agree to take any of the foregoing actions.
We refer to each of the foregoing actions as a change in recommendation.
Subject to the provisions described below, the Merger Agreement also provides that, after the date of the Merger Agreement, neither our board of directors nor any committee of our board of directors will cause or permit us to enter into any agreement or instrument relating to any takeover proposal.
Notwithstanding these restrictions, at any time before the adoption of the Merger Agreement by our shareholders, our board of directors may effect a change in recommendation either:
• in response to a material, event change or development occurring or arising after the date of the Merger Agreement, the existence and material consequences of which were not known or not reasonably foreseeable by our board of directors at or prior to the date of the Merger Agreement (and not relating to any takeover proposal or superior proposal), so long as:
• our board of directors reasonably determines in good faith, after consultation with our outside counsel and independent financial advisor, that the failure to make such change in recommendation would reasonably be expected to breach the fiduciary duties owed by our board of directors under law, and
58

Table of Contents

• we provided prior written notice to Buyer in advance and, during such the five business day period after such notice, we and our financial advisors and outside counsel negotiated with Buyer in good faith (to the extent Buyer desired to negotiate) to make such adjustments in the terms and conditions of the Merger Agreement so as to obviate (if possible) the need for such change in recommendation; or
• in circumstances relating to a superior proposal, so long as before making such change in recommendation or terminating the Merger Agreement:
• our board of directors reasonably determines in good faith, after consultation with our financial advisor and outside counsel, that, in light of such superior proposal, the failure to take such action would reasonably be expected to breach the fiduciary duties owed by our board of directors under law,
• we have provided to Buyer prior written notice of our receipt of a superior proposal that our board of directors has determined in good faith will, absent changes to the terms of the Merger Agreement, constitute a superior proposal and of our intention to make a change in recommendation
• during the five business day period after we provide notice to Buyer, we and our financial advisors and outside counsel have negotiated with Buyer in good faith (to the extent Buyer desires to negotiate) to make such adjustments in the terms of the Merger Agreement so that such superior proposal would no longer constitute a superior proposal (provided that any material modifications to such third party proposal will be deemed a new superior proposal that we will be required to again provide notice to Buyer and negotiate).
As used in the Merger Agreement and this proxy statement:
• <i>takeover proposal</i> means a proposal or offer from, or indication of interest in making a proposal or offer by, any person (other than Buyer or Merger Sub) relating to any (a) direct or indirect acquisition of assets (excluding sales of assets in the ordinary course of business) equal to twenty percent (20%) or more of the fair market value of our consolidated assets or to which twenty percent (20%) or more of our net revenues or net income on a consolidated basis are attributable, (b) direct or indirect acquisition of twenty percent (20%) or more of the voting equity interests, (c) tender offer or exchange offer that if consummated would result in any person beneficially owning (within the meaning of

• superior proposal means a bona fide written takeover proposal which did not result from a breach of the Merger Agreement that our board of directors determines in its good faith business judgment (after consultation with our outside legal counsel and financial advisors) to be more favorable to our shareholders from a financial point of view than the merger and other transactions contemplated by the Merger Agreement, taking into account (a) all financial considerations, (b) the identity of the third party making such takeover proposal, (c) the anticipated timing, conditions (including any financing condition or the reliability of any debt or equity funding commitments) and prospects for completion of takeover proposal, (d) the other terms and conditions of such takeover proposal and the implications, including relevant legal,

Section 13(d) of the Exchange Act) twenty percent (20%) or more of the voting equity interests, (d) merger, consolidation, other business combination or similar transaction involving us or any of our subsidiaries, pursuant to which such person would own twenty percent (20%) or more of the consolidated assets, net revenues or net income, taken as a whole, or (e) liquidation or dissolution (or the adoption of a plan of

liquidation or dissolution) or the declaration or payment of an extraordinary dividend (whether in cash or other property) by us.

regulatory and other aspects of such takeover proposal deemed relevant by our board of directors and (e) any revisions to the terms of the Merger Agreement and the merger proposed by Buyer.

Indemnification and Insurance

Buyer and Merger Sub agree that all existing rights to indemnification and exculpation by us under applicable law, our Articles or Bylaws in favor of each person who prior to the effective time of the merger had been an officer, director or employee of ours or who acts as a fiduciary under any of our employee benefit plans, whom we refer to as the indemnified parties, as provided in our Articles, Bylaws, or any other agreements in effect on the date of the Merger Agreement, including provisions relating to the advancement of expenses incurred in the defense

Table of Contents

of any action, suit, or other proceeding, will survive the merger and will remain in full force and effect for a period of not less than six years after the effective time of the merger. For six years after the effective time of the merger, Buyer will cause the articles of incorporation and bylaws of the surviving corporation to contain no less favorable provisions with respect to exculpation, indemnification and advancement of expenses of our directors, officers and employees. Buyer will also cause the surviving corporation to maintain for the benefit of our directors and officers, an insurance and indemnification policy for events occurring prior to the effective time of the merger, provided that the surviving corporation will not be required to pay an annual premium for such insurance in excess of 300% of the last annual premium paid prior to the date of the Merger Agreement. To the extent prepaid policies are obtained prior to the effective time of the merger, the preceding obligation will be deemed to have been satisfied.

If Buyer or the surviving corporation consolidates with or merges or converts into any other entity and is not the continuing or surviving corporation or entity of such consolidation, merger or conversion, or transfers all or substantially all of its properties and assets to any person or entity, then the Merger Agreement requires that proper provision be made so that the successors and assigns of Buyer or the surviving corporation will assume all of the applicable obligations described above. The indemnified parties are intended third-party beneficiaries of the indemnification and insurance provisions in the Merger Agreement.

Section 16 Matters

Prior to the effective time of the merger, our board of directors will take all steps required to cause the transactions involving the vesting and exercise of our stock options and our warrants in connection with the Merger Agreement by each individual who is subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to us to be exempt under Rule 16b-3 under the Exchange Act.

Financing

There is no financing contingency in the Merger Agreement.

Conditions to the Merger

Conditions to Each Party s Obligations. The respective obligations of each party to effect the merger are subject to the satisfaction or, to the extent permitted by law, waiver of each of the following conditions:

• the adoption of the Merger Agreement by the holders of a majority of the shares of our Common Stock outstanding on the record date for the special meeting on or before December 31, 2014;

• Agreemen	none of the parties are subject to any order or injunction that prohibits consummation of the transactions contemplated by the Merger t;
•	all consents, authorizations, order and approvals with any governmental entity required will have been obtained; and
	no action will be pending or threatened by or before any court in which an unfavorable judgment would prevent any of the is contemplated by the Merger Agreement or cause any such transaction to be declared unlawful or rescinded or that would reasonably impany material adverse effect.
	is to Buyer s and the Merger Sub s Obligations. The obligations of Buyer and the Merger Sub to consummate the merger are the satisfaction or, to the extent permitted by law, waiver of further conditions, including:
• prior to the	we must have performed in all material respects all obligations required to be performed by us under the Merger Agreement at or closing date of the merger;
• must be tru	our representations and warranties, except those related to authorization, validity and effect of agreements and our capitalization, as and correct as of the date of the Merger Agreement and as of the
	60

Table of Contents

closing date of the merger (provided that, for this purpose, our representations and warranties will be deemed not qualified by any references therein to materiality), except to the extent that such representation or warranty relates to an earlier time (in which case as of such earlier time), provided that this condition will be deemed satisfied even if not true and correct unless the failure would have or would reasonably be likely to have a company material adverse effect;

- our representations and warranties relating to authorization, validity and effect of agreements and capitalization must be true and correct in all respects as of the date of the Merger Agreement and as of the closing date of the merger;
- we will not be in default of any obligation under any material contract, unless such default would not reasonably be expected to have a company material adverse effect;
- Buyer must have received a certificate as to our incorporation and good standing and the good standing of Corgenix, Inc.;
- there will have been no change in our financial condition or results of operations from that reflected in the balance sheet so as to result, or be reasonably likely to result, in a company material adverse effect;
- Buyer must have received certain consents and agreements from third parties and release of certain liens;
- no action will be pending where an unfavorable injunction, judgment, order or ruling would prevent consummation of the merger or transactions contemplated by the Merger Agreement;
- dissenting shares will not constitute more than 10% of the shares issued and outstanding; and
- Buyer must have received a certificate signed by one of our senior executive officers as to the satisfaction of the conditions in the first, second, third and sixth bullets above.

Conditions to Our Obligations. Our obligation to consummate the merger is subject to the satisfaction or waiver of further conditions, including:

•	erger Sub must have performed in all material respects all obligations required to be performed by us under the Merger the closing date of the merger;
must be true and correct a representations and warra representation or warrant	tions and warranties of Buyer and Merger Sub, except those related to authorization, validity and effect of agreements, as of the date of the Merger Agreement and as of the closing date of the merger (provided that, for this purpose, our anties will be deemed not qualified by any references therein to materiality), except to the extent that such y relates to an earlier time (in which case as of such earlier time), provided that this condition will be deemed satisfied ct unless the failure would have or would reasonably be likely to have a Buyer material adverse effect;
	tions and warranties of Buyer and Merger Sub relating to authorization, validity and effect of agreements must be true as of the date of the Merger Agreement and as of the closing date of the merger;
	received a certificate from the Secretary of State of Delaware with respect to the incorporation and good standing of n the Secretary of State of Nevada with respect to the incorporation and good standing of Merger Sub;
	received all consents and approvals from third parties required in connection with the Merger Agreement and isactions contemplated by it;
	61

Table of Contents

• we will have received certificates signed by the President or Vice President and Secretary of each of Buyer and Merger Sub, attaching a copy of the resolutions of the board of directors of each approving the Merger Agreement and the consummation of the transactions; and
• we must have received a certificate signed by one of Buyer s and Merger Sub s senior executive officers as to the satisfaction of the conditions in the first, second and third bullets above.
Termination of the Merger Agreement
We, Buyer and the Merger Sub may terminate the Merger Agreement by mutual written consent at any time before the effective time of the merger. In addition, either Buyer or we may terminate the Merger Agreement, subject to certain conditions, at any time before the consummation of the merger if:
• the merger has not been consummated on or prior to December 31, 2014, provided that the right to terminate the Merger Agreement is not available to any party whose breach of any representation, warranty, covenant or agreement under the Merger Agreement has been the cause of, or resulted in, the failure of the merger to be consummated by December 31, 2014, which we refer to as the end date;
• any law or order enacted, issued, granted or promulgated by any governmental authority of competent jurisdiction in the United States that has the effect of making the merger illegal or that has the effect of prohibiting or otherwise preventing the consummation of the merger shall be in effect and, in the case of an order, shall have become final and non-appealable, provided that the right to terminate the Merger Agreement is not available to any party whose breach of any representation, warranty, covenant or agreement under the Merger Agreement has been the cause of, or resulted in, the issuance, promulgation, enforcement or entry of such law or order; or
• upon a vote at a duly held meeting of our shareholders the adoption of the Merger Agreement fails to obtain approval by the holders of a majority of the outstanding shares of our Common Stock.
Buyer may also terminate the Merger Agreement at any time prior to the effective time of the merger if:
• our board of directors has effected and not withdrawn a change in recommendation;

we have breached or failed to perform any of our representations, warranties, covenants or other agreements contained in the Merger

Agreement, in any case such that the breach, failure to perform or inaccuracy would give rise to a failure of a closing condition of Buyer and is incapable of being cured by the end date, provided that Buyer must give us 30 days written notice prior to terminating the Merger Agreement; or

• if prior to the receipt of approval from our shareholders at the special meeting, we enter into an acquisition agreement in respect of superior proposal pursuant to the terms of the Merger Agreement.
We may also terminate the Merger Agreement if:
• at any time before the receipt of the affirmative vote of the holders of a majority of the outstanding shares of our Common Stock to adopt the Merger Agreement, (i) our board of directors has received a superior proposal, (ii) we have complied with our obligations under the non-solicitation provisions of the Merger Agreement, and (iii) our board of directors concurrently approves, and we enter into, an alternative acquisition agreement with respect to the superior proposal; or
• at any time prior to the effective time of the merger, Buyer or Merger Sub has breached or failed to perform any of their respective representations, warranties, covenants or other agreements contained in the Merger Agreement, in any case such that the breach, failure to perform or inaccuracy would give rise to a failure of our closing conditions and is incapable of being cured by the end date, provided that we must give us 30 days written notice to Buyer prior to terminating the Merger Agreement.
62

Table of Contents

Subject to certain exceptions, if the Merger Agreement is terminated, it will become null and void, without liability of any party (or any shareholder, director, officer, employee, agent, consultant or representative of such party) to each other party thereto, except as specifically set forth in the Merger Agreement. No such termination will relieve or release any party from any liabilities or damages arising out of its willful and intentional breach of the Merger Agreement.

Termination Fees

We have agreed to pay Buyer a termination fee of \$324,000 plus out of pocket expenses up to \$400,000, which we refer to as the termination fee, if:

- Buyer or we terminate the Merger Agreement based on a superior proposal termination;
- Buyer terminates the agreement for our breach of the Merger Agreement; or
- Buyer terminates the Merger Agreement on the basis that our board of directors has effected a change in recommendation.

Buyer has agreed to pay us a termination fee of \$648,000 plus out of pocket expenses up to \$400,000, which we refer to as the reverse termination fee, if we terminate the Merger Agreement for a breach of the Merger Agreement by the Buyer or Merger Sub.

Expenses

Except as described above under Termination Fees, we, Buyer and Merger Sub are required to pay our own expenses in connection with the Merger Agreement and consummation of the transactions contemplated thereby, whether or not the merger is consummated.

Remedies

We, Buyer and the Merger Sub are entitled to seek and obtain an injunction, specific performance and other equitable relief to prevent breaches of the Merger Agreement to enforce specifically the terms and provisions of the Merger Agreement. The election to pursue an injunction, specific performance or other equitable relief will not restrict, impair or otherwise limit the parties from seeking to terminate the Merger agreement and Collect the termination fee or expenses described above under

Termination Fees and Expenses.

Governing Law/Jurisdiction

The Merger Agreement is governed by, and construed in accordance with, the laws of the State of Delaware, except that the NRS shall govern (a) our internal affairs, including the fiduciary duties of our board of directors, (b) the merger and (c) all other provision, or transactions contemplated by the Merger Agreement to which the NRS mandatorily applies, without regard to its principles of conflict of laws. All actions will be heard and determined exclusively in a state or federal court sitting in the state of Delaware.

Voting Agreement

Simultaneously with the execution of the Merger Agreement, Buyer entered into a voting agreement with Stephen P. Gouze, Robert Tutag, Douglass Simpson, William H. Critchfield and Ann L. Steinbarger, who are all of our executive officers and directors who are also shareholders. Together, such executive officers and directors represent an aggregate of approximately 5.5% of the outstanding shares of our Common Stock. Under the terms of the voting agreement, such executive officers and directors each agreed to vote, and irrevocably appointed Buyer as its proxy to vote, all of their shares (a) in favor of the approval of the Merger and adoption of the Merger Agreement; (b) against any other acquisition proposal or transaction that would reasonably be expected to prevent or delay the consummation of the merger or the Merger Agreement; (c) against any proposal that would

Table of Contents

reasonably be expected to result in a breach of the Merger Agreement by us; (d) in favor of any adjournment or postponement of the special meeting where approval of the shareholders is sought, as requested by Buyer; and (e) in favor of any other matter necessary to approve the Merger Agreement or otherwise requested by Buyer in order to consummate the Merger. Under the terms of the voting agreements, each such party agreed not to exercise any dissenter s rights in connection with the Merger Agreement and the transactions contemplated by the Merger Agreement. The voting agreement terminates on the earlier of (a) the effective date of the Merger, (b) the termination of the Merger Agreement and (c) the date our board of directors effects a change in board recommendation pursuant to the Merger Agreement.

DISSENTER S RIGHTS

Under NRS 92A.300 to 92A.500, holders of our Common Stock are entitled to assert dissenter s rights under Nevada Revised Statutes 92A.300 to 92A.500. By exercising dissenter s rights, shareholders can demand payment for their shares at a price equal to its fair value immediately before the corporate action. If we and the shareholder cannot agree on a fair value, we must then commence an appraisal process in state court to determine the fair value.

Nevada Revised Statutes 92A.300 through 92A.500 describe the appraisal process and the procedure shareholders must follow to exercise their right to dissent and demand payment. The full text of NRS 92A.300 to 92A.500 is attached to this proxy statement as Annex D. All references in NRS 92A.300 to 92A.500 to a stockholder or dissenter and in this summary to a shareholder are to the record holders of the shares of our Common Stock. The following discussion is a summary of the law pertaining to dissenter a rights under the NRS:

- 1) When a corporate action subject to dissenter s rights is submitted to a vote at shareholders meeting, the corporation must provide notice of dissenter s rights in the proxy materials, along with a copy of the statute describing these rights (NRS 92A.300 through 92A.500). A corporation only needs to notify shareholders of record who are entitled to dissenter s rights, not all beneficial owners.
- 2) If the corporate action is submitted to a vote at a shareholders meeting, any shareholders wishing to dissent must first deliver written notice, before the vote is taken, of their intent to demand payment for their shares if the proposed action is carried out. Second, these shareholders must not vote any of their shares in favor of the proposed action.
- 3) If the corporation receives requisite shareholder approval for the proposed action, it must then deliver a written dissenter s notice to all shareholders of record who are entitled to assert their dissenter s rights within 10 days of the effective date of the corporate action. The dissenter s notice must:
- State where the demand for payment must be sent and where and when share certificates, if any, must be deposited;
- Inform the holders of shares not represented by certificates the extent to which the transfer of the shares will be restricted after the corporation receives the demand for payment;

	Supply a form for demanding payment that includes the date of the first announcement to the news media or to the shareholders of of the proposed action and requires that the person asserting dissenters—rights certify whether or not they acquired beneficial ownership tes before that date;
• days after t	Set a date by which the corporation must receive the demand for payment, which must be no earlier than 30 days nor later than 60 the date the notice is delivered;
• received by	State that the shareholder shall be deemed to have waived the right to demand payment with respect to the shares unless the form is y the corporation by the specified date; and
•	Be accompanied by a copy of NRS 92A.300 to 92A.500.
	64

Table of Contents

4)	Shareholders who receive a dissenter s notice and who wishes to continue with the appraisal process must then:
•	Demand payment;
•	Certify that they were beneficial owners prior to the date specified in the dissenter s notice; and
•	Deposit the shareholder s certificates, if any, in accordance with the terms of the notice.
5) dissenter	Alternatively, shareholders may withdraw from the appraisal process by notifying the corporation in writing by the date specified in the s notice. After this date, dissenters may withdraw only with the written consent of the corporation.
	Once dissenting shareholders deposit the shareholder s certificates (or, in the case of uncertified shares, makes a demand for payment), all rights as a shareholder, unless they later withdraw from the appraisal process. The corporation may restrict the transfer of shares of ed shares from the date the corporation receives the dissenters demand for payment.
7) fair valu	Within 30 days of receiving demand for payment, the corporation must pay dissenters an amount the corporation estimates to be the e of the dissenter s share, plus accrued interest. This payment must be accompanied by:
•	The corporation s financial statement;
•	A statement of the corporation s estimate of the fair value of the shares; and
•	A statement of the dissenter s rights to contest the corporation s estimate of fair value.
	If dissenting shareholders wish to contest the corporation s estimate of fair value, they must notify the corporation in writing within 30 receiving the corporation s initial payment. Dissenters must provide their own estimate of fair value plus interest and demand that the ion pay the difference between this estimate and the corporation s estimate.

9) The corporation must either pay the additional amount or commence judicial proceedings within 60 days of receiving the demand and petition the court to determine the fair value of the shares and accrued interest.

MARKET PRICE INFORMATION

Our Common Stock is listed on the OTC Bulletin Board® under the trading symbol CONX. The following table provides the high and low sales prices of our Common Stock on the OTC Bulletin Board®, for each of the periods listed.

	Common Stock		
Hi	gh		Low
\$	0.15	\$	0.10
\$	0.24	\$	0.12
\$	0.20	\$	0.15
\$	0.25	\$	0.18
\$	0.23	\$	0.17
\$	0.23	\$	0.18
\$	0.44	\$	0.19
\$	0.39	\$	0.26
	\$ \$ \$ \$ \$	\$ 0.15 \$ 0.24 \$ 0.20 \$ 0.25 \$ 0.23 \$ 0.23 \$ 0.44	\$ 0.15 \$ \$ 0.24 \$ \$ 0.20 \$ \$ 0.25 \$ \$ \$ 0.25 \$ \$ \$ \$ 0.23 \$ \$ \$ 0.23 \$ \$ \$ 0.44 \$ \$

Table of Contents

The closing price of our Common Stock, as reported on the OTC Bulletin Board® on March 11, 2014, the last full trading day before the public announcement of our intention to consider strategic alternatives was \$0.23. On August 27, 2014 the last full trading day before the public announcement of the proposed merger the closing price of our Common Stock was \$0.34, and on , 2014, the latest practicable date before the printing of this proxy statement, the closing price was \$.

If the merger is consummated, each share of our Common Stock will be converted into the right to receive \$0.27 in cash, without interest and less any applicable withholding taxes, and our Common Stock will be removed from quotation on the OTC Bulletin Board® and there will be no further public market for shares of our Common Stock.

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

The following table sets forth, as of September 15, 2014, certain information regarding the ownership of our Common Stock by (i) each person known by us to be the beneficial owner of more than 5% of the outstanding shares of Common Stock (ii) each of our directors, (iii) each executive officer and (iv) all of our executive officers and directors as a group. Unless otherwise indicated, the address of each person shown is c/o Corgenix, 11575 Main Street, Suite 400, Broomfield, CO 80020. Beneficial ownership, for purposes of this table, includes options and warrants to purchase Common Stock that are either currently exercisable or will be exercisable within 60 days of June 30, 2014. No director or executive officer beneficially owned more than 5% of the Common Stock.

The percentage ownership data is based on shares of our Common Stock outstanding as of September 15, 2014, plus warrants and stock options outstanding in addition to common shares underlying convertible debt and redeemable convertible preferred stock. Under the rules of the SEC, beneficial ownership includes shares over which the indicated beneficial owner exercises voting and/or investment power. Shares of Common Stock subject to options or warrants that are currently exercisable, or will become exercisable within 60 days of September 15, 2014 are deemed outstanding for the purpose of computing the percentage ownership of the person holding the option or warrant, but are not deemed outstanding for the purpose of computing the percentage ownership of any other person. Except as otherwise noted, we believe that the beneficial owners of the shares of Common Stock listed below have sole voting and investment power with respect to all shares beneficially owned.

		Amount and Nature of Beneficial Owner		
Name and Address of Beneficial Owner	Number	Percent of Class		
Wescor, Inc. (wholly owned subsidiary of ELITech)	27,577,200(3)	45.29%		
Warrant Strategies Fund LLC	3,359,119(2)	4.99%		
350 Madison Avenue, 11th Floor				
New York, NY 10017				
Douglass T. Simpson(1)	809,171(4)	1.48%		
William H. Critchfield(1)	683,212(5)	1.25%		
Robert Tutag(1)	1,221,000(6)	2.24%		
Ann L. Steinbarger(1)	356,188(7)	0.65%		
Brandon J. Price(1)	50,000(8)	0.09%		
Stephen P. Gouze(1)	1,292,612(9)	2.37%		
David Ludvigson(1)	160,000(10)	0.29%		
Dennis Walczewski(1)	260,000(11)	0.48%		
Dennis Fusco(1)	80,000(12)	0.15%		
All current directors and current executive officers as a group (9 persons)	4,912,183(13)	8.87%		

(1)	Current director or officer
(2) any warrants if suc	Contractual restrictions in the warrants and purchase agreement with us prohibit Warrant Strategies Fund from exercising the exercise would cause them to exceed
	66

Table of Contents

4.99% beneficial ownership of Corgenix. Warrant Strategies Fund holds warrants to acquire up to 4,508,182 shares of Common Stock. (3) Includes 6,666,667 shares issuable upon the exercise of warrants. (4) Includes 373,333 shares issuable within 60 days of September 15, 2014 upon the exercise of stock options. (5) Includes 316,667 shares issuable within 60 days of September 15, 2014 upon the exercise of stock options. Includes 260,000 shares issuable within 60 days of September 15, 2014 upon the exercise of stock options. (6) Includes 173,333 shares issuable within 60 days of September 15, 2014 upon the exercise of stock options. (7) (8) Includes 50,000 shares issuable within 60 days of September 15, 2014 upon the exercise of stock options. (9) Includes 240,000 shares issuable within 60 days of September 15, 2014 upon the exercise of stock options. (10)Includes 160,000 shares issuable within 60 days of September 15, 2014 upon the exercise of stock options. (11)Includes 260,000 shares issuable within 60 days of September 15, 2014 upon the exercise of stock options. Includes 80,000 shares issuable within 60 days of September 15, 2014 upon the exercise of stock options. (12)Includes 1,913,333 shares issuable within 60 days of September 15, 2014 upon the exercise of stock options. (13)

Table of Contents

PROPOSAL TWO COMPENSATION ARRANGEMENTS FOR OUR NAMED EXECUTIVE OFFICERS IN CONNECTION WITH THE MERGER

As required by Item 402(t) of Regulation S-K and Section 14A(b) of the Exchange Act, we are providing our shareholders with the opportunity to cast an advisory, non-binding vote on the compensation that may become payable to our named executive officers in connection with the completion of the merger, which we refer to as the golden parachute proposal.

We believe that the compensation that may become payable to our named executive officers in connection with the completion of the merger is reasonable and demonstrates that our executive compensation program was designed appropriately and structured to ensure the retention of talented executives and a strong alignment of their interests with the long-term interests of our shareholders. This vote is not intended to address any specific item of compensation, but rather the overall compensation that may become payable to our named executive officers in connection with the completion of the merger. In addition, this vote is separate and independent from the vote of our shareholders on the merger proposal and the adjournment proposal. Our board of directors recommends that our shareholders vote FOR the following resolution:

RESOLVED, that the compensation that may become payable to the named executive officers named in this proxy statement in connection with the completion of the merger is approved.

This vote is advisory and, therefore, it will not be binding on us, nor will it overrule any prior decision or require our board of directors (or any committee thereof) to take any action.

The affirmative vote of the holders of a majority of the shares of our Common Stock present in person or by proxy and entitled to vote at the Special Meeting will be required to approve the golden parachute proposal. Abstentions will have the same effect as a vote AGAINST the approval of the golden parachute proposal. Assuming a quorum is present at the Special Meeting, broker non-votes will have no effect on the approval of the golden parachute proposal.

More information regarding the compensation that may become payable to our named executive officers in connection with the completion of the merger is set forth in the section captioned The Merger Interests of Our Directors and Executive Officers in the Merger beginning on page 40 of this proxy statement.

Our board of directors unanimously recommends that you vote FOR the approval of the compensation that may become payable to our named executive officers in connection with the merger.

Table of Contents

PROPOSAL THREE ADJOURNMENT OF THE SPECIAL MEETING

We are submitting a proposal for consideration at the Special Meeting to authorize the named proxies to approve one or more adjournments or postponements of the Special Meeting if there are not sufficient votes to adopt the Merger Agreement at the time of the Special Meeting, which we refer to as the adjournment proposal. Even though a quorum may be present at the Special Meeting, it is possible that we may not have received sufficient votes to adopt the Merger Agreement by the time of the Special Meeting. In that event, we would determine to adjourn or postpone the Special Meeting in order to solicit additional proxies. The adjournment proposal relates only to an adjournment of the Special Meeting for purposes of soliciting additional proxies to obtain the requisite shareholder approval to adopt the Merger Agreement. Any other adjournment of the Special Meeting (e.g., an adjournment required because of the absence of a quorum) would be voted upon pursuant to the discretionary authority granted by the proxy.

The affirmative vote of the holders of a majority of the shares of our Common Stock present in person or by proxy and entitled to vote at the Special Meeting will be required to approve the adjournment, if necessary, of the Special Meeting to solicit additional proxies in favor of the adoption of the Merger Agreement. Abstentions will have the same effect as a vote AGAINST the approval of the adjournment proposal. Assuming a quorum is present at the Special Meeting, broker non-votes will have no effect on the approval of the adjournment proposal.

Properly executed proxies will be voted FOR the adjournment proposal, unless otherwise noted on the proxies. If the Special Meeting is adjourned or postponed, we are not required to give notice of the time and place of the adjourned meeting unless our board of directors fixes a new record date for the Special Meeting.

Our board of directors unanimously recommends that you vote FOR the adjournment proposal so that proxies may be used for that purpose, should it become necessary.

69

m	. 1		c			
Tal	hl	e	ot	on	itei	nts

FUTURE SHAREHOLDER PROPOSALS

If the merger is consummated, we will not have public shareholders, and there will be no public participation in any future meetings of shareholders. However, if the merger is not completed, we plan to hold our 2014 Annual Meeting of Shareholders no later than Tuesday, December 16, 2014. Proposals of our shareholders intended for inclusion in the proxy statement and proxy card to be furnished to all shareholders entitled to vote at the 2014 Annual Meeting of Shareholders pursuant to Rule 14a-8 promulgated under the Exchange Act by the SEC must have been received at our principal executive offices by July 8, 2014.

If a shareholder wished to present a proposal before the 2014 Annual Meeting, but did not wish to have the proposal considered for inclusion in the proxy statement, the shareholder must have submitted a proposal in writing to Corgenix Medical Corporation, 11575 Main Street, Suite 400, Broomfield, Colorado, 80020, Attention: William Critchfield, no later than the close of business on August 19, 2014.

If a shareholder wished to nominate an individual for election to our board of directors at the 2014 Annual Meeting, the shareholder must have delivered a notice in writing to Corgenix Medical Corporation, 11575 Main Street, Suite 400, Broomfield, Colorado 80020, Attention: William Critchfield, no later than the close of business on August 19, 2014.

WHERE YOU CAN FIND MORE INFORMATION

We are subject to the informational reporting requirements of the Exchange Act and file annual, quarterly and current reports, proxy statements and other information with the SEC. You can read our SEC filings, including this proxy statement, through the Internet at the SEC s website at www.sec.gov. You may also read and copy any document we file with the SEC at its public reference facility at 100 F Street, N.E., Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information about the public reference room.

If you have any questions about this proxy statement, the Special Meeting or the merger or need assistance with the voting procedures, you should contact our proxy solicitor, Georgeson Inc., at (877) 278-4751.

Board of Directors

Corgenix Medical Corporation

, 2014

WHETHER OR NOT YOU ARE ABLE TO ATTEND THE SPECIAL MEETING IN PERSON, PLEASE SUBMIT YOUR PROXY VIA THE INTERNET, BY TELEPHONE, OR COMPLETE, SIGN AND DATE THE ENCLOSED PROXY CARD AND RETURN IT

IN THE ENVELOPE PROVIDED AS SOON AS POSSIBLE. IF YOU HAVE INTERNET ACCESS, WE ENCOURAGE YOU TO RECORD YOUR VOTE VIA THE INTERNET. THIS ACTION WILL NOT LIMIT YOUR RIGHT TO VOTE IN PERSON AT THE SPECIAL MEETING.

Table of Contents

	O'T TE	TT ^ T	TITIO	TANT
H) X H)		HON	VERS	$\mathbf{I}(\mathbf{I})\mathbf{N}$

ANNEX A

MERGER AGREEMENT

AGREEMENT AND PLAN OF MERGER

among

CORGENIX MEDICAL CORPORATION

and

CENTENNIAL MEDICAL HOLDINGS, INC.

and

CENTENNIAL INTEGRATED, INC.

Dated August 27, 2014

Table of Contents

TABLE OF CONTENTS

		Page
ARTICLE I PURCHASE AND SALE		A-2
1.1	The Merger	A-2
1.2	Closing	A-2
1.3	Effective Time	A-2
1.4	Effect of the Merger	A-2
1.5	Certificate of Incorporation and Bylaws of the Surviving Corporation	A-2
1.1 1.2 1.3 1.4 1.5 1.6	Directors and Officers of the Surviving Corporation	A-2
ARTICLE II EFFECT OF MERGER ON CA	APITAL STOCK	A-3
2.1	Conversion of Securities	A-3
<u>2.2</u>	Exchange of Certificates	A-4
2.3	Withholding Rights	A-5
2.4	Lost Certificates	A-5
2.3 2.4 2.5	Dissenting Shares	A-5 A-6
<u>2.6</u>	Treatment of Company Options and Warrants	A-0 A-6
2.0	Treatment of Company Options and Warrants	A-0
ARTICLE III REPRESENTATIONS AND V		A-7
<u>3.1</u>	Existence; Good Standing; Corporate Authority	A-7
<u>3.2</u>	Authorization, Validity, and Effect of Agreements	A-8
<u>3.3</u>	<u>Capitalization</u>	A-9
3.2 3.3 3.4 3.5 3.6 3.7 3.8	Other Interests	A-9
<u>3.5</u>	No Violation; Governmental Consents	A-9
<u>3.6</u>	SEC Reports; Financial Statements	A-10
<u>3.7</u>	Absence of Undisclosed Liabilities	A-11
3.8	Absence of Certain Changes or Events	A-12
<u>3.9</u>	Contracts	A-14
3.10	Title to Assets	A-15
3.11	Adequacy of Assets	A-15
3.12	Litigation, Claims and Proceedings	A-15
3.13	Taxes	A-15
3.14	Proprietary Rights	A-17
3.15	Labor Relations	A-19
3.16	Transactions with Affiliates	A-19
3.17	Permits; Compliance with Laws; Regulatory Compliance	A-19
3.18	Environmental Matters	A-20
3.19	Employee Plans and ERISA	A-20
3.20	Properties	A-21
3.21	Insurance	A-22
3.22	Brokerage and Related Fees	A-22
3.23	Knowledge	A-22
3.24	Anti-Takeover Provisions	A-23
		25

ARTICLE IV REPRESENTATIONS AND W	ARRANTIES OF PARENT AND MERGER SUB	A-23
4.1	Existence; Good Standing; Corporate Authority	A-23
<u>4.2</u>	Authorization, Validity, and Effect of Agreements	A-23
4.3	No Violation; Consents	A-23
4.3 4.4 4.5	Litigation, Claims and Proceedings	A-24
4.5	Brokerage and Related Fees	A-24
4.6	Financial Capacity; Solvency	A-25
ARTICLE V COVENANTS		A-25
<u>5.1</u>	Conduct of Business Prior to the Effective Time	A-25
	Filings; Other Action	A-25
5.3	No Solicitation	A-25
5.4	Shareholders Meeting; Preparation of Proxy Materials and Schedule 13E-3	A-28
5.5	Notice of Certain Events	A-29
5.6	Expenses	A-29
5.7	Banking Matters	A-30
5.8	Directors of the Company	A-30
5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9	Public Announcements	A-30
5.10	Indemnification of Directors and Officers	A-30
<u>5.11</u>	Parent Agreement Concerning Merger Sub	A-31
<u>5.12</u>	Section 16 Matters	A-31
<u>5.12</u> <u>5.13</u>	Stock Deregistration	A-31
<u>5.13</u> <u>5.14</u>	Takeover Statutes	A-32 A-32
<u>5.14</u> <u>5.15</u>	Participation in Litigation	A-32 A-32
<u>5.15</u>	Farticipation in Litigation	A-32
ARTICLE VI CONDITIONS		A-32
<u>6.1</u>	Conditions to Each Party s Obligation to Effect the Merger	A-32
<u>6.2</u>	Conditions to Obligation of the Company to Effect the Merger	A-33
<u>6.3</u>	Conditions to Obligation of Parent and Merger Sub to Effect the Merger	A-34
6.4	<u>Frustration of Closing Conditions</u>	A-36
ARTICLE VII SURVIVAL		A-36
<u>7.1</u>	Survival of Representations	A-36
ARTICLE VIII TERMINATION		A-36
8.1	<u>Termination By Mutual Consent</u>	A-36
	Termination By Either Parent or the Company	A-36
8.3	Termination By Parent	A-37
8.4	Termination By the Company	A-37
8.2 8.3 8.4 8.5 8.6 8.7 8.8	Effect of Termination	A-38
8.6	Parent Termination Fee	A-38
8.7	Company Termination Fee	A-40
8.8	Sole Remedy	A-40
<u>5.5</u>	<u>~~~~~~~</u>	71 10
	A-iii	

ARTICLE IX CONFIDENTIALITY		A-4 1
<u>9.1</u>	Confidentiality Covenants	A-41
9.2	Disclosure Pursuant to Legal Process	A-41
<u>9.2</u> <u>9.3</u>	Termination of Confidentiality Obligations	A-42
ARTICLE X GENERAL PROVISIONS		A-42
<u>10.1</u>	<u>Notices</u>	A-42
10.2	Entire Agreement; No Third-Party Beneficiaries	A-43
<u>10.2</u> <u>10.3</u>	Governing Law; Venue	A-43
10.4	Counterparts	A-44
10.5	Headings	A-44
10.6	Interpretation	A-44
10.7	Waivers	A-44
10.8	Incorporation of Exhibits	A-44
10.9	Severability	A-45
10.10	Enforcement of Agreement	A-45
10.11	Consent	A-45
10.12	Assignment	A-45
10.13	<u>Definitions</u>	A-45
	A-iv	

Table of Contents

INDEX OF DEFINITIONS

The following terms are defined in the Section of this Agreement set forth after such term below:

Acceptable Confidentiality Agreement Section 10.13
Affiliate Section 10.13
Agents Section 9.1

Agreement Introductory Paragraph

Articles of Merger Section 1.3 **Book-Entry Shares** Section 2.2(a) **Business Day** Section 10.13 **CERCLA** Section 10.13 Change of Board Recommendation Section 5.3(d) Closing Section 1.2 Section 1.2 Closing Date **COBRA** Section 3.19(d) Code Section 2.3

Company Introductory Paragraph

Company Acquisition Agreement Section 5.3(a) Company Board Recitals Company Common Stock Recitals Company Disclosure Schedule Article III Company Material Adverse Effect Section 3.8 Company Proxy Statement Section 5.4(a) Company Shareholders Meeting Section 5.4(a) Company Stock Option Section 2.5(a) Company Termination Fee Section 8.7(a) Confidential Information Section 9.1 Consent Section 3.5(b) Contracts Section 3.9 Disclosing Party Section 9.1 Dissenting Shares Section 2.5 **D&O** Insurance Section 5.10(c) Section 1.3 Effective Time End Date Section 8.2(a) **Environmental Laws** Section 10.13 **ERISA** Section 3.19(c) Section 3.5(b) Exchange Act Exchange Agent Section 2.2(a) **FDA** Section 3.17(c) **FDCA** Section 3.17(c) Financial Advisor Section 3.2(b) Section 3.6(b) **GAAP**

Governmental Entity

Hazardous Material

Intervening Event

Section 3.5(b)

Section 10.13

Section 10.13

Table of Contents

Warrants

Intervening Event Notice

Laws

Section 10.13

Liability

Section 10.13

Liens

Section 10.13

Merger

Section 10.13

Merger

Section 1.1

Merger Consideration

Merger Sub

Introductory Paragraph

NRS Section 1.1
Option Payments Section 2.6(a)
Order Section 10.13

Parent Introductory Paragraph

Parent Termination Fee Section 8.6(a) Section 2.2(a) Payment Fund Permitted Liens Section 3.10 Proprietary Rights Section 3.8(i) Proxy Statement Section 3.6(d) **Public Filings** Section 3.6(a) Real Property Section 3.20(a) Receiving Party Section 9.1 Release Section 10.13 Representatives Section 5.3(a) Requisite Company Vote Section 3.2 SEC Section 3.5(b) Secretary of State Section 1.3 Securities Act Section 3.6(a) Section 2.1(b) Shares Superior Proposal Section 10.13 Superior Proposal Notice Section 5.3(g) Surviving Corporation Section 1.1 Takeover Statute Section 3.24 Takeover Proposal Section 10.13 Section 3.13(1) Taxes Section 3.13(1) Tax Return Warrant Payment Section 2.6(b)

A-vi

Section 2.6(b)

Table of Contents

AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER (this *Agreement*), dated August 27, 2014 by and among and Corgenix Medical Corporation, a Nevada corporation (the *Company*), Centennial Medical Holdings, Inc., a Delaware corporation (*Parent*), and a wholly owned subsidiary of Parent, Centennial Integrated, Inc., a Nevada corporation (*Merger Sub*).

RECITALS

WHEREAS, following the acceptance for payment of Shares, upon the terms and subject to the conditions set forth in this Agreement, Merger Sub will be merged with and into the Company, with the Company continuing as the Surviving Corporation in accordance with the Nevada Revised Statutes, as amended (the *NRS*), whereby each issued and outstanding Share of common stock, par value \$0.001 per share, of the Company (the *Company Common Stock*) (other than Shares to be cancelled in accordance with Section 2.1(b) and other than Dissenting Shares) will be converted into the right to receive the Merger Consideration, payable to the holder in cash, without interest;

WHEREAS, the Board of Directors of the Company (the *Company Board*) has, upon the terms and subject to the conditions set forth herein, unanimously (a) determined that the transactions contemplated by this Agreement, including the Merger, are fair to and in the best interests of the Company and its shareholders, (b) approved and declared advisable the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby, including the Merger, and (c) resolved to recommend that the Company s shareholders tender their Shares and adopt this Agreement and thereby approve the transactions contemplated hereby, including the Merger;

WHEREAS, the respective Boards of Directors of Parent and Merger Sub have, upon the terms and subject to the conditions set forth in this Agreement, unanimously (i) determined that the transactions contemplated by this Agreement, including the Merger, are fair to and in the best interests of Parent and Merger Sub and their respective shareholders and (ii) approved and declared advisable this Agreement and the transactions contemplated hereby, including the Merger;

WHEREAS, the parties desire to make certain representations, warranties, covenants and agreements in connection with the Merger and the transactions contemplated by this Agreement and also to prescribe certain conditions to the Merger.

NOW, THEREFORE, in consideration of the foregoing, and of the representations, warranties, covenants, and agreements contained herein, the parties hereto hereby agree as follows:

Table of Contents

ARTICLE I

PURCHASE AND SALE

The Merger. Upon the terms and subject to the conditions set forth in this Agreement and in accordance with the NRS, at the Effective Time, the Merger Sub will be merged with and into the Company, the separate corporate existence of the Merger Sub will cease and the Company will continue as the surviving corporation (sometimes hereinafter referred to as the *Surviving Corporation*) in the merger (the *Merger*).
1.2 Closing. Subject to the terms and conditions of this Agreement, the closing of the Merger (the <i>Closing</i>) will take place at the offices of Husch Blackwell LLP, 1700 Lincoln Street, Suite 4700, Denver, Colorado 80202 at 10:00 a.m., Mountain Time on a date to be specified by the parties, which shall be no later than the fifth Business Day after the satisfaction or, to the extent permitted by applicable Law, waiver of the conditions set forth in Article VI (other than any such conditions which by their nature cannot be satisfied until the Closing Date, which shall be required to be so satisfied or, to the extent permitted by applicable Law, waived on the Closing Date), unless another date or place is agreed to in writing by the parties. The date on which the Closing occurs is referred to as the <i>Closing Date</i> .
1.3 Effective Time. Subject to the provisions of this Agreement, on the Closing Date, the Merger Sub and the Company will cause articles of merger (the <i>Articles of Merger</i>) to be executed and filed with the Secretary of State of the State of Nevada (the <i>Secretary of State</i>) in such form and executed as provided in the NRS. The Merger will become effective on the date and at the time at which the Articles of Merger have been duly filed with the Secretary of State or such other time as is agreed upon by the parties hereto and specified in the Articles of Merger, and such time is hereinafter referred to as the <i>Effective Time</i> .
1.4 Effect of the Merger . The Merger will have the effect as provided in the applicable provisions of the NRS. Without limiting the generality of the foregoing, upon the Merger, all the rights, privileges, immunities, powers and franchises of the Company and the Merger Sub will vest in the Surviving Corporation and all obligations, duties, debts and liabilities of the Company and the Merger Sub will be the obligations, duties, debts and liabilities of the Surviving Corporation.
1.5 Certificate of Incorporation and Bylaws of the Surviving Corporation. As of the Effective Time, the certificate of incorporation and bylaws of Merger Sub, as in effect immediately prior to the Effective Time, will be the certificate of incorporation and bylaws of the Surviving Corporation until thereafter amended as provided therein or by applicable Law.
1.6 Directors and Officers of the Surviving Corporation . The directors of Merger Sub immediately prior to the Effective Time, from and after the Effective Time, will be the directors of the Surviving Corporation, until their respective successors are duly elected or appointed and qualified or their earlier death, resignation or removal in accordance with the certificate of incorporation and bylaws of the Surviving Corporation. The officers of the Company immediately prior to the Effective Time, from and after the Effective Time, will

continue as the officers of the Surviving Corporation, each to hold office in accordance with the certificate of incorporation and bylaws of the Surviving Corporation until their respective successors are duly elected or appointed and qualified or their earlier death, resignation or removal.

Table of Contents

ARTICLE II

EFFECT OF MERGER ON CAPITAL STOCK

2.1 holders of any securities	Conversion of Securities. At the Effective Time, by virtue of the Merger and without any action on the part of the of Merger Sub or the Company:
(a) validly issued, fully paid	Each issued and outstanding share of common stock of Merger Sub will be converted into and become one and non-assessable share of common stock, no par value per share, of the Surviving Corporation.
	Any Shares of Company Common Stock (<i>Shares</i>) that are owned by the Company as treasury stock, any subsidiary of the Company, and any Shares owned by Parent or Merger Sub or any subsidiary of Parent or Merger ly canceled and will cease to exist and no consideration will be delivered in exchange therefor.
Consideration), payabl	Each issued and outstanding Share (other than Shares to be canceled in accordance with Section 2.1(b)) neelled and cease to exist, and will be converted into the right to receive an amount in cash equal to \$0.27 (the <i>Merger</i> to the holder thereof without interest upon surrender, in the manner provided in this Agreement, of the certificate (or ownership) formerly representing such Share.
	A-3

7D 1	1			c.	\sim			
Tal	٦I	\boldsymbol{e}	\cap 1	1		۱n	tei	ntc

2.2	Exchange of Certificates.
-----	---------------------------

- Prior to the Effective Time, Parent will appoint an exchange agent reasonably acceptable to the Company (the Exchange Agent) to act (a) as the agent for the purpose of exchanging for the Merger Consideration for: (i) the Certificates, or (ii) book-entry shares which immediately prior to the Effective Time represented the Shares of Company Common Stock (the Book-Entry Shares). On and after the Effective Time, Parent will deposit, or cause the Surviving Corporation to deposit, with the Exchange Agent, sufficient funds to pay the aggregate Merger Consideration that is payable in respect of all of the Shares of Company Common Stock represented by the Certificates and the Book-Entry Shares (the Payment Fund) in amounts and at the times necessary for such payments. If for any reason (including losses) the Payment Fund is inadequate to pay the amounts to which holders of Shares will be entitled under Section 2.1(b), Parent will take all steps necessary to enable or cause the Surviving Corporation promptly to deposit in trust additional cash with the Exchange Agent sufficient to make all payments required under this Agreement, and Parent and the Surviving Corporation will in any event be liable for the payment thereof. The Payment Fund will not be used for any other purpose. The Surviving Corporation will pay all charges and expenses, including those of the Exchange Agent, in connection with the exchange of Shares for the Merger Consideration. Promptly after the Effective Time, Parent will send, or will cause the Exchange Agent to send, to each record holder of Shares of Company Common Stock at the Effective Time, a letter of transmittal and instructions (which will specify that the delivery will be effected, and risk of loss and title will pass, only upon proper delivery of the Certificates or transfer of the Book-Entry Shares to the Exchange Agent) for use in such exchange. The Payment Fund, pending its disbursement to the holders of Shares, will be invested by the Exchange Agent as directed by Parent; provided that (i) Parent will not direct the Exchange Agent to make any such investments that are speculative in nature and (ii) no such investment or losses will affect the amounts payable to such holders and Parent will promptly replace or cause to be replaced any funds deposited with the Exchange Agent that are lost through any investment so as to ensure that the Payment Fund is at all times maintained at a level sufficient for the Exchange Agent to pay the Merger Consideration. Earnings from investments will be the sole and exclusive property of Parent and the Surviving Corporation.
- (b) Each holder of Shares of Company Common Stock that have been converted into the right to receive the Merger Consideration will be entitled to receive the Merger Consideration in respect of the Company Common Stock represented by a Certificate or Book-Entry Share upon (i) surrender to the Exchange Agent of a Certificate, together with a duly completed and validly executed letter of transmittal and such other documents as may reasonably be requested by the Exchange Agent, or (ii) receipt of an agent s message by the Exchange Agent (or such other evidence, if any, of transfer as the Exchange Agent may reasonably request) in the case of Book-Entry Shares. Until so surrendered or transferred, as the case may be, each such Certificate or Book-Entry Share, as applicable, will represent after the Effective Time for all purposes only the right to receive the Merger Consideration payable in respect thereof. No interest will be paid or accrued on the cash payable upon the surrender or transfer of any Certificate or Book-Entry Share. Upon payment of the Merger Consideration pursuant to the provisions of this Article II, each Certificate or Certificates so surrendered will immediately be cancelled.
- (c) If any portion of the Merger Consideration is to be paid to a person other than the person in whose name the surrendered Certificate or the transferred Book-Entry Share,

Table of Contents

as applicable, is registered, it will be a condition to such payment that (i) such Certificate will be properly endorsed or will otherwise be in proper form for transfer or such Book-Entry Share will be properly transferred, and (ii) the person requesting such payment will pay to the Exchange Agent any transfer or other Tax required as a result of such payment to a person other than the registered holder of such Certificate or Book-Entry Share, as applicable, or establish to the reasonable satisfaction of the Exchange Agent that such Tax has been paid or is not payable.
(d) All Merger Consideration paid upon the surrender of Certificates or transfer of Book-Entry Shares in accordance with the terms hereof will be deemed to have been paid in full satisfaction of all rights pertaining to the Shares of Company Common Stock formerly represented by such Certificate or Book-Entry Shares, and from and after the Effective Time, there will be no further registration of transfers of Shares of Company Common Stock on the stock transfer books of the Surviving Corporation. If, after the Effective Time, Certificates or Book-Entry Shares are presented to the Surviving Corporation, they will be cancelled and exchanged for the Merger Consideration provided for, and in accordance with the procedures set forth, in this Article II.
(e) Any portion of the Payment Fund that remains unclaimed by the holders of Shares six (6) months after the Effective Time will be returned to Parent, upon demand, and any such holder who has not exchanged Shares of Company Common Stock for the Merger Consideration in accordance with this Section 2.2 prior to that time will thereafter look only to Parent for payment of the Merger Consideration. Notwithstanding the foregoing, Parent will not be liable to any holder of Shares of Company Common Stock for any amounts paid to a public official or Governmental Entity pursuant to applicable abandoned property, escheat or similar Laws. Any amounts remaining unclaimed by holders of Shares of Company Common Stock three (3) years after the Effective Time (or such earlier date, immediately prior to such time when the amounts would otherwise escheat to or become property of any Governmental Entity) will become, to the extent permitted by applicable Law, the property of Parent free and clear of any claims or interest of any person previously entitled thereto.
Withholding Rights. Each of the Exchange Agent, Parent, Merger Sub and the Surviving Corporation will be entitled to deduct and withhold from the consideration otherwise payable to any person pursuant to this Article II such amounts as may be required to be deducted and withheld with respect to the making of such payment under the Internal Revenue Code of 1986, as amended, and applicable Treasury Regulations issued pursuant thereto (the <i>Code</i>), or any provision of state, local or foreign Tax Law. To the extent that amounts are so deducted and withheld by the Exchange Agent, Parent, Merger Sub or the Surviving Corporation, as the case may be, such amounts will be treated for all purposes of this Agreement as having been paid to the person in respect of which the Exchange Agent, Parent, Merger Sub or the Surviving Corporation, as the case may be, made such deduction and withholding.
2.4 Lost Certificates. If any Certificate will have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if required by Parent, the posting by such person of a bond, in such reasonable amount as Parent may direct, as indemnity against any claim that may be made against it with respect to such Certificate, the Exchange Agent will issue, in exchange for such

Table of Contents

lost, stolen or destroyed Certificate, the Merger Consideration to be paid in respect of the Shares formerly represented by such Certificate as contemplated under this Article II.

Dissenting Shares. Notwithstanding anything in this Agreement to the contrary, Shares outstanding immediately prior to the Effective Time and held by a holder who is entitled to demand and has properly demanded appraisal for such Shares in accordance with, and who complies in all respects with, NRS Sections 92A.300 through 92A.500 (such Shares, the *Dissenting Shares*) will not be converted into the right to receive the Merger Consideration, and will instead represent the right to receive payment of the fair value of such Dissenting Shares in accordance with and to the extent provided by NRS Sections 92A.300 through 92A.500. If any such holder fails to perfect or otherwise waives, withdraws or loses his right to appraisal under NRS Sections 92A.300 through 92A.500 or other applicable Law, then the right of such holder to be paid the fair value of such Dissenting Shares will cease and such Dissenting Shares will be deemed to have been converted, as of the Effective Time, into and will be exchangeable solely for the right to receive the Merger Consideration, without interest. The Company will give Parent prompt notice of any demands received by the Company for appraisal of Shares, attempted withdrawals of such demands and any other instruments served pursuant to the NRS and received by the Company relating to rights to be paid the fair value of Dissenting Shares, and Parent will have the right to participate in all negotiations and proceedings with respect to such demands. Prior to the Effective Time, the Company will not, except with the prior written consent of Parent, make any payment with respect to, or settle or compromise or offer to settle or compromise, any such demands, or approve any withdrawal of any such demands, or agree to do any of the foregoing.

2.6 Treatment of Company Options and Warrants.

- (a) The Company will take all requisite action so that, at the Effective Time, each option to acquire Shares of Company Common Stock (each, a *Company Stock Option*) that is outstanding immediately prior to the Effective Time, whether or not then vested or exercisable, will be, by virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company, the holder of that Company Stock Option or any other person, cancelled and converted into the right to receive from the Surviving Corporation as promptly as reasonably practicable after the Effective Time, an amount in cash, without interest, equal to the product of (i) the aggregate number of Shares of Company Common Stock subject to such Company Stock Option and (ii) the excess, if any, of the Merger Consideration over the per Share exercise price under such Company Stock Option, less any taxes to be withheld in accordance with Section 2.3 (the *Option Payments*); provided, that if the per Share exercise price is greater than the Merger Consideration, then such Company Stock Option will be cancelled without any payment therefor. From and after the Effective Time, any such cancelled Company Stock Option will no longer be exercisable by the former holder, but will only entitle the holder to the payment of the Option Payments.
- (b) The Company will take all requisite action so that, at the Effective Time, each warrant to purchase Shares of Company Common Stock, all of which are listed on <u>Schedule 2.6</u> of the Company Disclosure Schedule (collectively, the *Warrants*), that is outstanding immediately prior to the Effective Time and not terminated pursuant to its terms in connection

Table of Contents

with the Merger will be cancelled and, in exchange, each form holder of any such cancelled Warrant will be entitled to receive from the Surviving Corporation, as promptly as reasonably practicable after the Effective Time, in consideration of the cancellation of the Warrant, an amount in cash, without interest, equal to the product of (i) the total number of Shares previously subject to such Warrant and (ii) the excess, of any, of the Merger Consideration over the exercise price per Share previously subject to the such Warrant, less any taxes to be withheld in accordance with Section 2.3 (the *Warrant Payment*); provided, that if the per Share exercise price is greater than the Merger Consideration, then such Warrant will be cancelled without any payment therefor. From and after the Effective Time, any such cancelled Warrant will no longer be exercisable by the former holder, but will only entitle the holder to the payment of the Warrant Payment, and the Company will use reasonable efforts to obtain all necessary consents to ensure that former holders of the Warrants will have no rights other than the right to receive the Warrant Payment.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth in (i) the Company Public Filings filed prior to the date of this Agreement (without giving effect to any amendment to any such Company Public Filings filed on or after the date hereof) to the extent reasonably apparent from the face of such information therein that it qualifies any representation and warranty in this Article III (but excluding any forward-looking disclosures contained in Forward Looking Statements and Risk Factors sections of the Company Public Filings and any other disclosures included therein to the extent they are primarily predictive, cautionary or forward-looking in nature) or (ii) the correspondingly numbered section of the Company Disclosure Schedule delivered to Parent simultaneously with the execution of this Agreement (the *Company Disclosure Schedule*) that relates to such section or in another section of the Company Disclosure Schedules to the extent that it is reasonably apparent on the face of such disclosure that such disclosure is applicable to such Section, the Company represents and warrants to Parent and Merger Sub as of the date of this Agreement as follows:

4	3.1	Fv	istence: Good	Standing.	Cornorate /	uthority
J).1	r.x	istence: C r ood	i Standing:	Corborate A	X ULNOTILV

- (a) The Company is a corporation duly incorporated, validly existing, and in good standing under the laws of the State of Nevada. The Company is qualified to do business as a foreign corporation and is in good standing in the jurisdictions, which are listed on <u>Schedule 3.1</u> of the Company Disclosure Schedule, where the character of the properties owned or leased by it or the transaction of its business makes such qualification necessary, except where the failure to be so qualified or licensed or to be in good standing, would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- (b) The Company has all requisite corporate power and authority to own, operate, and lease its properties and assets and carry on its business as presently conducted and as proposed to be conducted.

Table of Contents

(c)	The copies of the Company s Articles of Incorporation and Bylaws, which have been delivered to Parent, include any and all
amendmen	ts made thereto at any time prior to the date of this Agreement and are true, correct, and complete and in full force and effect. The
Company i	s not in violation of any provision of its Articles of Incorporation or Bylaws in any material respect.

3.2 Authorization, Validity, and Effect of Agreements.

- (a) The Company has the requisite corporate power and authority to execute and deliver this Agreement and all agreements and documents contemplated hereby and to perform its obligations hereunder and to consummate the Merger. Subject to approval of this Agreement by the Company s shareholders by the affirmative vote or consent of the holders of a majority of the outstanding Shares of Company Common Stock (the *Requisite Company Vote*), the consummation by the Company of the transactions contemplated hereby has been duly authorized by all requisite corporate action of the Company. This Agreement has been duly executed and delivered by the Company and, assuming the due authorization, execution and delivery by Parent and Merger Sub, constitutes, and all agreements and documents contemplated hereby (when executed and delivered pursuant hereto for value received) will constitute valid and legally binding obligations of the Company, enforceable against the Company in accordance with their respective terms, subject to shareholder approval and except to the extent that enforceability may be limited by applicable bankruptcy, insolvency, moratorium, or other similar Laws relating to creditors rights and general principles of equity.
- (b) The Company Board has received the written opinion of Inverness Advisors, a division of KEMA Partners LLC (the *Financial Advisor*), dated the date of such opinion, to the effect that, as of the date thereof and subject to the qualifications, considerations, assumptions and limitations set forth therein, the Merger Consideration to be received by the holders of Shares is fair, from a financial point of view, to such holders, a copy of which opinion will be delivered to Parent, solely for informational purposes, promptly after the receipt thereof by the Company Board. The Financial Advisor has consented to the inclusion of a copy of such opinion in the Proxy Statement. Parent and Merger Sub acknowledge and agree that such an opinion may not be relied on by Parent, Merger Sub or any of their respective Affiliates.

Table	of	Contents

3.3 Capitalization.
(a) The authorized capital stock of the Company consists of (a) 200,000,000 Shares of Company Common Stock, of which 54,228,834 Shares are issued and outstanding as of August 25, 2014 and (b) 5,000,000 shares of preferred stock, \$.001 par value per share, of which no shares are issued and outstanding as of August 25, 2014. All of the issued and outstanding Shares of Company Common Stock have been duly authorized, and are validly issued, fully paid, and non-assessable. Except as set forth on Schedule 3.3 of the Company Disclosure Schedule, there are no outstanding or authorized options, warrants, purchase rights, subscription rights, conversion rights, exchange rights, or other contracts or commitments that could require the Company to issue, sell, or otherwise cause to become outstanding any of its capital stock and no voting trusts, proxies, or other agreements or understandings with respect to the voting of the capital stock of the Company.
(b) The Company does not have outstanding any bonds, debentures, notes or other obligations the holders of which have the right to vote (or convertible into or exercisable for securities having the right to vote) with the shareholders of the Company on any matter.
(c) All Shares subject to issuance upon due exercise of a Company Stock Option or Warrant, upon issuance on the terms and conditions specified in the instruments pursuant to which they are issuable, will be duly authorized, validly issued, fully paid and non-assessable.
(d) Schedule 3.3 sets forth a list of (i) all holders of Company Stock Options and Warrants, (ii) the number of shares subject to a Company Stock Option or Warrant, (iii) the exercise price of each Company Stock Option and Warrant and (iv) the date the Company Stock Option or Warrant expires, each as of August 25, 2014.
3.4 Other Interests. Except as set forth on <u>Schedule 3.4</u> of the Company Disclosure Schedule, the Company does not own, directly or indirectly, any interest or investment in any corporation, partnership, joint venture, business, trust, or entity other than investments in short term investment securities.
3.5 No Violation; Governmental Consents.
(a) Neither the execution and delivery by the Company of this Agreement and all agreements and documents contemplated hereby nor the performance of this Agreement or the consummation by the Company of the transactions contemplated hereby or thereby in accordance with the terms hereof or thereof will: (i) conflict with or result in a breach of any provisions of the Articles of Incorporation or Bylaws of the

Company; (ii) except as set forth on Schedule 3.5 of the Company Disclosure Schedule, violate, conflict with, result in a breach of any provision of, constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, result in the termination or in a right of termination or cancellation of, accelerate the performance required by, result in the triggering of any payment or other obligations pursuant to, result in the creation of any lien, security interest, charge or encumbrance upon any of the properties of the Company under, or result in being declared void, voidable, or without further binding effect, any of the terms, conditions, or provisions of any note, bond, mortgage,

indenture, loan agreement, deed of trust, or any license, franchise, permit,

Table of Contents

lease, contract, agreement or other instrument, commitment, or obligation to which the Company is a party, or by which the Company or any of its properties is bound or affected; (iii) violate any law, statute, rule, regulation, judgment, or decree applicable to the Company; or (iv) subject to compliance with the requirements set forth in Section 3.5(b) and in the case of the consummation of the Merger, obtaining the Requisite Company Vote, require any Consent (as defined in Section 3.5(b)) of any Governmental Entity (as defined in Section 3.5(b)), except, in the case of each of clauses (ii), (iii) and (iv), where any conflicts, violations, breaches, defaults, alterations, terminations, amendments, accelerations, cancellations or Liens, or where the failure to obtain any Consents, in each case, would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

- (b) No consent, approval, order or authorization of, or registration, declaration or filing with, or notice to (any of the foregoing being a *Consent*), any supranational, national, state, municipal, local or foreign government, any instrumentality, subdivision, court, administrative agency or commission or other governmental authority, or any quasi-governmental or private body exercising any regulatory or other governmental or quasi-governmental authority (a *Governmental Entity*) is required to be obtained or made by the Company in connection with the execution, delivery and performance by the Company of this Agreement or the consummation by the Company of the Merger and other transactions contemplated hereby, except for: (i) the filing of the Articles of Merger with the Secretary of State of the State of Nevada; (ii) the filing of the Company Proxy Statement with the Securities and Exchange Commission (*SEC*) in accordance with the Securities Exchange Act of 1934, as amended and the rules and regulations promulgated thereunder (the *Exchange Act*), and such reports under the Exchange Act as may be required in connection with this Agreement, the Merger and the other transactions contemplated by this Agreement; (iii) the other Consents of Governmental Entities listed in Schedule 3.5 of the Company Disclosure Schedule; and (iv) such other Consents which if not obtained or made would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- (c) The Company Board has taken all required actions such that the restrictions on business combinations contained in any anti-takeover Laws and regulations of any Governmental Entity will not apply with respect to or as a result of the execution of this Agreement or the consummation of the transactions contemplated hereby, including the Merger, without any further action on the part of the shareholders or the Company Board.
- 3.6 SEC Reports; Financial Statements.
- (a) The Company has timely filed with or furnished to, as applicable the SEC all reports, schedules, forms, statements and other documents required to be filed or furnished by the Company under the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder (the Securities Act) and the Exchange Act, including pursuant to Section 13(a) or 15(d) thereof, for the three (3) years preceding the date hereof (the foregoing materials, including the exhibits thereto and documents incorporated by reference therein, being collectively referred to herein as the Public Filings). As of their respective dates, the Public Filings complied in all material respects with the requirements of the Securities Act or Exchange Act, as applicable, and none of the Public Filings, when filed, contained any untrue statement of

Table of Contents

a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of
the circumstances under which they were made, not misleading. The Company is in compliance, in all material respects, with all the provisions
of the Sarbanes-Oxley Act of 2002 that are applicable to it.

- The financial statements of the Company included in the Public Filings comply in all material respects with applicable accounting requirements and the rules and regulations of the SEC with respect thereto as in effect at the time of filing. Such financial statements have been prepared in accordance with United States generally accepted accounting principles applied on a consistent basis during the periods involved (*GAAP*), except as may be otherwise specified in such financial statements or the notes thereto and except that unaudited financial statements may not contain all footnotes required by GAAP, and fairly present in all material respects the financial position of the Company as of and for the dates thereof and the results of operations and cash flows for the periods then ended, subject, in the case of unaudited statements, to normal, immaterial, year-end audit adjustments as permitted by GAAP and the applicable rules and regulations of the SEC.
- (c) The Company maintains a system of internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) that is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements in accordance with GAAP.
- Each document required to be filed by the Company with the SEC or required to be distributed or otherwise disseminated to the Company s shareholders in connection with the Merger, including the proxy statement to be filed with the SEC relating to the Company Shareholders Meeting (such proxy statement, as amended or supplemented from time to time, the *Proxy Statement*) to be held in connection with the adoption of this Agreement, if required, a Schedule 13E-3, and any amendments or supplements thereto, when filed, distributed or disseminated, as applicable, will comply as to form and substance in all material respects with the requirements of the Exchange Act and the rules and regulations thereunder. The information supplied or to be supplied by the Company specifically for inclusion or incorporation by reference in the Proxy Statement and any required Schedule 13E-3 will not, at the date it is first mailed to the Company s shareholders and at the time of the Company Shareholders Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated in the Proxy Statement and the Schedule 13E-3 or necessary in order to make the statements in the Proxy Statement and the Schedule 13E-3, in light of the circumstances under which they are made, not misleading. Notwithstanding anything to the contrary in this section, no representation or warranty is made by the Company with respect to information contained or incorporated by reference in the Proxy Statement or the Schedule 13E-3 supplied by or on behalf of Parent or Merger Sub specifically for inclusion or incorporation by reference in the Proxy Statement or the Schedule 13E-3.
- 3.7 **Absence of Undisclosed Liabilities.** The Company does not have any obligation or liability of any kind (whether accrued, absolute, contingent, unliquidated, civil, criminal, or otherwise and whether due or to become due), including accrued payroll, commissions, bonuses, retirement benefits or other employee benefit costs and vacation accruals,

Table of Contents

except for Permitted Liens (as defined in Section 3.10 below).

whether or not any such liability or obligation would have been required to be disclosed on a balance sheet prepared in accordance with GAAP, except as and to the extent reflected or reserved against in the Company s Public Filings, or incurred in the ordinary course of business since March 31, 2014 or would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.	
	Absence of Certain Changes or Events. Since March 31, 2014, except as set forth on Schedule 3.8 of the Company re Schedule, and except as specifically permitted or required by this Agreement or specifically consented to in writing by Parent and ub, the Company has not:
	declared, set aside, paid, or made any dividend or other distribution on or in respect of any shares of its capital stock or directly or redeemed, retired, purchased, or otherwise acquired any such shares or any option, warrant, conversion privilege, preemptive right, or agreement to acquire the same or any other securities convertible into or evidencing the right to purchase or otherwise acquire the
(b)	made any amendments to its Articles of Incorporation or Bylaws:
-	made any change in the number of shares of its capital stock authorized, issued, or outstanding, or authorized, issued or granted, any carrant, conversion privilege, preemptive right, or other right to acquire the same or any other securities convertible into or evidencing to acquire the same;
(d)	incurred any indebtedness for borrowed money;
(e) course of Schedule	incurred any obligation or liability (contingent or otherwise) except (i) normal trade or business obligations incurred in the ordinary business, and (ii) obligations under the contracts, agreements and leases described on Schedule 3.9 of the Company Disclosure ;
(f) paid to un	discharged or satisfied any lien or encumbrance or paid any obligations or liability (fixed or contingent) other than current liabilities arrelated parties, wages paid to officers and employees and director s fees paid to directors, each in the ordinary course of business;
(g)	mortgaged, pledged, or subjected to any lien, charge, or other encumbrance any of its properties or assets (tangible or intangible)

(h)	sold, assigned, leased, transferred or otherwise disposed of, or agreed to sell, assign, lease, transfer or
otherwise	e dispose of, any of its tangible assets other than sales of inventory in the ordinary course of business;
(i)	other than in the ordinary course of business, sold, assigned, licensed, transferred, or otherwise disposed of, or agreed to sell, assign.
()	ransfer or otherwise
neense, u	dister of otherwise
	A-12

Table of Contents

dispose of, any of its patents, inventions, shop rights, know-how, trade secrets, confidential information, registered or unregistered trademarks, service marks, logos, corporate names, trade names, and other trademark rights, trade dress, or other designations or combinations of such designations that are distinctive of its goods or services and that are used by the Company in a manner that identifies its goods or services and distinguishes them from the goods or services of others, works of authorship and any registered or unregistered copyright therein, or other intangible assets, and all registrations for, and applications for registration of, any of the foregoing (collectively, <i>Proprietary Rights</i>) or disclosed any of its confidential Proprietary Rights to any person (other than Parent and Merger Sub);
(j) entered into any transaction, contract, or commitment other than in the ordinary course of business;
(k) made any capital expenditures or any commitment therefor in excess of \$50,000 in the aggregate;
(l) adopted or made any change in any executive compensation plan, bonus plan, incentive compensation plan, deferred compensation agreement, or other employee benefit plan or arrangement;
(m) entered into any employment or consulting agreement or arrangement, or, except for normal bonuses or wage or salary increases pursuant to and consistent with existing plans or programs and consistent with past practices, granted or paid any bonus, or made or granted any general wage or salary increase or any specific increase in the wages or salary of any employee;
(n) suffered any casualty loss or damage, whether or not such loss or damage will have been covered by insurance;
(o) canceled or compromised any debt or claim except for adjustments made in the ordinary course of business that, in the aggregate, are not material, or waived or released any rights that are material;
(p) terminated, amended, or modified any agreement or instrument described on <u>Schedule 3.9</u> of the Company Disclosure Schedule;
(q) entered into any transaction with any shareholder, officer, director, or key employee of the Company or any Affiliate of any such person other than the payment of wages and salaries and other benefits under employee benefit plans in existence as of the date hereof;

made any loans or advances to, guaranties for the benefit of, or investments in, any person (other than customary wage or travel

advances in accordance with past practices which are less than \$10,000 in the aggregate);

(s) aggreg	suffered the loss of services of any employee, whether voluntarily or involuntarily, which loss or losses, individually or in the ate, has had, or would reasonably be expected to have, a Company Material Adverse Effect;
(t)	had any material change in its relations with its employees;
(u)	merged or consolidated with, or acquired all or substantially all of the assets, capital stock, or business of any other person;
-	there has not been any event, circumstance, change, occurrence, state of facts or effect (including the incurrence of any liabilities of ture, whether or not accrued, contingent or otherwise) that would reasonably be expected to have, individually or in the aggregate, a my Material Adverse Effect; or
(w)	agreed or committed to do any of the things described in this Section 3.8.
indebte Compa or are o Regula written	Contracts. Schedule 3.9 of the Company Disclosure Schedule contains a list of all written or oral contracts, tments, leases, and other agreements (including, without limitation, promissory notes, loan agreements, and other evidences of edness, guarantees, agreements with distributors, suppliers, dealers, franchisors and customers, and service agreements) to which the any is a party or by which the Company or its properties are bound pursuant to which the obligations thereunder of either party thereto are, contemplated as being, \$50,000 or more in any one fiscal year or any material contract (as such term is defined in Item 601(b)(10) of tion S-K of the Securities Act) (collectively, the <i>Contracts</i>). The Company has delivered to Parent a correct and complete copy of each agreement listed on Schedule 3.9 and a written summary setting forth the terms and conditions of each oral agreement referred to on the 3.9. With respect to each such Contract:
(a)	the Contract is legal, valid, binding, enforceable, and in full force and effect;
(b) continu	except as set forth on <u>Schedule 3.9</u> of the Company Disclosure Schedule, to the knowledge of the Company, the Contract will be to be legal, valid, binding, enforceable, and in full force and effect on identical terms following the consummation of the Merger;
(c) would	to the knowledge of the Company, no party is in breach or default, and no event has occurred which, with notice or lapse of time constitute a breach or default, or permit termination, modification, or acceleration, under the Contract;
(d)	no party has repudiated any provision of the Contract;

(e) the Company has not received any written claim of material default under any Contract and, to the knowledge of the Company, no fact or event exists that could give rise to any claim of material default under any Contract; and
(f) the Company has not received, as of the date of this Agreement, any notice in writing from any party that such party intends to terminate any Contract.
3.10 Title to Assets. Except as set forth on <u>Schedule 3.10</u> of the Company Disclosure Schedule (the <i>Permitted Liens</i>), the Company has good title, free and clear of all liens, charges, and encumbrances (except for (a) liens for non-delinquent Taxes and assessments, and (b) liens of landlords and other lessors arising under statute), to, or a valid leasehold interest in, all of the personal property (i) reflected on the Company balance sheet as of March 31, 2014 or acquired by the Company subsequent to the date thereof (other than assets that have been sold or otherwise disposed of in the ordinary course of business since March 31, 2014), or (ii) used in the business or operations of the Company.
3.11 Adequacy of Assets. The Company owns or has a valid leasehold interest in all assets, real and personal properties, contract rights and licenses necessary for the continued operation of the Company in substantially the same manner in which it has been and is now operating. Except as set forth on Schedule 3.11: (a) the Company owns or has the right to use pursuant to license, sublicense, agreement, or permission all Proprietary Rights (as defined in Section 3.8) necessary for the operation of the business of the Company as presently conducted, (b) each Proprietary Right owned or used by the Company immediately prior to the Effective Time will be owned or available for use by the Company on substantially similar terms and conditions immediately subsequent to the Effective Time hereunder, and (c) the Company has taken all reasonable action to maintain and protect each Proprietary Right that it owns or uses.
3.12 Litigation, Claims and Proceedings. Except as set forth on <u>Schedule 3.12</u> of the Company Disclosure Schedule, there are no actions, suits, legal or administrative proceedings or investigations pending or, to the knowledge of the Company, threatened, against or relating to the Company, its officers, directors or employees, its properties, assets or business or the transactions contemplated by this Agreement, other than any such actions that would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
3.13 Taxes.
(a) Except as set forth on Schedule 3.13(a) of the Company Disclosure Schedule, all returns and reports of all Taxes (as hereinafter defined) required to be filed by the Company have been timely filed and are true, correct, and complete in all material respects, and all Taxes payable pursuant thereto have been timely paid.
(b) Except as set forth on <u>Schedule 3.13(b)</u> of the Company Disclosure Schedule, no deficiency or adjustment in respect of any Taxes that was assessed against the Company remains unpaid and no such claim or assessment is pending or, to the knowledge of the Company, threatened.

The Company has made all withholding of Taxes required to be made under all applicable federal, state, and local tax regulations and such withholdings have either been paid on a timely basis to the respective governmental agencies or set aside in accounts for such purpose or accrued, reserved against and entered upon the books of the Company. There are no outstanding agreements or waivers extending the statutory period of limitations applicable to any tax return or tax liability of the Company, and to the knowledge of the Company, there is no proposed liability for any Taxes for which there is not an adequate reserve reflected on the Company financial statements. The unpaid Taxes of the Company (i) did not, as of the end of the most recent period for which financial statements have been prepared, exceed the reserve for Tax liability (rather than any reserve for deferred Taxes established to reflect timing differences between book and Tax income), and (ii) do not exceed that reserve as adjusted for the passage of time through the Effective Time in accordance with the past custom and practice of the Company in filing its Tax returns.
(d) There is no dispute or claim concerning any Tax liability of the Company either (i) claimed or raised by any authority in writing, or (ii) as to which the Company has knowledge.
(e) The Company has delivered to Parent correct and complete copies of all federal and state income Tax returns of the Company, examination reports and statements of deficiencies assessed against or agreed to by the Company for the five (5) years preceding the date hereof. The Company is not a party to any Tax allocation or sharing agreement.
(f) The Company has not been a member of an affiliated group filing federal income tax returns, and does not have any liability for the Taxes of any person (other than the Company).
(g) No item of income attributable to transactions occurring on or before the Effective Time will be required to be included in taxable income by the Company in a subsequent taxable year by reason of the Company reporting income on the installment sales method of accounting, the cash method of accounting, the completed contract method of accounting or the percentage of complete-capitalized cost method of accounting.
(h) Except as set forth on <u>Schedule 3.13(h)</u> of the Company Disclosure Schedule, the Company has not made, and is under no obligation to make, payments that are or, if made, would be, <i>parachute payments</i> under Section 280G of the Code, and no such obligation will arise as a result of this Agreement or other agreements entered into in connection with this Agreement.
(i) No audit or other administrative or judicial proceeding with respect to any Taxes due from the Company or any Tax Return of the Company is pending or threatened in writing by any Governmental Entity. Any assessed deficiency resulting from any audit or examination relating to Taxes by any Governmental Entity has been timely paid.
A-16

(j) payable.	There are no encumbrances for unpaid Taxes on the assets of the Company, except for encumbrances for Taxes not yet due and
(k) the Compa	No claim has been made in writing by a taxing authority in a jurisdiction where the Company does not currently file Tax Returns that ny is or may be subject to taxation by that jurisdiction.
payroll-rela and interes	Taxes means all federal, state, county, local, foreign and other taxes and governmental assessments, including but not limited to, ess, estimated taxes, withholding taxes, transfer taxes, excise taxes, sales taxes, real and personal property taxes, ad valorem taxes, atted taxes, employment taxes, franchise taxes and import duties, together with any related liabilities penalties, fines or additions to tax it. Tax Return means any report, return (including information return), claim for refund, election or declaration required to be supplied rernmental Entity with respect to Taxes, including any schedule or attachment thereto, and including any amendments thereof.
3.14	Proprietary Rights.
knowledge Proprietary written cha claim that t	Except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, and to the of the Company, (i) the Company has not interfered with, infringed upon, misappropriated, or otherwise come into conflict with any Rights of third parties; (ii) the Company (and its employees with responsibility for Proprietary Rights matters) has not received any arge, complaint, claims, demand, or notice alleging any such interference, infringement, misappropriation, or violation (including any the Company must license or refrain from using any Proprietary Rights of any third party); and (iii) no third party has interfered with, upon, misappropriated, or otherwise come into conflict with any Proprietary Rights of the Company.
Proprietary identifies e other perm nondiscloss trade name Company I	Schedule 3.14(b) of the Company Disclosure Schedule identifies each Proprietary Right that the Company owns. Schedule the Company Disclosure Schedule identifies each registration that has been issued to the Company with respect to any of its Rights, identifies each pending application for registration that the Company has made with respect to any of its Proprietary Rights, each item of copyrightable Proprietary Rights (whether or not registration has been sought), and identifies each license, agreement, or ission which Company has granted to any third party with respect to any of its Proprietary Rights, including any service, ure or escrow agreements (together with any exceptions). Schedule 3.14(b) of the Company Disclosure Schedule also identifies each or unregistered trademark used by Company in connection with its business. Except as set forth on Schedule 3.14(b) of the Disclosure Schedule, with respect to each Proprietary Right required to be identified on Schedule 3.14(b) of the Company Disclosure nd with respect to each item of copyrightable Proprietary Rights:
(i) license, or Effect;	the Company possesses all right, title, and interest in and to the Proprietary Right, free and clear of any lien, other restriction, except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse
	A 17

(ii)	the Proprietary Right is not subject to any outstanding injunction, judgment, order, decree, ruling, or charge;
	no Action is pending or, to the knowledge of Company, threatened that challenges the legality, validity, use or the ability to authorize the use (including use by way of reproduction of copies, distribution of copies, orks, public performance or display, or any other use of any kind or nature), sale or ownership of the Proprietary
	he Company has never agreed to indemnify any person for or against any interference, infringement, conflict with respect to the Proprietary Right.
	Schedule 3.14(c) of the Company Disclosure Schedule identifies each Proprietary Right that any third ny uses pursuant to license, sublicense, agreement, or permission. Except as set forth on Schedule 3.14(c) of the ule, with respect to each Proprietary Right required to be identified on Schedule 3.14(c) of the Company Disclosure
(i) enforceable against the Con	the license, sublicense, agreement, or permission covering the Proprietary Right is legal, valid, binding, apany and against the other parties thereto, and in full force and effect;
(ii) Company and against the of	the license, sublicense, agreement, or permission will continue to be legal, valid, binding, enforceable against the her parties thereto, and in full force and effect on substantially similar terms following the Effective Time;
	to the knowledge of the Company, no party to the license, sublicense, agreement, or permission is in breach or curred which with notice or lapse of time would constitute a breach or default or permit termination, modification, or
(iv) I	no party to the license, sublicense, agreement, or permission has repudiated in writing any provision thereof;
(v) true and correct with respec	with respect to each sublicense, the representations and warranties set forth in subsections (i) through (iv) above are to the underlying license;
(vi) t judgment, order, decree, rul	o the knowledge of the Company, the underlying Proprietary Right is not subject to any outstanding injunction, ing, or charge;

Table of Contents

(vii) enforceability of the u	no Action is pending or, to the knowledge of the Company, threatened, that challenges the legality, validity, or inderlying Proprietary Right;
(viii) permission; and	the Company has not granted any sublicense or similar right with respect to the license, sublicense, agreement, or
(ix) upgrade or similar obl of ninety (90) days or	no such licenses, agreements or permissions commit the Company to continued maintenance, support, improvement, ligation with respect to any of the Proprietary Rights, which obligation cannot be terminated by the Company upon notice less.
(d) Right necessary for th	Schedules 3.14(b) and 3.14(c) of the Company Disclosure Schedule identify every material Proprietary e business of the Company as presently conducted.
conditions of employr harassment, retaliation	Labor Relations. None of the employees of the Company are represented by any labor union or organization. The fance in all material respects with all federal and state Laws respecting employment and employment practices, terms and ment, wages and hours, hiring, termination of employment, plant closing and mass layoff, employment discrimination, and reasonable accommodation, leaves of absence, employee health and safety, leasing and supply of temporary and gement of independent contractors, including proper classification of same, and payroll taxes and has not engaged in any
	Transactions with Affiliates. Except as set forth on <u>Schedule 3.16</u> of the Company Disclosure Schedule, the Company ontract, lease, agreement, or other commitment with any officer, director, or shareholder of the Company or any Affiliate and there are no loans outstanding from the Company to, or to the Company from, any such person or any Affiliate of any
3.17	Permits; Compliance with Laws; Regulatory Compliance.
consents, and authoriz Company Disclosure	8.17(a) of the Company Disclosure Schedule contains a true, correct, and complete list of all material permits, licenses, cations of any government or governmental authority held by the Company. Except as set forth on Schedule 3.17(a) of the Schedule, the Company holds all material permits, licenses, consents, and authorizations issued by any government or by that are necessary for the conduct of its business.
	any has not received any written notice of any violation of any law, ordinance, regulation, permit or order applicable to the less conducted by it, except where such non-compliance would not reasonably be expected to have, individually or in the

aggregate, a Company Material Adverse Effect. The conduct of the business of the Company, as conducted currently and within the last five

(5) years prior to the Effective Time, complies with all applicable and material Laws, and the Company is not in violation in any

Table of Contents
material respect of any order, judgment or decree of any court, governmental authority or arbitration board or tribunal.
(c) Except as set forth on Schedule 3.17(c), the business of the Company as presently conducted does not require the Company to obtain any clearance or approval under the Food, Drug and Cosmetic Act (the FDCA) or the regulations of the U.S. Food and Drug Administration (FDA) promulgated thereunder.
3.18 Environmental Matters. Except as set forth on Schedule 3.18 of the Company Disclosure Schedule or as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, (i) the Company is in compliance with all applicable Environmental Laws and with all permits, licenses and approvals required under Environmental Laws for the conduct of business; (ii) there has been no Release of any Hazardous Material at any Real Property during the period such properties were owned, leased, or operated by the Company, or any disposal or treatment by or on behalf of the Company at a third party location, in either case in concentrations or under conditions or circumstances that would reasonably be expected to result in liability to the Company under any Environmental Laws or would require reporting, investigation, remediation, or other corrective or response action by the Company under an Environmental Law; (iii) there is no pending or, to the knowledge of the Company, threatened, claim, lawsuit, administrative proceeding, information request, written notice letter or investigation against the Company alleging a violation of, or liability under or pursuant to, any Environmental Law, which violation or liability is unresolved, nor is there any order, judgment or decree that imposes any obligations under any Environmental Law, including any obligation, expressly or by operation of law, for corrective or remedial action; and (iv) there are no underground tanks, PCBs, or asbestos containing materials on the Real Property.
3.19 Employee Plans and ERISA.
(a) Solely for purposes of this Section 3.19, any reference to Affiliate means and includes any trade or business (whether or not incorporated) under common control with the Company within the meaning of Section 414(b), (c), (m) or (o) of the Code.
(b) Schedule 3.19 of the Company Disclosure Schedule contains a true, correct and complete list of all material pension, benefit, profit sharing, retirement, deferred compensation, welfare, insurance, disability, bonus, vacation pay, severance pay and other similar plans, programs and agreements whether reduced to writing or not, relating to the employees or former employees of Company or its Affiliates (collectively the Employee Plans) under which or to which Company or any Affiliate contributes or has ever contributed or has been required to contribute for the benefit of its employees or former employees, and, except as set forth in Schedule 3.19, there are no Employee Plans with respect to which Company or any Affiliate has any obligations, contingent or otherwise, past or present, under applicable law or the terms of any such Employee Plan.
(c) Neither Company nor any Affiliate has contributed to, nor does Company or any Affiliate have any past or present obligation to contribute to, any pension plan subject to

Table of Contents

(a)

directives from any governmental bodies with jurisdiction over the Real

Title IV of the Employee Retirement Income Security Act of 1974 (<i>ERISA</i>) or any multiemployer plan as defined in Section 3(37) of ERISA.
The Employee Plans have been operated and maintained in all material respects in compliance with all applicable Laws including, but not limited to ERISA, the Code and the health care continuation provisions of the Consolidated Omnibus Budget Reconciliation Act of 1985 (<i>COBRA</i>). Company and its Affiliates have performed in all material respects all obligations required to be performed under, and are not in material violation in any respect of, and there has not been any material default or violation by any other party with respect to, any of the Employee Plans.
(e) Neither Company nor its Affiliates has any obligation to provide life or medical insurance benefits to former or retired employees or beneficiaries thereof, including but not limited to any obligations under COBRA.
(f) Neither Company nor its Affiliates, nor any of their directors, officers, employees or agents, or any party of interest or disqualified person, as such terms are defined in Section 3 of ERISA and Section 4975 of the Code, has, with respect to any Employee Plan, engaged in or been a party to any nonexempt prohibited transaction, as such term is defined in Section 406 of ERISA or Section 4975 of the Code which would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
(g) To the extent applicable, each Employee Plan and all amendments thereto intended to qualify under Section 401(a) of the Code have been determined by the Internal Revenue Service to so qualify, and the trusts created thereunder have been determined to be exempt from tax under the provisions of Section 501(a) of the Code.
(h) Nothing has occurred which might cause the loss of such qualification or exemption, and no such Employee Plan has been operated in a manner which would cause it to be disqualified in its operation.
(i) Except as set forth on Schedule 3.19(i), the consummation of the Merger will not, either alone or in combination with another event, (i) entitle any current or former employee or officer of the Company to any material severance pay, unemployment compensation or any other payment, except as expressly provided in this Agreement, or (ii) accelerate the time of payment or vesting, or materially increase the amount of compensation due any such employee or officer.
3.20 Properties.

Attached as Schedule 3.20 of the Company Disclosure Schedule is a description of all of the Company s interests in real property used

or occupied by it in the conduct of the Business (the *Real Property*). All Real Property leases used in the conduct of the business are described on <u>Schedule 3.20</u> of the Company Disclosure Schedule, are in full force and effect, and the Company holds a valid and existing leasehold interest under each of the leases for the terms set forth on such schedule. The Company has not received any notices, demands, or other

Table of Contents

Property asserting that any current or past use of or condition on or about the Real Property, or any part thereof, may violate, or otherwise give
rise to liability or costs under, Laws (including without limitation, Environmental Laws, as defined in Section 10.13) or any notice stating that
any part of the Real Property may be subject to condemnation or similar proceedings.

(b) Set forth on Schedule 3.20 of the Company Disclosure Schedule is a complete list of (a) each vehicle owned or leased by the Company and (b) each asset of the Company with a net book value or fair market value greater than \$100,000. The Company has good and marketable title to, or a valid leasehold interest in, all of its assets, including without limitation, the assets listed on Schedule 3.20 of the Company Disclosure Schedule, and all assets used by the Company in the conduct of its business, subject to no security interests, except as would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. All facilities, machinery, equipment, fixtures, vehicles and other tangible personal property owned, leased or used by the Company are in good operating condition and repair, normal wear and tear excepted, and are adequate and sufficient for the Company s business, as presently conducted.

3.21 Insurance.

- (a) Attached hereto as <u>Schedule 3.21(a)</u> of the Company Disclosure Schedule is a list of all policies of fire, casualty, liability, property or other forms of insurance and all fidelity bonds held by or applicable to the Company, which Schedule sets forth in respect of each such policy the policy name, policy number, carrier, term, type of coverage, deductible amount of self-insured retention amount, limits of coverage and annual premium. The Company has complied in all material respects with the terms and provisions of such policies.
- (b) There is no material claim pending under any of the Company s insurance policies as which coverage has been denied or disputed by the underwriters of such policies.
- (c) To the knowledge of the Company, there has been no threatened termination of, or material premium increase outside the ordinary course of business with respect to, any of the Company's insurance policies.
- (d) Except as set forth on <u>Schedule 3.21(d)</u>, the consummation of the Merger will not cause the termination or modification of any of the Company s insurance policies.
- 3.22 **Brokerage and Related Fees.** Except as set forth on <u>Schedule 3.22</u> of the Company Disclosure Schedule, no broker, investment banker, financial advisor or other person is entitled to any investment bankers fees, brokerage commissions, finders fees, or similar compensation in connection with the Merger based on any arrangement or agreement made by or on behalf of the Company.
- 3.23 **Knowledge.** Whenever a representation or warranty is made to the knowledge of the Company, or similar words, it will mean the actual knowledge of the executive officers of the Company after reasonable inquiry from those persons who report directly to such executive officers. Executive officers shall have the meaning set forth in Rule 3b- 7of the Exchange Act.

Table of Contents

Anti-Takeover Provisions. The Company is not party to a shareholders rights agreement, poison pill or similar agreement or plan. None of the requirements or restrictions of (a) the Nevada combinations with interested stockholders statutes, NRS 78.411 through 78.444, inclusive, or (b) the Nevada acquisition of controlling interest statutes, NRS 78.378 through 78.3793, inclusive (each, a Takeover Statute and together, the Takeover Statutes) would apply to prevent the consummation of any of the transactions contemplated by this Agreement, including the Merger. The Company Board has adopted such resolutions as are necessary so that the Takeover Statutes are rendered inapplicable to the Merger and the other transactions contemplated by this Agreement.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Parent and the Merger Sub herby jointly and severally represent and warrant to the Company as of the date of this Agreement as follows:

- 4.1 Existence; Good Standing; Corporate Authority.
- (a) Parent is a corporation duly incorporated, validly existing, and in good standing under the laws of Delaware, except where the failure to be in good standing would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect. Merger Sub is a corporation duly incorporated, validly existing, and in good standing under the laws of Nevada, except where the failure to be in good standing would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.
- (b) Each of Parent and Merger Sub has all requisite corporate power and authority to own, operate, and lease its properties and carry on its business as presently conducted and as proposed to be conducted.
- Authorization, Validity, and Effect of Agreements. Each of Parent and Merger Sub has the requisite corporate power and authority to execute and deliver this Agreement and all agreements and documents contemplated hereby. The consummation by each of Parent and Merger Sub of the transactions contemplated hereby has been duly authorized by all requisite corporate action. This Agreement has been duly executed and delivered by each of Parent and Merger Sub and, assuming the due authorization, execution and delivery by the Company, constitutes, and all agreements and documents contemplated hereby (when executed and delivered) will constitute, the valid and legally binding obligations of Parent and Merger Sub enforceable in accordance with their respective terms, except to the extent that enforceability may be limited by applicable bankruptcy, insolvency, moratorium, or similar Laws relating to creditors rights and general principles of equity.
- 4.3 No Violation; Consents.

(a) Neither the execution and delivery by Parent and Merger Sub of this Agreement and all agreements and documents contemplated hereby, nor the consummation by

Table of Contents

Parent and Merger Sub of the transactions contemplated hereby or thereby in accordance with the terms hereof or thereof, will: (i) conflict with or result in a breach of any provisions of the Articles of Incorporation or Bylaws of Parent or Merger Sub; (ii) violate any law, statute, rule, regulation, judgment, or decree applicable to Parent or Merger Sub (iii) violate, conflict with, result in a breach of any provision of, constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, result in the termination or in a right of termination or cancellation of, accelerate the performance required by, result in the triggering of any payment or other obligations pursuant to, result in the creation of any lien, security interest, charge or encumbrance upon any of the properties of Parent or Merger Sub under, or result in being declared void, voidable, or without further binding effect, any of the terms, conditions, or provisions of any note, bond, mortgage, indenture, loan agreement, deed of trust, or any license, franchise, permit, lease, contract, agreement or other instrument, commitment, or obligation to which Parent or Merger Sub is a party, or by which Parent or Merger Sub or any of their properties is bound or affected; or (iv) subject to compliance with the requirements set forth in Section 4.3(b), require any Consent of any Governmental Entity, except for any conflicts, violations, breaches, defaults, alterations, terminations, amendments, accelerations, cancellations or liens that would not reasonably be expected to have, individually or in the aggregate, a Parent Material Adverse Effect.

- (b) No Consent of any Governmental Entity is required to be obtained or made by Parent or Merger Sub in connection with the execution, delivery and performance by Parent and Merger Sub of this Agreement or the consummation by Parent and Merger Sub of the Merger and other transactions contemplated hereby, except for: (i) the filing of the Articles of Merger with the Secretary of State of the State of Nevada and appropriate documents with the relevant authorities of other states in which the Company and/or Parent are qualified to do business; (ii) the filing of the Company Proxy Statement with the SEC in accordance with the Exchange Act, and such reports under the Exchange Act as may be required in connection with this Agreement, the Merger and the other transactions contemplated by this Agreement; and (iii) such other Consents which if not obtained or made would not, individually or in the aggregate, reasonably be expected to have a Parent Material Adverse
- 4.4 **Litigation, Claims and Proceedings.** Except as set forth on <u>Schedule 4.4</u> of the Parent and Merger Sub Disclosure Schedule, there are no actions, suits, legal or administrative proceedings or investigations pending or threatened, against or relating to the Parent or Merger Sub, their officers, directors or employees, their properties, assets or business or the transactions contemplated by this Agreement, in each case that would, individually or in the aggregate, reasonably be expected to have a Parent Material Adverse Effect.
- 4.5 **Brokerage and Related Fees.** Except as set forth on <u>Schedule 4.5</u> of the Parent and Merger Sub Disclosure Schedule, no broker, investment banker, financial advisor or other person is entitled to any investment bankers fees, brokerage commissions, finders fees, or similar compensation in connection with the Merger based on any arrangement or agreement made by or on behalf of Parent or Merger Sub.

Table of Contents
4.6 Financial Capacity; Solvency.
(a) Parent has, and at the Effective Time will have, sufficient funds available, by combination of cash on hand and committed debt financing, to pay the Merger Consideration and otherwise consummate the Merger and perform Parent and Merger Sub s obligations hereunder.
(b) Immediately after giving effect to the Merger and the other transactions contemplated hereby, including the payment of the Merger Consideration and all other amounts required to be paid in connection with the consummation of the Merger and the other transactions contemplated hereby, assuming the accuracy of the representations and warranties of the Company set forth in Article III, the Surviving Corporation will be solvent, at and immediately after the Effective Time.
ARTICLE V
COVENANTS
5.1 Conduct of Business Prior to the Effective Time. From the date hereof until the Effective Time, except as otherwise provided in this Agreement or consented to in writing by Parent (which consent will not be unreasonably withheld or delayed), the Company will: (a) conduct the business of the Company in the ordinary course of business; and (b) use reasonable efforts to maintain and preserve intact the current organization, business and franchise of the Company and to preserve the rights, franchises, goodwill and relationships of its employees, customers, lenders, suppliers, regulators and others having business relationships with the Company. From the date hereof until the Effective Time, except as consented to in writing by Parent (which consent will not be unreasonably withheld or delayed), the Company will not take any action that would cause any of the changes, events or conditions described in Section 3.8 to occur.
Filings; Other Action. Subject to the terms and conditions herein provided, the Company and Parent will and will cause any appropriate other party to: (a) use reasonable efforts to cooperate with one another in (i) determining what filings, if any, are required to be made prior to the Effective Time with, and which Consents are required to be obtained prior to the Effective Time from any Governmental Entity in connection with the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby and (ii) timely making all such filings and timely seeking all such Consents and (b) use reasonable efforts to take, or cause to be taken, all other action and do, or cause to be done, all other things necessary, proper, or appropriate to consummate and make effective the transactions contemplated by this Agreement.
5.3 No Solicitation.
(a) The Company will not, and will not authorize or permit its directors, officers, employees, advisors and investment bankers (with respect to any person, the foregoing persons are referred to herein as such person s <i>Representatives</i>) to, directly or indirectly, solicit, initiate or knowingly take any action to facilitate or knowingly encourage the submission of any Takeover Proposal or the making of any proposal that

could reasonably be expected to

Table of Contents

lead to any Takeover Proposal, or, subject to Section 5.3(b), (i) enter into, participate or engage in, or maintain or continue any discussions or negotiations relating to, any Takeover Proposal with any person other than Parent or Merger Sub, (ii) furnish to any person other than Parent or Merger Sub any non-public information that the Company believes would be used for the purposes of formulating or furthering any Takeover Proposal, (iii) amend or grant any waiver or release under any standstill or similar agreement with respect to any class of equity securities of the Company, or (iv) enter into any agreement in principle, letter of intent, term sheet, acquisition agreement, merger agreement, option agreement, joint venture agreement, partnership agreement or other Contract relating to any Takeover Proposal (each, a *Company Acquisition Agreement*). The Company will cease immediately and cause to be terminated, and will not authorize or knowingly permit any of its or their Representatives to continue, any and all existing activities, discussions or negotiations, if any, with any third party conducted prior to the date hereof with respect to any Takeover Proposal and will use its reasonable efforts to cause any such third party (or its Agents) in possession of non-public information in respect of the Company that was furnished by or on behalf of the Company to return or destroy (and confirm destruction of) all such information.

- (b) Notwithstanding Section 5.3(a), prior to the receipt of the Requisite Company Vote, the Company Board, directly or indirectly through any Representative, may (i) participate in negotiations or discussions with any third party that makes a bona fide, unsolicited Takeover Proposal that the Company Board believes in good faith, after consultation with outside legal counsel and an independent financial advisor, constitutes or would reasonably be expected to result in a Superior Proposal, (ii) thereafter furnish to such third party non-public information relating to the Company pursuant to an executed confidentiality agreement that constitutes an Acceptable Confidentiality Agreement (a copy of which confidentiality agreement will be promptly provided for informational purposes only to Parent), and/or (iii) take any action that any court of competent jurisdiction orders the Company to take.
- (c) The Company will notify Parent promptly after receipt by the Company (or any of its Representatives) of any Takeover Proposal. In such notice, the Company will identify the third party making, and provide a reasonably detailed summary of the material terms and conditions of, any such Takeover Proposal.
- Subject to Sections 5.3(e), (f) and (g), from the date hereof to the earlier to occur of the termination of this Agreement in accordance with its terms and the Effective Time, neither the Company Board nor any committee thereof will (i) withhold, change, amend, withdraw or qualify (or modify in a manner adverse to Parent) or propose to withhold, change, amend, withdraw or qualify (or modify in a manner adverse to Parent) the approval, recommendation or declaration of advisability by the Company Board or any such committee of this Agreement, the Merger or any of the other transactions contemplated hereby (each such action set forth in this Section 5.3(d)(i) being referred to as a *Change of Board Recommendation*); (ii) adopt, approve, recommend or otherwise declare advisable the adoption of any Takeover Proposal, publicly propose to adopt, approve or recommend any Takeover Proposal or publicly take a neutral position or no position with respect to a Takeover Proposal; (iii) submit any Takeover Proposal or any matter related thereto to the vote of the shareholders of

the Company or (iv) resolve or agree to take any of the actions described in the foregoing clauses (i)-(iii).
(e) Notwithstanding anything to the contrary contained herein, but subject to Sections 5.3(f) and 5.3(g), prior to the date of approval of this Agreement by the shareholders of the Company, the Company Board may effect a Change of Board Recommendation:
(i) if the Company Board reasonably determines in good faith, after consultation with its outside legal counsel and its independent financial advisor, that, in light of an Intervening Event, the failure to take such action would reasonably be expected to breach the fiduciary duties owed by the Company Board to the shareholders of the Company under applicable Law; or
(ii) if the Company Board reasonably determines in good faith, after consultation with outside legal counsel and its independent financial advisor, that, in light of the Company s receipt of a Superior Proposal, the failure to take such action would reasonably be expected to breach the fiduciary duties owed by the Company Board to the shareholders of the Company under applicable Law.
(f) If the Company Board intends to effect a Change of Board Recommendation as a result of an Intervening Event, prior to making such Change of Board Recommendation, the Company will (a) provide written notice to Parent (an Intervening Event Notice) advising Parent that the Company Board intends to make such Change of Board Recommendation and specifying the nature of the Intervening Event and (b) to the extent possible given the nature of the Intervening Event, negotiate with and cause its financial and legal advisors to negotiate with Parent and its representatives in good faith during the five (5) Business Day period after Parent receives the Intervening Event Notice (to the extent Parent desires to negotiate) to make such adjustments in the terms and conditions of this Agreement, so that the Company Board would not be required to make such Change of Board Recommendation in order to meet its fiduciary duties owed to the shareholders of the Company under applicable Law.
A-27

Table of Contents

If the Company receives a Superior Proposal, it will (a) provide written notice to Parent (a Superior Proposal Notice) advising Parent that the Company Board has received a Superior Proposal, specifying the material terms and conditions of such Superior Proposal, identifying the party making such Superior Proposal and indicating that the Company Board intends to effect a Change of Board Recommendation and (b) negotiate with and cause its financial and legal advisors to negotiate with Parent and its representatives in good faith (to the extent Parent desires to negotiate) to make such adjustments in the terms and conditions of this Agreement, so that such third party proposal or offer would cease to constitute a Superior Proposal. The Company may only accept a Superior Proposal if, following the end of the five (5) Business Day period after Parent receives the Superior Proposal Notice, the Company Board (based upon advice by its outside financial advisor and legal counsel), determines in good faith, taking into account any changes to this Agreement proposal in writing by Parent in response to the Superior Proposal Notice or otherwise, that the Superior Proposal giving rise to the Superior Proposal Notice continues to constitute a Superior Proposal; provided, that any material modifications to such third party proposal or offer that the Company Board has determined to be a Superior Proposal will be deemed a new Superior Proposal and the Company will be required to again comply with the requirements of this section.

5.4 Shareholders Meeting; Preparation of Proxy Materials and Schedule 13E-3.

- (a) Subject to the terms set forth in this Agreement, the Company will take all action necessary to duly call, give notice of, convene and hold a special meeting of the holders of the Company Common Stock to consider the adoption of this Agreement (Company Shareholders Meeting) as soon as reasonably practicable after the date of this Agreement, and, in connection therewith, the Company will mail the letter to the shareholders, notice of meeting, proxy statement and forms of proxy (collectively, the Company Proxy Statement), to be filed with the SEC in connection with this Agreement to the holders of Company Common Stock in advance of such meeting. Subject to Section 5.3 hereof, the Company Proxy Statement will include the favorable recommendation from the Company Board. Subject to Section 5.3 hereof, the Company will use reasonable efforts to (i) solicit from the holders of Company Common Stock proxies in favor of the adoption of this Agreement and approval of the Merger and (ii) take all other actions necessary or advisable to secure the vote or consent of the holders of Company Common Stock required by applicable Law to obtain such approval. The Company will keep Parent and Merger Sub updated with respect to proxy solicitation results as requested by Parent and Merger Sub. Once the Company Shareholders Meeting has been called and noticed, the Company will not postpone or adjourn the Company Shareholders Meeting without the consent of Parent (other than (i) in order to obtain a quorum of its shareholders or (ii) as reasonably determined by the Company to comply with applicable Law). Notwithstanding anything contained herein to the contrary, the Company will not be required to hold the Company Shareholders Meeting if this Agreement is terminated before the meeting is held.
- (b) In connection with the Company Shareholders Meeting, as soon as reasonably practicable following the date of this Agreement the Company, with the assistance of Parent and Merger Sub, will prepare and file the Company Proxy Statement with the SEC. Concurrently with the preparation of the Proxy Statement, the Company, Parent and Merger Sub

Table of Contents

will jointly prepare and cause to be filed a Schedule 13E-3 with the SEC to the extent required by applicable Law. Each of the Company, Parent and Merger Sub will use its reasonable efforts so that the Company Proxy Statement and any required Schedule 13E-3 will comply as to form in all material respects with the requirements of the Exchange Act and the rules and regulations promulgated thereunder. Parent, Merger Sub and the Company will cooperate and consult with each other in the preparation of the Company Proxy Statement and any required Schedule 13E-3. Without limiting the generality of the foregoing, each of Parent and Merger Sub will furnish the Company the information relating to it required by the Exchange Act to be set forth in the Company Proxy Statement and any required Schedule 13E-3. The Company will use its reasonable efforts to resolve, and each party agrees to consult and cooperate with the other party in resolving, all SEC comments with respect to the Company Proxy Statement or Schedule 13E-3 as promptly as practicable after receipt thereof and to cause the Company Proxy Statement and any required Schedule 13E-3 in definitive form to be cleared by the SEC and mailed to the Company s shareholders as promptly as reasonably practicable following filing with the SEC. Each of Parent, Merger Sub and the Company agree to correct any information provided by it for use in the Company Proxy Statement and any required Schedule 13E-3 which will have become false or misleading and the Company will promptly prepare and mail to its shareholders an amendment or supplement setting forth such correction. The Company will as soon as reasonably practicable notify Parent of the receipt of any comments from the SEC with respect to the Company Proxy Statement and any required Schedule 13E-3 or for additional information.

- Notice of Certain Events. The Company will notify Parent and Merger Sub, and Parent and Merger Sub will notify the Company, promptly of (i) any notice or other communication from any person alleging that the consent of such person is or may be required in connection with the transactions contemplated by this Agreement, (ii) any notice or other communication from any Governmental Entity in connection with the transactions contemplated by this Agreement, (iii) any legal actions commenced, or to such party—s knowledge, threatened, against the Company or the Parent, as applicable, that are related to the transactions contemplated by this Agreement, and (iv) any event, change or effect between the date of this Agreement and the Effective Time which causes or is reasonably likely to cause the failure of the conditions set forth in Section 6.3 of this Agreement (in the case of the Company) or Section 6.2 of this Agreement (in the case of Parent and Merger Sub), to be satisfied. In no event will (x) the delivery of any notice by a party pursuant to this Section 5.5 limit or otherwise affect the respective rights, obligations, representations, warranties, covenants or agreements of the parties or the conditions to the obligations of the parties under this Agreement, or (y) disclosure by the Company or Parent be deemed to amend or supplement the Company Disclosure Schedule or constitute an exception to any representation or warranty. This Section 5.5 will not constitute a covenant or agreement for purposes of Section 6.2(a) or Section 6.3(a).
- 5.6 **Expenses.** Whether or not the Merger is consummated, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby will be paid by the party incurring such expenses.

Table of Contents

5.7 Company s bank accounts Effective Time.	Banking Matters. The Company will take all such action to remove the current authorized signatories from the and credit lines and to substitute officers of Merger Sub designated by Parent as such authorized signatories as of the
5.8 the Effective Time.	Directors of the Company. All directors of the Company will submit their resignations, effective concurrently with
public release or announce the Company and Parent (opermitted by Section 5.3 o Governmental Entity to wh	Public Announcements. The initial press release with respect to this Agreement and the transactions contemplated atually agreed to by the Company and Parent. Thereafter, each of the Company, Parent and Merger Sub agrees that no ment concerning the transactions contemplated hereby will be issued by any party without the prior written consent of which consent will not be unreasonably withheld or delayed), except as such release or announcement may be required by applicable Law or the rules or regulations of any applicable United States securities exchange or nich the relevant party is subject, wherever situated, in which case the party required to make the release or with the other party about, and allow the other party reasonable time to comment on such release or announcement in
5.10 In	demnification of Directors and Officers.
all past and present directors as of the date of this Agreed indemnification agreement arising out of acts or omisse. Parent and the Surviving Conduction with respect to the Company Articles of Incorporate the Parent or the Surviving the Parent or the Surviving this Section 5.8(a), to	six years from and after the Effective Time, Parent and the Surviving Corporation will indemnify and hold harmless rs, officers and employees of the Company (the <i>Indemnified Persons</i>) to the same extent such persons are indemnified ment by the Company pursuant to applicable Law, the Company Articles of Incorporation and Bylaws and s, if any, in existence on the date of this Agreement with any directors, officers and employees of the Company cions in their capacity as directors, officers or employees of the Company occurring at or prior to the Effective Time. Corporation will advance expenses (including reasonable legal fees and expenses) incurred in the defense of any matters subject to indemnification pursuant to this Section 5.8(a) in accordance with the procedures set forth in the poration, Bylaws and indemnification agreements, if any, in existence on the date of this Agreement. In no event riving Corporation be required to indemnify and hold harmless, or advance expenses to, the Indemnified Persons or any greater extent than that required under applicable Law, the Company Articles of Incorporation and Bylaws and s, if any, in existence on the date of this Agreement.
Surviving Corporation to c	six years from and after the Effective Time, Parent will cause the certificate of incorporation and bylaws of the ontain provisions no less favorable with respect to exculpation, indemnification and advancement of expenses of loyees of the Company for periods at or prior to the Effective Time than are currently set forth in the Company

Articles of Incorporation and Bylaws. Parent will cause the indemnification agreements, if any, in existence on the date of this

7D 1	1			c.	\sim			
Tal	٦I	\boldsymbol{e}	\cap 1	1		۱n	tei	ntc

following the Effective Time.
(c) For six years from and after the Effective Time, Parent will cause the Surviving Corporation to maintain for the benefit of the
Company s directors and officers, as of the date of this Agreement and as of the Effective Time, an insurance and indemnification policy that
provides coverage for events occurring prior to the Effective Time (the D&O Insurance) that is substantially equivalent to and in any event is
less favorable in the aggregate than the Company s existing policy (true and complete copies which have been previously provided to Parent) of

Agreement with any of the directors, officers or employees of the Company to continue in full force and effect in accordance with their terms

- provides coverage for events occurring prior to the Effective Time (the *D&O Insurance*) that is substantially equivalent to and in any event not less favorable in the aggregate than the Company s existing policy (true and complete copies which have been previously provided to Parent) or, if substantially equivalent insurance coverage is unavailable, the best available coverage; *provided, however*, that the Surviving Corporation will not be required to pay an annual premium for the D&O Insurance in excess of 300% of the last annual premium paid prior to the date of this Agreement. The provisions of the immediately preceding sentence will be deemed to have been satisfied if prepaid policies have been obtained prior to the Effective Time, which policies provide such directors and officers with coverage for an aggregate period of six years with respect to claims arising from facts or events that occurred on or before the Effective Time, including, without limitation, in respect of the transactions contemplated by this Agreement. If such prepaid policies have been obtained prior to the Effective Time, Parent will cause the Surviving Corporation to maintain such policies in full force and effect, and continue to honor the obligations thereunder.
- (d) In the event Parent or the Surviving Corporation (i) consolidates with or merges into any other person and will not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any person, then proper provision will be made so that such continuing or surviving corporation or entity or transferee of such assets, as the case may be, will assume the obligations set forth in this Section 5.10.
- (e) The obligations under this Section 5.10 will not be terminated or modified in such a manner as to adversely affect any indemnitee to whom this Section 5.10 applies without the consent of such affected indemnitee (it being expressly agreed that the indemnitees to whom this Section 5.10 applies will be third party beneficiaries of this Section 5.10).
- 5.11 **Parent Agreement Concerning Merger Sub.** Parent agrees to cause Merger Sub to comply with its obligations under this Agreement.
- 5.12 **Section 16 Matters**. Prior to the Effective Time, the Company Board, or an appropriate committee of non-employee directors thereof, will adopt a resolution consistent with the interpretive guidance of the SEC so that the disposition by any officer or director of the Company who is a covered Person of the Company for purposes of Section 16 of the Exchange Act (*Section 16*) of Shares, Company Stock Options or Warrants pursuant to this Agreement and the Merger will be an exempt transaction for purposes of Section 16. Prior to the Effective Time, the Company will take all such steps as may be required to cause each agreement, arrangement or understanding entered into by the Company on or after the date hereof with any of its officers, directors or employees pursuant to which consideration is paid to such officer, director or employee to be approved as an employment compensation, severance or other

employee benefit arrangement within the meaning of Rule 14d-10(d)(1) under the Exchange Act and to satisfy the requirements of the non-exclusive safe harbor set forth in Rule 14d-10(d) under the Exchange Act.
5.13 Stock Deregistration . The Company will cooperate with Parent, which will use reasonable efforts to take, or cause to be taken, all actions, and do or cause to be done all things, reasonably necessary, proper or advisable under applicable Laws and rules and policies of the SEC to cause the deregistration of the Shares under the Exchange Act as promptly as practicable after (and not in any event prior to) the Effective Time.
Takeover Statutes. If any Takeover Statute is or may become applicable to the Company, Parent or any of the transactions contemplated by this Agreement, the parties will use their respective reasonable efforts (a) to take all action necessary so that no Takeover Statute is or becomes applicable to the Company, Parent or any of the transactions contemplated by this Agreement and (b) if any such Takeover Statute is or becomes applicable to any of the foregoing, to take all action reasonably necessary (including, in the case of the Company and the Company Board, grant all necessary approvals or amend the Company s bylaws) so that the transactions contemplated by this Agreement may be consummated as promptly as practicable on the terms contemplated by this Agreement, including all actions to lawfully eliminate or minimize the effects of such statute, regulation or provision on the Merger and the other transactions contemplated by this Agreement.
Participation in Litigation. Prior to the Effective Time, (a) the Company will give prompt notice to Parent, and Parent will give prompt notice to the Company, of any Actions commenced or, to the knowledge of such party, threatened, against such party and/or its directors which relate to this Agreement or the transactions contemplated by this Agreement, and (b) the Company will give Parent the opportunity to participate in the defense or settlement of any shareholder Action against the Company and/or its directors relating to this Agreement or the transactions contemplated by this Agreement, at Parent s sole cost and expense, and no such Action will be settled or compromised without Parent s prior written consent (which consent should not be unreasonably withheld, conditioned or delayed).
ARTICLE VI
CONDITIONS
6.1 Conditions to Each Party s Obligation to Effect the Merger. The respective obligations of each party to effect the Merger will be subject to the fulfillment at or prior to the Closing Date of the following conditions:
(a) This Agreement and the Merger will have been approved by the Requisite Company Vote on or before December 31, 2014.
(b) None of the parties hereto will be subject to any order or injunction of a court of competent jurisdiction that prohibits the consummation of the transactions contemplated by this Agreement. In the event any such order or injunction will have been issued, each party agrees to use its reasonable efforts to have any such injunction lifted or order reversed.

(c) All consents, authorizations, orders, and approvals of (or filings or registrations with) any Governmental Entity required in connection with the execution, delivery, and performance of this Agreement will have been obtained or made, except for filings in connection with the Merger and any other documents required to be filed after the Effective Time.
(d) No Action will be pending or threatened by or before any court or governmental body in which an unfavorable judgment, order, or decree would prevent any of the transactions contemplated hereby or cause any such transaction to be declared unlawful or rescinded or that would reasonably be expected to cause a Company Material Adverse Effect.
6.2 Conditions to Obligation of the Company to Effect the Merger. In addition to the conditions set forth in Section 6.1 above, the obligation of the Company to effect the Merger will be subject to the fulfillment at or prior to the Closing Date of the following conditions:
(a) Each of Parent and Merger Sub will have performed in all material respects its agreements and covenants contained in this Agreement required to be performed on or prior to the Closing Date.
(b) Other than the representations and warranties of Parent and Merger Sub contained in Section 4.2 (Authorization, Validity, and Effect of Agreements), the representations and warranties of Parent and Merger Sub contained in this Agreement and in any document delivered in connection herewith (disregarding for this purpose any limitation or qualification by materiality or Parent Material Adverse Effect) will be true and correct as of the Closing Date, and the Company will have received a certificate of the President or a Vice President of each of Parent and Merger Sub, dated the Closing Date, certifying to such effect; <i>provided, however</i> , that notwithstanding anything herein to the contrary, this Section 6.2(b) will be deemed to have been satisfied even if such representations or warranties are not true and correct, unless the failure to perform or the failure of any of the representations or warranties to be so true and correct would have or would reasonably be likely to have, individually or in the aggregate, a Parent Material Adverse Effect.
(c) The representations and warranties of Parent and Merger Sub contained in Section 4.2 (Authorization, Validity, and Effect of Agreements) will be true and correct in all respects as of the date hereof and as of the Closing Date, as if made on the Closing Date.
(d) There will have been delivered to the Company certificates, dated within five days of the Closing Date, (i) of the Secretary of State of the State of Delaware, with respect to the incorporation and good standing of Parent and (ii) of the Secretary of State of the State of Nevada, with respect to the incorporation and good standing of Merger Sub.
(e) All consents and approvals of any third parties required in connection with the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby will have been obtained and delivered to the Company.

(f) There will have been delivered to the Company certificates, dated the Closing Date, of the President or Vice President and Secretary, respectively, of Parent and Merger Sub, attaching a true and complete copy of the resolutions of the Board of Directors of Parent and Merger Sub approving the execution and delivery of this Agreement and authorizing the consummation of the transactions contemplated hereby.
(g) Each of Parent and Merger Sub will have delivered to the Company a certificate of the President or a Vice President of Parent and Merger Sub, respectively, dated the Closing Date, certifying as to the satisfaction of the conditions specified in Sections 6.2(a), 6.2(b) and 6.2(c).
6.3 Conditions to Obligation of Parent and Merger Sub to Effect the Merger. In addition to the conditions set forth in Section 6.1 above, the obligations of Parent and Merger Sub to effect the Merger will be subject to the fulfillment at or prior to the Closing Date of the following conditions:
(a) The Company will have performed in all material respects its agreements and covenants contained in this Agreement required to be performed on or prior to the Closing Date.
(b) Other than the representations and warranties of the Company contained in Section 3.2 (Authorization, Validity, and Effect of Agreements) and Section 3.3 (Capitalization), the representations and warranties of the Company contained in this Agreement and in any document delivered in connection herewith (disregarding for this purpose any limitation or qualification by materiality or Company Material Adverse Effect) will be true and correct as of the Closing Date as if made on the Closing Date (except to the extent expressly made as of an earlier date, in which case as of such earlier date); <i>provided, however</i> , that notwithstanding anything herein to the contrary, this Section 6.3(b) will be deemed to have been satisfied even if such representations or warranties are not true and correct, unless the failure to perform or the failure of any of the representations or warranties to be so true and correct would have or would be reasonably likely to have a Company Material Adverse Effect.
(c) The representations and warranties of the Company contained in Section 3.2 (Authorization, Validity, and Effect of Agreements) and Section 3.3 (Capitalization) will be true and correct in all respects (except, in the case of Section 3.3, for de minimis inaccuracies) as of the date hereof and as of the Closing Date, as if made on the Closing Date (except to the extent expressly made as of an earlier date, in which case as of such earlier date).
A-34

Other than with respect to a default identified in the Company Disclosure Schedule as of the date of this Agreement, the Company will not be in default of any obligation, where said default cannot be cured by the Closing Date, under any of the Material Contracts, unless any such defaults, individually or in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect.	7
(e) There will have been delivered to Parent (i) a certificate of the Secretary of State of Nevada and any other state where the Company is qualified to do business as to the good standing of the Company as of a date not more than fifteen (15) Business Days prior to the Closing Date and (ii) a certificate of the Secretary of State of Delaware and any other state where Corgenix, Inc. is qualified to do business as to the good standing of Corgenix, Inc., as of a date not more than fifteen (15) Business Days prior to the Closing Date.	
(f) There will have been no change in the financial condition or results of operations of the Company from that reflected in the balance sheet of the Company so as to result, or be reasonably likely to result, in a Company Material Adverse Effect.	•
There will have been delivered to Parent a certificate dated as of the Closing Date, duly executed by the Secretary the Company given by him or her on behalf of the Company and not in his or her individual capacity, certifying as to an attached copy of the resolutions of the shareholders of the Company authorizing and approving the execution, delivery and performance of, and the consummation of the transactions contemplated by, this Agreement, and stating that such resolutions have not been amended, modified, revoked or rescinded.	ne
(h) There will have been delivered to Parent the consents set forth on <u>Schedule 3.5</u> .	
(i) There will have been delivered to Parent an agreement, in form and substance satisfactory to the Parent, confirming the terminatio of that certain First Amended and Restated Joint Product Development Agreement, by and between the Company, Financiere ELITech SAS, an Wescor, Inc., dated July 28, 2011, duly executed by the parties thereto.	
(j) There will have been delivered to the Parent evidence satisfactory to the Parent of the termination and release of the certain liens annotated on <u>Schedule 3.10</u> as subject to this Section 6.3(j).	
(k) No Action will be pending before any court or quasi-judicial or administrative agency of any federal, state, local, or foreign jurisdiction wherein an unfavorable injunction, judgment, order, decree, ruling, or charge would prevent consummation of the Merger or the transactions contemplated by this Agreement.	
(l) Dissenting Shares will not constitute more than ten percent (10%) of the Shares issued and outstanding as of the record date for, and entitled to vote at, the Company Shareholders Meeting.	

7D 1	1			c.	\sim			
Tal	٦I	\boldsymbol{e}	\cap 1	1		۱n	tei	ntc

(m) The Company will have delivered to Parent a certificate, dated the Closing Date, signed by a senior executive officer of the Company, certifying as to the satisfaction of the conditions specified in Sections 6.3(a), 6.3(b), 6.3(c) and 6.3(f).
6.4 Frustration of Closing Conditions None of the Company, Parent or Merger Sub may rely on the failure of any condition set forth in Article VI to be satisfied if such failure was caused by such party s failure to act in good faith to comply with this Agreement and consummate the Merger and the other transactions contemplated by this Agreement.
ARTICLE VII
SURVIVAL
7.1 Survival of Representations. None of the representations and warranties contained in this Agreement or in any instrument delivered under this Agreement will survive the Effective Time. This Section 7.1 does not limit any covenant of the parties to this Agreement which, by its terms, contemplates performance after the Effective Time. The confidentiality provisions of Article IX will (a) survive termination of this Agreement in accordance with their terms and (b) terminate as of the Effective Time.
ARTICLE VIII
TERMINATION
8.1 Termination By Mutual Consent. This Agreement may be terminated at any time prior to the Effective Time (notwithstanding any approval of this Agreement by the shareholders of the Company) by mutual written consent of Parent, Merger Sub and the Company.
8.2 Termination By Either Parent or the Company. This Agreement may be terminated by either Parent or the Company at any time prior to the Effective Time (notwithstanding any approval of this Agreement by the shareholders of the Company):
(a) if the Merger has not been consummated on or before December 31, 2014 (the <i>End Date</i>); <i>provided, however</i> , that the right to terminate this Agreement pursuant to this Section 8.2(a) will not be available to any party whose breach of any representation, warranty, covenant or agreement set forth in this Agreement has been the cause of, or resulted in, the failure of the Merger to be consummated on or before the End Date;

(b) if any Governmental Entity of competent jurisdiction will have enacted, issued, promulgated, enforced or entered any Law or Order making illegal, permanently enjoining or otherwise permanently prohibiting the consummation of the Merger or the other transactions contemplated by this Agreement, and such Law or Order will have become final and nonappealable; *provided, however*, that the right to terminate this Agreement pursuant to this Section 8.2(b) will not be available to any party whose breach of any representation, warranty,

covenant or agreement set forth in this Agreement has been the cause of, or resulted in, the issuance, promulgation, enforcement or entry of any such Law or Order; or
(c) if this Agreement has been submitted to the shareholders of the Company for adoption at a duly convened Company Shareholders Meeting and the Requisite Company Vote will not have been obtained at such meeting (including any adjournment or postponement thereof).
8.3 Termination By Parent. This Agreement may be terminated by Parent at any time prior to the Effective Time (notwithstanding any approval of this Agreement by the shareholders of the Company)
(a) if there will have been a breach of any representation, warranty, covenant or agreement on the part of the Company set forth in this Agreement such that the conditions to the Closing of the Merger set forth in Sections 6.3(a), 6.3(b) or 6.3(c) would not be satisfied and such breach is incapable of being cured by the End Date; <i>provided that</i> Parent will have given the Company at least 30 days written notice prior to such termination stating Parent s intention to terminate this Agreement pursuant to this Section 8.3;
(b) the Company Board will have effected and not withdrawn a Change of Board Recommendation; or
(c) if prior to the receipt of the Requisite Company Vote at the Company Shareholders Meeting, the Company Board authorizes the Company, in full compliance with the terms of this Agreement, including Section 5.3(b) hereof, to enter into a Company Acquisition Agreement (other than an Acceptable Confidentiality Agreement) in respect of a Superior Proposal.
8.4 Termination By the Company. This Agreement may be terminated by the Company at any time prior to the Effective Time (notwithstanding, in the case of Section 8.4(b) immediately below, any approval of this Agreement by the shareholders of the Company):
(a) if prior to the receipt of the Requisite Company Vote at the Company Shareholders Meeting, the Company Board authorizes the Company, in full compliance with the terms of this Agreement, including Section 5.3(b) hereof, to enter into a Company Acquisition Agreement (other than an Acceptable Confidentiality Agreement) in respect of a Superior Proposal; or
(b) if there will have been a breach of any representation, warranty, covenant or agreement on the part of Parent or Merger Sub set forth in this Agreement such that the conditions to the Closing of the Merger set forth in Section 6.2(a) or (b), as applicable, would not be satisfied and, in either such case, such breach is incapable of being cured by the End Date; <i>provided that</i> the Company will have given Parent at least 30 days written notice prior to such termination stating the Company s intention to terminate this Agreement pursuant to this Section 8.4(b).

	rmination. In the event of the termination of this Agreement in accordance with this Article, this and there will be no liability on the part of any party hereto except as set forth in this Article VIII,
8.6 Parent Tern	nination Fee.
(a) In the event that:	
withdrawn), after the date hereof and pri no Company Shareholders Meeting), (B terminated by the Company or Parent pu Agreement, the Company consummates	roposal or offer with respect to a Takeover Proposal will have been proposed or communicated (and not or to the Company Shareholders Meeting (or prior to the termination of this Agreement if there has been of following the occurrence of an event described in the preceding clause (A), this Agreement is ursuant to Section 8.2(a) or Section 8.2(c) and (C) within 12 months after the termination of this or enters into a definitive agreement in connection with, any such Takeover Proposal); provided that for ences to 20% in the definition of Takeover Proposal will be deemed to be references to 50%;
(ii) this Agreement is	terminated by Parent pursuant to Section 8.3; or
(iii) this Agreement is to	erminated by the Company pursuant to Section 8.4(a),
the out-of-pocket fees and expenses, not incurred by Parent, Merger Sub or their wire transfer of same day funds as prom a termination referred to in clause (ii), (I connection with a Takeover Proposal in	its designees an amount (the <i>Parent Termination Fee</i>) equal to the sum of (x) \$324,000, plus (y) all of to exceed \$400,000 (including, without limitation, all fees and expenses of counsel and accountants), Affiliates in connection with this Agreement and the transactions contemplated by this Agreement, by ptly as possible (but in any event (A) within five (5) Business Days after such termination in the case of B) within two (2) Business Days following the entry by the Company into the definitive agreement in the case of a termination referred to in clause (i) or (C) prior to or concurrently with the termination of ion pursuant to clause (iii)); <i>it being understood</i> that in no event will the Company be required to pay the e occasion.
	A-38

Table of Contents

- (b) In the event that the Company fails to pay the Parent Termination Fee when due and in accordance with the requirements of this Agreement, the Company will reimburse Parent for reasonable costs and expenses actually incurred or accrued by Parent (including fees and expenses of counsel) in connection with the collection under and enforcement of this Section 8.6, together with interest on such unpaid Parent Termination Fee commencing on the date that the Parent Termination Fee became due, at the prime rate as published in the Wall Street Journal Table of Money Rates on such date plus 3.00%. Such collection expenses will not otherwise diminish in any way the payment obligations hereunder.
- (c) Each of the Company, Parent and Merger Sub acknowledges that (i) the agreements contained in this Section 8.6 are an integral part of the transactions contemplated by this Agreement, (ii) the damages resulting from termination of this Agreement under circumstances where a Parent Termination Fee is payable are uncertain and incapable of accurate calculation and therefore, the amount payable pursuant to Section 8.6(a) is not a penalty but rather constitutes an amount akin to liquidated damages in a reasonable amount that will compensate Parent for the efforts and resources expended and opportunities foregone while negotiating this Agreement and in reliance on this Agreement and on the expectation of the consummation of the transactions contemplated by this Agreement and (iii) without the agreements contained in this Section 8.6, Parent would not have entered into this Agreement.

Tabl	e of	Con	tents
1 au	L OI	COII	wiits

8./ Company Terinination re	8.7	Company '	Termination Fe
-----------------------------	-----	-----------	----------------

- (a) In the event that this Agreement is terminated by the Company pursuant to Section 8.4(b), then Parent will pay to the Company or its designees an amount (the *Company Termination Fee*) equal to the sum of (x) \$648,000, plus (y) all of the out-of-pocket fees and expenses, not to exceed \$400,000 (including, without limitation, all fees and expenses of counsel and accountants), incurred by the Company or its Affiliates in connection with this Agreement and the transactions contemplated by this Agreement, by wire transfer of same day funds as promptly as possible (but in any event within five (5) Business Days after such termination; *it being understood* that in no event will Parent be required to pay the Company Termination Fee on more than one occasion.
- (b) In the event that Parent fails to pay the Company Termination Fee when due and in accordance with the requirements of this Agreement, Parent will reimburse the Company for reasonable costs and expenses actually incurred or accrued by the Company (including fees and expenses of counsel) in connection with the collection under and enforcement of this Section 8.7, together with interest on such unpaid Company Termination Fee commencing on the date that the Company Termination Fee became due, at the prime rate as published in the Wall Street Journal Table of Money Rates on such date plus 3.00%. Such collection expenses will not otherwise diminish in any way the payment obligations hereunder.
- (c) Each of the Company, Parent and Merger Sub acknowledges that (i) the agreements contained in this Section 8.7 are an integral part of the transactions contemplated by this Agreement, (ii) the damages resulting from termination of this Agreement under circumstances where a Company Termination Fee is payable are uncertain and incapable of accurate calculation and therefore, the amount payable pursuant to Section 8.7(a) is not a penalty but rather constitutes an amount akin to liquidated damages in a reasonable amount that will compensate the Company for the efforts and resources expended and opportunities foregone while negotiating this Agreement and in reliance on this Agreement and on the expectation of the consummation of the transactions contemplated by this Agreement and (iii) without the agreements contained in this Section 8.7, the Company would not have entered into this Agreement.
- 8.8 **Sole Remedy.** The parties hereto acknowledge and agree that the remedies provided in Section 5.10, Section 8.6, Section 8.7 and Section 10.10 will be the parties—sole and exclusive remedies for any breaches of this Agreement or any Action arising from or related to the transactions contemplated by this Agreement, other than in the case of fraud. Notwithstanding anything to the contrary in this Agreement, the parties hereby acknowledge that in the event that the Parent Termination Fee becomes payable and is paid by the Company pursuant to Section 8.6, the Termination Fee will be Parent—s and Merger Sub—s sole and exclusive remedy for monetary damages under this Agreement. Notwithstanding anything to the contrary in this Agreement, the parties hereby acknowledge that in the event that the Company Termination Fee becomes payable and is paid by the Parent pursuant to Section 8.7, the Company Termination Fee will be the Company—s sole and exclusive remedy for monetary damages under this Agreement.

m	. 1		c			
Tal	hI	e	ot	on	ıte	nts

ARTICLE IX

CONFIDENTIALITY

9.1 Confidentiality Covenants. Parent, Merger Sub and the Company acknowledge that each may have received access to Confidential Information (as hereinafter defined) of the other in the course of investigations and negotiations prior to Closing. Each party who receives any Confidential Information (a Receiving Party) from any other party hereto (the Disclosing Party), may disclose any such Confidential Information to such party s employees, attorneys, accountants, financial advisors or agents or representatives that have a need to know such Information to facilitate or assist with the consummation of the transactions contemplated hereby (collectively, Agents). Subject to the foregoing exception, and the exception hereinafter set forth in Section 9.2 below (i) a Receiving Party will keep, and will cause its Agents to keep, all Confidential Information received from a Disclosing Party hereunder strictly confidential and will not disclose, and will cause its Agents not to disclose, any such Confidential Information to any third party; and (ii) any Receiving Party and its Agents will not make any uses of Confidential Information received from a Disclosing Party except to facilitate or assist with the consummation of the transactions contemplated hereby. Confidential Information will include any business, financial, technical or other information, including, but not limited to, business plans, forecasts, marketing plans or initiatives, customer, client and vendor lists, training materials developed by the Disclosing Party, information regarding the identities, qualifications and compensation being paid to key employees, information received from customers, vendors or clients with the expectation, whether explicit or implicit, that such information would be protected from disclosure or dissemination to third parties, and other information the value of which to the Disclosing Party is dependent on the non-disclosure of such information. Confidential Information will not include information that, although disclosed or made available by a Disclosing Party or any of its Representatives to a Receiving Party or any of its Agents, (i) can be obtained by persons not subject to confidentiality or use restrictions from public sources, including periodicals, government and industry publications and other media that is readily accessible to the public or competitors of the Disclosing Party, (ii) has been disclosed by the Disclosing Party or any of its Agents to any unaffiliated third parties without the imposition of any restrictions or prohibitions on disclosure or use thereof and has been, as a result, disclosed by that third party to other third parties, or (iii) information that the Receiving Party can demonstrate convincingly was in its possession prior to its disclosure to the Receiving Party by the Disclosing Party or any of its Agents, provided that the Receiving Party had not obtained possession of such Confidential Information from any one that the Receiving Party knew or should have known was subject to restrictions on its right to disclose such information to the Receiving Party, either pursuant to an agreement or by reason of his position or relationship with the Disclosing Party.

Disclosure Pursuant to Legal Process. If a Receiving Party is required by subpoena or other legal process, or by Laws applicable to it, to disclose or produce any Confidential Information belonging to a Disclosing Party, then, the Receiving Party will (i) provide the Disclosing Party prompt notice thereof and copies, if possible, and, if not, a description, of the Confidential Information requested or required to be produced so that Disclosing Party may seek an order to quash such subpoena or other legal process or an appropriate protective order or may elect to waive compliance with the provisions of this Article IX as to any portion or all of such Confidential Information (ii) consult with the Disclosing Party as to the advisability of taking legally available steps to quash or narrow such request, and

Table of Contents

(iii) provide such reasonable cooperation as the Disclosing Party may request in connection with efforts by the Disclosing Party to quash the subpoena or other legal process or to obtain a protective order with respect to the Confidential Information being sought. If, in the absence of a protective order or the receipt of a waiver hereunder, a Receiving Party is nonetheless, in the opinion of his legal counsel, compelled to disclose or produce any such Confidential Information of the Disclosing Party to any tribunal legally authorized to request and entitled to receive such Confidential Information or to any government agency with which the Receiving Party is required by law to file any such Confidential Information or otherwise stand liable for contempt or suffer other censure or penalty or liability, the Disclosing Party may disclose or produce such Confidential Information to such tribunal or government agency, notwithstanding the fact that such information may, as a result become available to the public, without incurring liability hereunder to the Disclosing Party; *provided, however*, that the Receiving Party will give the Disclosing Party written notice of the Confidential Information to be so disclosed or produced as far in advance of its disclosure or production as is practicable and will use its reasonable efforts to obtain, to the greatest extent practicable, an order or other reliable assurance that confidential treatment will be accorded to such Confidential Information so required to be disclosed or produced.

9.3 **Termination of Confidentiality Obligations.** The obligations of the parties under this Article IX will terminate on the Closing of the transactions contemplated hereby. In the event of a termination of this Agreement, the respective obligations under this Article IX will survive for a period of five (5) years from the date of such termination.

ARTICLE X

GENERAL PROVISIONS

Notices. All notices, requests, consents, claims, demands, waivers and other communications hereunder will be in writing and will be deemed to have been given: (a) when delivered by hand (with written confirmation of receipt); (b) when received by the addressee if sent by a nationally recognized overnight courier (receipt requested); (c) on the date sent by facsimile or e-mail of a PDF document (with confirmation of transmission) if sent during normal business hours of the recipient, and on the next Business Day if sent after normal business hours of the recipient; or (d) on the third day after the date mailed, by certified or registered mail, return receipt requested, postage prepaid. Such communications must be sent to the respective parties at the following addresses (or at such other address for a party as shall be specified in a notice given in accordance with this Section 10.1):

If to the Company: 11575 Main Street, Suite 400

Broomfield, CO 80020 Facsimile: 303-457-4519

E-mail: Dsimpson@Corgenix.com and Wcritchfield@Corgenix.com Attention: Chief Executive Officer and Chief Financial Officer

Table of Contents

with a copy to: Husch Blackwell LLP (which will not Facsimile: 303-749-7272

constitute notice to E-mail: Robert.Attai@huschblackwell.com

the Company) Attention: Robert Attai

If to Parent or c/o Water Street Healthcare Partners Merger Sub: 333 West Wacker Drive, Suite 2800

Chicago, IL 60606

Attention: Rob Womsley and Jeff Holway

with a copy to: Much Shelist, PC
(which will not constitute notice to E-mail: mshaw@muchshelist.com

Parent or Merger Sub) Attention: Michael Shaw

and (which will not constitute notice to Parent or Merger Sub)

Kirkland & Ellis LLP 300 North LaSalle Chicago, IL 60654

Attention: Martin A. DiLoreto Jr., P.C.

10.2 Entire Agreement; No Third-Party Beneficiaries. This Agreement, the Exhibits, the Company Disclosure Schedule, the Parent and Merger Sub Disclosure Schedule, the confidentiality agreements between the parties hereto and any schedules or agreements delivered or to be delivered in connection with this Agreement (a) constitute the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings among the parties with respect thereto and (b) are not intended to confer upon any person other than the parties hereto any rights or remedies. No information previously provided, or any addition to or modification of any provision of this Agreement will be binding upon any party hereto unless made in writing and signed by all parties hereto.

Governing Law; Venue. This Agreement will be governed by, and construed in accordance with, the Laws of the State of Delaware (except that the NRS shall govern (a) the internal affairs of the Company and Merger Sub and any other entities that are organized and existing under the Laws of the State of Nevada, including the fiduciary duties of the directors and officers thereof, (b) the Merger, and (c) all other provisions of, or transactions contemplated by, this Agreement, or any other matters, to which the NRS mandatorily applies), without giving effect to any applicable principles of conflict of laws that would cause the laws of another jurisdiction to govern this Agreement. All actions arising out of or relating to this Agreement will be heard and determined exclusively in any state or federal court sitting in the State of Delaware. Consistent with the preceding sentence, each of the parties hereto hereby (x) submits to the exclusive jurisdiction of any federal or state court sitting in the State of Delaware for the purpose of any Action arising out of or relating to this Agreement brought by any party hereto and (y) irrevocably waives, and agrees not to assert by way of motion, defense, or otherwise, in any such Action, any claim that it is not subject personally to the jurisdiction of the above-named

courts, that the Action is brought in an inconvenient forum, that the venue of the Action is improper, or that this Agreement or the transactions contemplated by this Agreement may not be enforced in or by any of the above-named courts.
10.4 Counterparts. This Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered will be an original, but all such counterparts will together constitute one and the same instrument. Each counterpart may consist of a number of copies hereof each signed by less than all, but together signed by all of the parties hereto. Executed counterparts transmitted by fax or PDF signature will be effective as originals.
10.5 Headings. Headings of the Articles and Sections of this Agreement are for the convenience of the parties only, and will be given no substantive or interpretive effect whatsoever.
Interpretation. In this Agreement, unless the context otherwise requires, words describing the singular number will include the plural and vice versa, and words denoting any gender will include all genders and words denoting natural persons will include corporations and partnerships and vice versa. The parties have participated jointly in the negotiation and drafting of this Agreement with the advice of counsel. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the parties, and no presumption or burden of proof will arise favoring or disfavoring any party by virtue of the authorship of any provisions of this Agreement.
Waivers. Except as provided in this Agreement, no action taken pursuant to this Agreement, including, without limitation, any investigation by or on behalf of any party, will be deemed to constitute a waiver by the party taking such action of compliance with any representations, warranties, covenants or agreements contained in this Agreement. The waiver by any party hereto of a breach of any provision hereunder will not operate or be construed as a waiver of any prior or subsequent breach of the same or any other provision hereunder. No failure or delay by a party hereto in exercising any right, power or remedy under this Agreement, and no course of dealing between the parties hereto, will operate as a waiver of any such right, power or remedy of the party. No single or partial exercise of any right, power or remedy under this Agreement by a party hereto, nor any abandonment or discontinuance of steps to enforce any such right, power or remedy, will preclude such party from any other or further exercise thereof or the exercise of any other right, power or remedy hereunder. The election of any remedy by a party hereto will not constitute a waiver of the right of such party to pursue other available remedies. No notice to or demand on a party not expressly required under this Agreement will entitle the party receiving such notice or demand to any other or further notice or demand in similar or other circumstances or constitute a waiver of the rights of the party giving such notice or demand to any other or further action in any circumstances without such notice or demand.
10.8 Incorporation of Exhibits. The Company Disclosure Schedule, the Parent and Merger Sub Disclosure Schedule and all Exhibits and schedules attached hereto and referred to herein are hereby incorporated herein and made a part hereof for all purposes as if fully set forth herein.
A-44

Severability. Any term or provision of this Agreement which is invalid or unenforceable in any jurisdiction will, as to that jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms and provisions of this Agreement or affecting the validity or enforceability of any of the terms or provisions of this Agreement in any other jurisdiction unless the same is material to the terms of this Agreement, in the judgment of either party to this Agreement, in which case the parties will negotiate in good faith to revise the same so as to be valid or enforceable. If any provision of this Agreement is so broad as to be unenforceable, the provision will be interpreted to be only so broad as is enforceable.
10.10 Enforcement of Agreement. The parties hereto agree that irreparable damage would occur in the event that any provision of this Agreement was not performed in accordance with its specific terms or was otherwise breached, or were threatened to be breached. It is accordingly agreed that the parties will be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof, this being in addition to any other remedy to which they are entitled at law or in equity.
10.11 Consent. Whenever the consent or approval of a party is required by the terms of this Agreement, unless otherwise provided, the same will not be unreasonably withheld or delayed.
Assignment. Neither this Agreement nor any of the rights, interests or obligations under this Agreement will be assigned, in whole or in part, by operation of Law or otherwise by any of the parties without the prior written consent of the other parties. Any purported assignment without such consent will be null and void. Subject to the preceding sentences, this Agreement will be binding upon, inure to the benefit of, and be enforceable by, the parties and their respective successors and assigns.
Definitions. For purposes of this Agreement, the following terms will have the following meanings when used herein with initial capital letters:
Acceptable Confidentiality Agreement means a confidentiality and standstill agreement that contains confidentiality and standstill provisions that are no less favorable to the Company than those contained in any confidentiality agreement among the parties hereto.
Action means any litigation, suit, claim, arbitration, claim, demand letter, action, proceeding (whether at law, in equity, in contract in tort or otherwise) or other investigation, including any of the foregoing conducted by or before any Governmental Entity.
Affiliate means any person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, a person.
Business Day means any day, other than Saturday, Sunday or any day on which banking institutions located in Nevada are authorized or required by Law or other governmental action to close.

Table of Contents

Company Material Adverse Effect means any change, event, development, condition, occurrence or effect (x) that prevents or materially delays, or would reasonably be expected to prevent or materially delay, consummation of the Merger or performance by the Company of any of its material obligations under this Agreement or (y) that is, or would reasonably be expected to be, materially adverse to the business, financial condition, assets, liabilities or results of operations of the Company; provided, however, that none of the following will be deemed in themselves, either alone or in combination, to constitute, and that none of the following will be taken into account for purposes of the preceding clause (y) in determining whether there has been or will be, a Company Material Adverse Effect: (a) any change generally affecting the economy, financial markets or political, economic or regulatory conditions in the United States or any other geographic region in which the Company conduct business to the extent the Company is not materially and disproportionately affected thereby relative to other companies in the life sciences industry; (b) general financial, credit or capital market conditions, including interest rates or exchange rates, or any changes therein, to the extent the Company is not materially and disproportionately affected thereby relative to other companies in the life sciences industry; (c) any change attributable to the negotiation, execution, announcement, pendency or pursuit of the transactions contemplated hereby, including the Merger, including any litigation resulting therefrom, any cancellation of or delays in customer orders, and any disruption in supplier, distributor, partner or similar relationships; (d) any change arising from or relating to compliance with the terms of this Agreement, or action taken, or failure to act, to which Parent has consented; (e) acts of war (whether or not declared), the commencement, continuation or escalation of a war, acts of armed hostility, sabotage or terrorism or other international or national calamity or any material worsening of such conditions threatened or existing as of the date of this Agreement; (f) any hurricane, earthquake, flood, or other natural disasters or acts of God; (g) changes in Laws after the date hereof, to the extent the Company is not materially and disproportionately affected thereby relative to other companies in healthcare & pharmaceuticals industry; (h) changes in GAAP after the date hereof, to the extent the Company is not materially and disproportionately affected thereby relative to other companies in the life sciences industry; (i) any failure by the Company to meet any published or internally prepared estimates of revenues, earnings or other economic performance for any period ending on or after the date of this Agreement (it being understood that the facts and circumstances giving rise to such failure may be deemed to constitute, and may be taken into account in determining whether there has been, a Company Material Adverse Effect if such facts and circumstances are not otherwise described in clauses (a)- (h) or (j) of the definition); or (j) a decline in the price of the Company Common Stock on the OTCBB (it being understood that the facts and circumstances giving rise to such decline may be deemed to constitute, and may be taken into account in determining whether there has been, a Company Material Adverse Effect if such facts and circumstances are not otherwise described in clauses (a)- (i) of the definition).

Environmental Laws means all applicable federal, state and local statutes, laws, regulations, rules, ordinances, codes, licenses, permits, orders, standards, guidelines, policies, and similar items of any governmental authority having jurisdiction and all applicable judicial, administrative, and regulatory decrees, judgments, and orders and common law relating to the protection of human health and safety, natural resources or the environment including, without limitation, all requirements pertaining to the reporting, licensing, permitting, use, handling,

Table of Contents

generation, storage, treatment, transportation, disposal, release, discharge, investigation, and remediation of Hazardous Material.

Hazardous Material means any substance, material or agent, including any constituent, raw material, product or byproduct thereof, (i) the presence of which requires reporting, investigation or remediation under any applicable statute, regulation, ordinance or order; or (ii) which is defined as a hazardous waste, hazardous substance, pollutant or contaminant, including petroleum, under any statute, regulation, rule, or ordinance of any governmental authority having jurisdiction; or (iii) which is toxic, explosive, corrosive, flammable, infectious, radioactive, carcinogenic, mutagenic, or otherwise hazardous and is regulated by any governmental authority having jurisdiction.

Intervening Event means a material event, change or development with respect to the Company and the Company Subsidiaries taken as a whole arising after the date of this Agreement, which is (i) unknown to, nor reasonably foreseeable by, the Company Board as of or prior to the date of this Agreement and (ii) becomes known to or by the Company Board prior to the Effective Time; provided, however, that in no event will the receipt of a Takeover Proposal or Superior Proposal constitute an Intervening Event.

Laws means any domestic or foreign laws, common law, statutes, ordinances, rules, regulations, codes, Orders or legally enforceable requirements enacted, issued, adopted, promulgated, enforced, ordered or applied by any Governmental Entity.

Liability means any liability, indebtedness or obligation of any kind (whether accrued, absolute, contingent, matured, unmatured or otherwise, and whether or not required to be recorded or reflected on a balance sheet under GAAP).

Liens means, with respect to any property or asset, all pledges, liens, mortgages, charges, encumbrances, hypothecations, options, rights of first refusal, rights of first offer and security interests of any kind or nature whatsoever.

Order means any order, writ, assessment, decision, injunction, decree, ruling or judgment of a Governmental Entity whether temporary, preliminary or permanent.

Parent Material Adverse Effect means any change, event, development, condition, occurrence or effect that prevents or materially delays, or would reasonably be expected to prevent, consummation of the Merger or performance by Parent or Merger Sub of any of their material obligations under this Agreement.

Release means a release or threatened release of a hazardous substance (as defined in the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended (CERCLA)) or any other release, emission, disposal, or discharge into the environment, including to air, surface water, groundwater, land, surface and subsurface strata, or any use, storage, transport or handling of Hazardous Material on, under, about, or from the Company Real Property other than as expressly authorized by applicable law, ordinance, permit, order or other authorization.

Table of Contents

Superior Proposal means a bona fide written Takeover Proposal (with all references to 20% in the definition of Takeover Proposal being deemed to be references to 50%), which Takeover Proposal did not result from a breach of Section 5.3 that the Company Board determines in its good faith business judgment (after consultation with the Company s outside legal counsel and financial advisors) to be more favorable to the holders of Shares from a financial point of view than the Merger and other transactions contemplated by this Agreement, taking into account (a) all financial considerations, (b) the identity of the third party making such Takeover Proposal, (c) the anticipated timing, conditions (including any financing condition or the reliability of any debt or equity funding commitments) and prospects for completion of such Takeover Proposal, (d) the other terms and conditions of such Takeover Proposal and the implications thereof on the Company, including relevant legal, regulatory and other aspects of such Takeover Proposal deemed relevant by the Company Board and (e) any revisions to the terms of this Agreement and the Merger proposed by Parent pursuant to Section 5.3.

Takeover Proposal means a proposal or offer from, or indication of interest in making a proposal or offer by, any person (other than Parent or Merger Sub) relating to any (a) direct or indirect acquisition of assets of the Company (excluding sales of assets in the ordinary course of business) equal to twenty percent (20%) or more of the fair market value of the Company s consolidated assets or to which twenty percent (20%) or more of the Company s net revenues or net income on a consolidated basis are attributable, (b) direct or indirect acquisition of twenty percent (20%) or more of the voting equity interests of the Company, (c) tender offer or exchange offer that if consummated would result in any person beneficially owning (within the meaning of Section 13(d) of the Exchange Act) twenty percent (20%) or more of the voting equity interests of the Company, (d) merger, consolidation, other business combination or similar transaction involving the Company or any of its subsidiaries, pursuant to which such person would own twenty percent (20%) or more of the consolidated assets, net revenues or net income of the Company, taken as a whole, or (e) liquidation or dissolution (or the adoption of a plan of liquidation or dissolution) of the Company or the declaration or payment of an extraordinary dividend (whether in cash or other property) by the Company.

.

(The rest of this page left intentionally blank.)

Table of Contents

IN WITNESS WHEREOF, the parties have executed this Agreement and caused the same to be duly delivered on their behalf on the day and year first written above.

CORGENIX MEDICAL CORPORATION

By: /s/ Douglass T. Simpson
Name: Douglass T. Simpson

Title: President and Chief Executive Officer

CENTENNIAL MEDICAL HOLDINGS, INC.

By: /s/ Robert Womsley
Name: Robert Womsley
Title: Vice President

CENTENNIAL INTEGRATED, INC.

By: /s/ Robert Womsley
Name: Robert Womsley
Title: President

[Signature Page to Agreement and Plan of Merger]

7D 1	1			c.	\sim			
Tal	٦I	\boldsymbol{e}	\cap 1	1		۱n	tei	ntc

$\mathbf{E}\mathbf{X}\mathbf{E}$	CU	TION	I VERSION

ANNEX B

VOTING AGREEMENT

This VOTING AGREEMENT (this *Agreement*) is entered into as of August 27, 2014 by and among Centennial Medical Holdings, Inc. (*Parent*), and the shareholders of Corgenix Medical Corporation, a Nevada corporation (the *Company*) listed on Schedule A hereto (each, a *Shareholder* and collectively, the *Shareholders*). Capitalized terms used but not defined herein shall have the meanings ascribed to such terms in the Merger Agreement (as defined below).

WHEREAS, Parent, Merger Sub and the Company have, concurrently with the execution of this Agreement, entered into an Agreement and Plan of Merger, dated as of the date hereof (as may be amended, supplemented or otherwise modified, the *Merger Agreement*), which provides, among other things, for the merger of Merger Sub with and into the Company, with the Company continuing as the surviving corporation and a wholly-owned subsidiary of Parent (the *Merger*), upon the terms and subject to the conditions set forth in the Merger Agreement;

WHEREAS, the Shareholders and Parent are executing this Agreement concurrently with the execution of the Merger Agreement;

WHEREAS, as of the date hereof, the Shareholders are the beneficial owners (as defined under Rule 13d-3 of the Exchange Act) of (i) certain Shares, (ii) certain Company Stock Options to acquire Shares, and/or (iii) certain Warrants, as set forth opposite each Shareholder s name on Schedule A hereto (such Shares, Company Stock Options and Warrants, together with any other Shares acquired (whether beneficially or of record) by the Shareholders after the date hereof and prior to the earlier of the Effective Time and the termination of all of the Shareholders obligations under this Agreement, including any Shares acquired by means of purchase, dividend or distribution, or issued upon the exercise of any Company Stock Options or Warrants or the conversion of any convertible securities or otherwise, being collectively referred to herein as the Securities shall only include the securities covered by clause (i), (ii) or (iii) that are entitled to be voted, or for which the Shareholders is entitled to consent or act, with respect thereto (which shall not include Securities that are subject to issuance upon the exercise of any Company Stock Option or Warrant);

WHEREAS, receipt of the Requisite Company Vote is a condition to the consummation of the Merger; and

WHEREAS, as a condition to the willingness of Parent to enter into the Merger Agreement and as an inducement and in consideration therefor, each Shareholder has agreed to enter into this Agreement.

NOW, THEREFORE, in consideration of the foregoing, the mutual covenants and agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Shareholders, severally and not jointly, and Parent agree as follows:

ARTICLE I

VOTING; GRANT AND APPOINTMENT OF PROXY

Section 1.1 *Voting*. From and after the date hereof until the earlier of (a) the Effective Time, (b) the termination of the Merger Agreement pursuant to and in compliance with the terms therein, or (c)

Table of Contents

the date the Company Board effects a Change of Board Recommendation pursuant to Section 5.03 of the Merger Agreement (such earlier time,
the Expiration Time), each Shareholder irrevocably and unconditionally hereby agrees that at any meeting (whether annual or special and each
adjourned or postponed meeting) of the Company s shareholders, however called, or in connection with any written resolution of the Company s
shareholders, such Shareholder shall (i) appear at such meeting or otherwise cause its Securities to be counted as present thereat for purposes of
determining whether a quorum is present and (ii) vote or cause to be voted (including by proxy or written resolution, if applicable) all of such
Shareholder s Securities,

- (a) for approval of the Merger Agreement and the transactions contemplated by the Merger Agreement,
- (b) except to the extent set forth in this Agreement, against any other action, agreement or transaction that is intended, that could reasonably be expected, or the effect of which could reasonably be expected, to materially impede, interfere with, delay, postpone, discourage or adversely affect the Merger or any of the other transactions contemplated by the Merger Agreement or this Agreement or the performance by such Shareholder of its, his or her obligations under this Agreement, including, without limitation: (i) any extraordinary corporate transaction, such as a scheme of arrangement, merger, consolidation or other business combination involving the Company or any Company Subsidiaries (other than the Merger); (ii) a sale, lease or transfer of a material amount of assets of the Company or any Company Subsidiary or a reorganization, recapitalization or liquidation of the Company or any Company Subsidiary; (iii) an election of new members to the board of directors of the Company, other than nominees to the board of directors of the Company who are serving as directors of the Company on the date of this Agreement or as otherwise provided in the Merger Agreement; (iv) any material change in the present capitalization or dividend policy of the Company or any amendment or other change to the Company s articles of incorporation or bylaws, except if approved in writing by Parent; (v) any other action that would require the consent of Parent pursuant to Section 5.1 of the Merger Agreement, except if approved in writing by Parent; or (vi) any other material change in the Company s corporate structure or business, except if approved in writing by Parent,
- (c) against any action, proposal, transaction or agreement that would reasonably be expected to result in a breach in any respect of any covenant, representation or warranty or any other obligation or agreement of the Company contained in the Merger Agreement, or of such Shareholder contained in this Agreement,
- (d) in favor of any adjournment or postponement of the Company Shareholders Meeting as may be requested by Parent, and
- (e) in favor of any other matter necessary to the consummation of the transactions contemplated by the Merger Agreement or otherwise reasonably requested by Parent in order to consummate the transactions contemplated by the Merger Agreement.

Section 1.2 Grant of Irrevocable Proxy; Appointment of Proxy.

(a) Each Shareholder hereby irrevocably appoints Parent and any designee thereof as its proxy and attorney-in-fact (with full power of substitution), to vote or cause to be voted (including by proxy or written resolution, if applicable) the Securities in accordance with *Section 1.1* at any annual or special meeting of the Shareholders of the Company, however called, including any

Table of Contents

adjournment or postponement thereof, at which any of the matters described in *Section 1.1* is to be considered. Each Shareholder represents that all proxies, powers of attorney, instructions or other requests given by such Shareholder prior to the execution of this Agreement in respect of the voting of such Shareholder s Securities, if any, are not irrevocable and each Shareholder hereby revokes (or causes to be revoked) any and all previous proxies, powers of attorney, instructions or other requests with respect to such Shareholder s Securities. Each Shareholder shall take such further action or execute such other instruments as may be necessary to effectuate the intent of this proxy.

(b) Each Shareholder affirms that the irrevocable proxy set forth in this *Section 1.2* is given in connection with the execution of the Merger Agreement, and that such irrevocable proxy is given to secure the performance of the duties of such Shareholder under this Agreement. Each Shareholder further affirms that the irrevocable proxy is coupled with an interest and, except as set forth in this *Section 1.2*, is intended to be irrevocable prior to the Expiration Time. If for any reason the proxy granted herein is not irrevocable, then each Shareholder agrees to vote such Shareholder a Securities in accordance with *Section 1.1* above as instructed by Parent in writing prior to the Expiration Time. The parties agree that the foregoing is a voting agreement.

Section 1.3 Restrictions on Transfers. Each Shareholder hereby agrees that, from the date hereof until the Expiration Time, such Shareholder shall not, directly or indirectly, (a) sell (constructively or otherwise), transfer, assign, tender in any tender or exchange offer, pledge, grant, encumber, hypothecate or similarly dispose of (by merger, testamentary disposition, operation of law or otherwise) (collectively, <u>Transfer</u>), either voluntarily or involuntarily, or enter into any contract, option or other arrangement or understanding with respect to the Transfer of any Securities, including, without limitation, any swap transaction, option, warrant, forward purchase or sale transaction, futures transaction, cap transaction, floor transaction, collar transaction or any other similar transaction (including any option with respect to any such transaction) or combination of any such transactions, in each case involving any Securities and (x) has, or would reasonably be expected to have, the effect of reducing or limiting such Shareholder s economic interest in such Securities and/or (y) grants a third party the right to vote or direct the voting of such Securities (any such transaction, a <u>Derivative Transaction</u>), (b) deposit any Securities into a voting trust or enter into a voting agreement or arrangement or grant any proxy or power of attorney with respect thereto that is inconsistent with this Agreement, (c) convert or exchange, or take any action which would result in the conversion or exchange, of any Securities, (d) knowingly take any action that would make any representation or warranty of such Shareholder set forth in this Agreement untrue or incorrect or have the effect of preventing, disabling, or delaying such Shareholder from performing any of his, her, or its obligations under this Agreement, or (e) agree (whether or not in writing) to take any of the actions referred to in the foregoing clauses (a), (b) (c) or (d). Notwithstanding anything to the contrary herein, nothing shall require (and no Shareholder undertakes an obligation and makes a representation or warranty related to) the conversion, exercise or exchange of any security for which Shareholder has beneficial ownership into securities entitled to be voted, or for which Shareholder is entitled to consent or act, with respect thereto, including without limitation, any Company Stock Option or Warrant.

ARTICLE II

NO SOLICITATION

Section 2.1 Restricted Activities. Prior to the Expiration Time, each Shareholder in its capacity as a shareholder of the Company, shall not, directly or indirectly, (i) initiate, solicit, initiate or

Table of Contents

knowingly take any action to facilitate or knowingly encourage the submission of any Takeover Proposal or the making of any proposal that could reasonably be expected to lead to any Takeover Proposal or (ii) enter into, participate or engage in, or maintain or continue any discussion or negotiations relating to, any Takeover Proposal with any person other than Parent or Merger Sub, (iii) furnish to any person other than Parent or Merger Sub any non-public information that the Shareholder believes would be used for the purposes of formulating or furthering any Takeover Proposal, (iv) amend or grant any waiver or release under any standstill or similar agreement with respect to any class of equity securities of the Company, or (v) enter into any agreement in principle, letter of intent, term sheet, acquisition agreement, merger agreement, option agreement, joint venture agreement, partnership agreement or other Contract relating to any Takeover Proposal. The Shareholder will cease immediately and cause to be terminated any and all existing activities, discussions or negotiations, if any, with any third party conducted prior to the date hereof with respect to any Takeover Proposal.

Section 2.2 Notification. Each Shareholder, in its capacity as a shareholder of the Company, shall and shall cause such Shareholder s
Representatives to, immediately cease and cause to be terminated any discussions or negotiations with any parties that may have been conducted
heretofore with respect to a Takeover Proposal. From and after the date hereof until the Expiration Time, each Shareholder shall promptly advise
Parent in writing of (i) any Competing Transaction, (ii) any request it receives in its capacity as a shareholder of the Company for non-public
information relating to the Company or any Subsidiary, and (iii) any inquiry or request for discussion or negotiation it receives in its capacity as
a shareholder of the Company regarding a Competing Transaction, including in each case any Takeover Proposal, including the identity of the
person making any such Competing Transaction or inquiry and the terms of any such Competing Transaction or indication or
inquiry (including, if applicable, copies of any written requests, proposals or offers, including proposed agreements). Each Shareholder, in its
capacity as a shareholder of the Company, shall keep Parent reasonably informed on a reasonably current basis of the status and terms (including
any material changes to the terms thereof) of any such Competing Transaction or indication or inquiry (including, if applicable, any revised
copies of written requests, proposals and offers) and the status of any such discussions or negotiations to the extent known by such Shareholder.
This Article II shall not apply to any Takeover Proposal received by the Company. For the avoidance of doubt, nothing in this Article II will
limit or affect any actions or omissions taken by Shareholder in Shareholder s capacity as a director or officer of the Company, including in
exercising rights under the Merger Agreement (including, without limitation, under Section 5.3 thereof), and no such actions or omissions shall
be deemed a breach of this Article II.

ARTICLE III

REPRESENTATIONS, WARRANTIES AND COVENANTS OF THE SHAREHOLDERS

Section 3.1 Representations and Warranties. Each Shareholder represents and warrants to Parent as follows: (a) such Shareholder has full legal right, power, capacity and authority to execute and deliver this Agreement, to perform such Shareholder s obligations hereunder and to consummate the transactions contemplated hereby; (b) this Agreement has been duly executed and delivered by such Shareholder and the execution, delivery and performance of this Agreement by such Shareholder and the consummation of the transactions contemplated hereby have been duly authorized by all necessary action on the part of such Shareholder and no other actions or proceedings on the part of such Shareholder are necessary to authorize this Agreement or to consummate the transactions contemplated hereby, (c) assuming this Agreement constitutes the valid

Table of Contents

and binding agreement of Parent, this Agreement constitutes the valid and binding agreement of such Shareholder, enforceable against such Shareholder in accordance with its terms, (d) the execution and delivery of this Agreement by such Shareholder does not, and the consummation of the transactions contemplated hereby and the compliance with the provisions hereof will not, conflict with or violate any law or agreement binding upon such Shareholder or such Shareholder s Securities, nor require any authorization, consent or approval of, or filing with, any Governmental Entity, except for filings with the Securities and Exchange Commission by such Shareholder, (e) except for such transfer restrictions of general applicability as may be provided under the Securities Act and the blue sky laws of the various states of the United States, such Shareholder owns, beneficially and of record, or controls all of its Securities, and all of such Securities are free and clear of any proxy, voting restriction, adverse claim or other Lien (other than any restrictions created by this Agreement), and has sole or shared (together with affiliates controlled by such Shareholder) voting power and power of disposition with respect to such Securities, with no restrictions on such Shareholder s rights of voting or disposition pertaining thereto, and no person other than such Shareholder has any right to direct or approve the voting or disposition of any of such Shareholder s Securities, and (f) such Shareholder has not Transferred any Securities pursuant to any Derivative Transaction. Each Shareholder understands and acknowledges that Parent is entering into the Merger Agreement in reliance upon such Shareholder s execution, delivery and performance of this Agreement.

- (a) agrees, prior to the Expiration Time, not to take any action that would make any representation or warranty of such Shareholder contained herein untrue or incorrect or have or could have the effect of preventing, impeding or interfering with or adversely affecting the performance by such Shareholder of its obligations under this Agreement;
- (b) irrevocably waives, and agrees not to exercise, any rights of appraisal or rights of dissent from the Merger that such Shareholder may have with respect to such Shareholder s Securities (including without limitation any rights under Chapter 92A of the NRS) prior to the Expiration Time:
- (c) agrees to promptly notify Parent of the number of any new Securities acquired by the Shareholder after the date hereof and prior to the Expiration Time;
- (d) agrees to permit the Company to publish and disclose in the Proxy Statement, such Shareholder s identity and ownership of Shares or other equity securities of the Company and the nature of such Shareholder s commitments, arrangements and understandings under this Agreement; and
- (e) agrees that, upon request of Parent, such Shareholder shall execute and deliver any additional documents, consents or instruments and take such further actions as may reasonably be deemed by Parent to be necessary or desirable to carry out the provisions of this Agreement.

ARTICLE IV

Section 4.1 *Representations and Warranties of Parent*. Parent hereby represents and warrants to each Shareholder as follows: (a) this Agreement has been duly and validly authorized by Parent s board of directors, (b) this Agreement has been duly executed and delivered by a duly authorized

Table of Contents

officer or other representative of Parent, and (c) assuming this Agreement constitutes a valid and binding agreement of Shareholders, this Agreement constitutes a valid and binding agreement of Parent, enforceable against Parent in accordance with its terms, and (d) the execution and delivery of this Agreement by Parent does not, and the consummation of the transactions contemplated hereby and the compliance with the provisions hereof will not, conflict with or violate any law or agreement binding upon Parent, nor require any authorization, consent or approval of, or filing with, any Governmental Entity, except for filings with the Securities and Exchange Commission, as applicable.

ARTICLE V

TERMINATION

Section 5.1 *Termination*. This Agreement shall terminate and be of no further force or effect upon the earlier to occur of (a) the Closing, (b) the date of termination of the Merger Agreement in accordance with its terms and (c) the date the Company Board effects a Change of Board Recommendation pursuant to Section 5.03 of the Merger Agreement. Notwithstanding the preceding sentence, this *Article V* and *Article VI* shall survive any termination of this Agreement. Nothing in this *Article V* shall relieve or otherwise limit any party s liability for any breach of this Agreement prior to termination or any willful breach of this Agreement.

ARTICLE VI

MISCELLANEOUS

Section 6.1 *Notices*. All notices and other communications hereunder shall be in writing (in the English language) and shall be deemed duly given (i) upon receipt if delivered personally, or if by email or facsimile, upon confirmation of receipt by email or facsimile, (ii) one Business Day after being sent by express courier service, or (iii) three Business Days after being sent by registered or certified mail, return receipt requested. All notices hereunder shall be delivered to the address set forth on the signature pages hereto under each party s name, or pursuant to such other instructions as may be designated in writing by the party to receive such notice.

Section 6.2 *Directors and Officers*. Notwithstanding any provision of this Agreement to the contrary, nothing in this Agreement shall (or require the Shareholders to attempt to) (i) limit or restrict any Shareholder who is a director or officer of the Company from acting in such capacity or voting in such person s sole discretion on any matter or (ii) be construed to prohibit, limit or restrict Shareholder from exercising Shareholder s fiduciary duties as an officer or director to the Company or its shareholders (it being understood that this Agreement shall apply to each Shareholder solely in such Shareholder s capacity as a shareholder of the Company).

Section 6.3 *Severability*. Any term or provision of this Agreement which is invalid or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms and provisions of this Agreement in any other jurisdiction. If any provision of this Agreement is so broad as to be unenforceable, such provision shall be interpreted to be only as broad as is enforceable.

Section 6.4 *Entire Agreement*. This Agreement and the Merger Agreement embody the complete agreement and understanding among the parties hereto with respect to the subject matter hereof and thereof and supersede and preempt any prior understandings, agreements or

Table of Contents

representations by or among the parties, written or oral, which may have related to the subject matter hereof in any way.

Section 6.5 Specific Performance. Each party acknowledges that monetary damages would not be an adequate remedy in the event that any covenant or agreement in this Agreement is not performed in accordance with its terms, and it is therefore agreed that, in addition to and without limiting any other remedy or right it may have, each party will have the right to an injunction, temporary restraining order or other equitable relief in any court of competent jurisdiction enjoining any such breach and enforcing specifically the terms and provisions hereof. Each party hereto agrees not to oppose the granting of such relief in the event a court determines that such a breach has occurred, and to waive any requirement for the securing or posting of any bond in connection with such remedy. All rights, powers, and remedies provided under this Agreement or otherwise available in respect hereof at law or in equity shall be cumulative and not alternative, and the exercise or beginning of the exercise of any thereof by a party hereto shall not preclude the simultaneous or later exercise of any other such right, power or remedy by it.

Section 6.6 *Amendments; Waivers*. At any time prior to the Expiration Time, any provision of this Agreement may be amended or waived if, and only if, such amendment or waiver is in writing and signed, in the case of an amendment, by the Shareholders and Parent, or in the case of a waiver, by the party against whom the waiver is to be effective. Notwithstanding the foregoing, no failure or delay by a party hereto in exercising any right hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise of any other right hereunder.

Section 6.7 Governing Law. This Agreement shall be governed and construed in accordance with the laws of the State of Delaware, without regard to any applicable conflicts of law principles that would cause the application of the laws of any other jurisdiction.

Section 6.8 *Jurisdiction; Enforcement*. The parties agree that any Action brought by any party to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby shall be brought in any state or federal court sitting in the State of Delaware. Each of the parties submits to the jurisdiction of any such court in any Action seeking to enforce any provision of, or based on any matter arising out of, or in connection with, this Agreement or the transactions contemplated hereby, and hereby irrevocably waives the benefit of jurisdiction derived from present or future domicile or otherwise in such Action. Each party irrevocably waives, to the fullest extent permitted by law, any objection that it may now or hereafter have to the laying of the venue of any such proceeding in any such court or that any such proceeding brought in any such court has been brought in an inconvenient forum.

Section 6.9 WAIVER OF JURY TRIAL. EACH OF THE PARTIES TO THIS AGREEMENT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO TRIAL BY JURY IN ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY DOCUMENTS OR INSTRUMENTS REFERRED TO IN THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY, OR THE ACTIONS OF EACH OF THE PARTIES IN NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT OF THIS AGREEMENT.

Table of Contents

Section 6.10 *No Third Party Beneficiaries*. There are no third party beneficiaries of this Agreement and nothing in this Agreement, express or implied, is intended to confer on any person other than the parties hereto (and their respective successors, heirs and permitted assigns), any rights, remedies, obligations or liabilities.

Section 6.11 Assignment; Binding Effect. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto (whether by operation of law or otherwise) without the prior written consent of the other parties. Subject to the preceding sentence, this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and permitted assigns and, in the case of each Shareholder that is an individual, his or her estate, heirs, beneficiaries, personal representatives and executors.

Section 6.12 No Presumption Against Drafting Party. Each of the parties to this Agreement acknowledges that it has been represented by independent counsel in connection with this Agreement and the transactions contemplated by this Agreement. Accordingly, any rule of law or any legal decision that would require interpretation of any claimed ambiguities in this Agreement against the drafting party has no application and is expressly waived.

Section 6.13 *Counterparts*. This Agreement may be executed in two or more consecutive counterparts (including by facsimile or email pdf format), each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument, and shall become effective when one or more counterparts have been signed by each of the parties and delivered (by telecopy, email pdf format or otherwise) to the other parties.

[Signature Pages to follow]

B-8

Table of Contents

IN WITNESS WHEREOF, the parties hereto have duly executed and delivered this Voting Agreement as of the date and year first written above.

PARENT

Centennial Medical Holdings, Inc.

By: /s/ Robert Womsley
Name: Robert Womsley
Title: Vice President

[Signature Page to Voting Agreement]

B-9

Table of Contents

IN WITNESS WHEREOF, the parties hereto have duly executed and delivered this Voting Agreement as of the date and year first written above.

STOCKHOLDERS

/s/ Stephen P. Gouze Stephen P. Gouze

/s/ Robert Tutag Robert Tutag

/s/ Douglass Simpson Douglass Simpson

/s/ William H. Critchfield William H. Critchfield

/s/ Ann L. Steinbarger Ann L. Steinbarger

[Signature Page to Voting Agreement]

B-10

Table of Contents

Schedule A

Securities

As of August 25, 2014

		Percentage of			
Shareholder	Shares	Outstanding Shares	Company Options	Warrants	Total
			•	warrants	
Stephen P. Gouze	1,052,612	1.94%	240,000	0	1,292,612
Robert Tutag	961,000	1.77%	260,000	0	1,221,000
Douglass Simpson	435,838	0.80%	560,000	0	995,838
William H. Critchfield	366,545	0.68%	450,000	0	816,545
Ann L. Steinbarger	182,855	0.34%	280,000	0	462,855

Table	of	Contents

Α	N.	N	FX	(

INVERNESS ADVISORS

FAIRNESS OPINION

August	26,	20	14

Board of Directors

Corgenix Medical Corporation

11575 Main Street, Number 400

Broomfield, CO 80020

Members of the Board of Directors:

We understand that Corgenix Medical Corporation, a Nevada corporation (the Company), Centennial Medical Holdings, Inc., a Delaware corporation (Parent), and a wholly-owned subsidiary of Parent, Centennial Integrated, Inc., a Nevada corporation (Merger Sub), plan to enter into an Agreement and Plan of Merger (the Merger Agreement) that provides for, among other things, the merger of Merger Sub with and into the Company, as a result of which the Company would become a wholly owned subsidiary of Parent (the Transaction). Pursuant to the Transaction, each issued and outstanding share of common stock, par value \$0.001 per share, of the Company (the Company Common Stock) (other than shares of Company Common Stock to be cancelled in accordance with the Merger Agreement and other than Dissenting Shares (as defined in the Merger Agreement) will be converted into the right to receive an amount in cash equal to \$0.27, payable to the holder in cash, without interest (the Consideration). The terms and conditions of the Transaction are more fully set forth in the Merger Agreement.

You have asked for our opinion as to whether, as of the date hereof, the Consideration to be received by the holders of Company Common Stock pursuant to the Merger Agreement is fair from a financial point of view, to the holders of the Company Common Stock. For purposes of the opinion set forth herein, we have, among other things:

- 1. reviewed a draft of the Merger Agreement, dated August 25, 2014 (the Draft Agreement), including the financial terms and conditions set forth therein:
- 2. reviewed the Company s audited financial statements for the fiscal year ended June 30, 2013, the Company s unaudited financial statements for the nine months ended March 31, 2014 and a preliminary draft of the Company s audited financial statements for the fiscal year

ended June 30, 2014;

INVERNESS ADVISORS 601 Montgomery St, Suite 1210 San Francisco, CA 94111

All securities business conducted through KEMA Partners LLC a registered broker-dealer

C-1

Table of Contents

3. reviewed certain other business, operating and financial data of the Company, prepared and furnished to us by the Company s management, including certain financial forecasts, projections and analyses for the Company prepared and furnished to us by the Company s management for the fiscal years ending June 30, 2014 through 2018 (the Forecasts);
4. held discussions with the senior management team of the Company concerning the business, past and current operations, financial condition and future prospects of the Company, and certain other matters we believed necessary or appropriate to our inquiry;
5. compared the financial performance of the Company and trading activity of the Company Common Stock with that of certain other publicly-traded companies that we deemed relevant;
6. compared the financial terms of the Transaction with the financial terms, to the extent publicly available, of other transactions that we deemed relevant;
7. reviewed historical stock prices and trading volumes of the Company Common Stock;
8. reviewed the Company s draft annual report on Form 10-K for the fiscal year ended June 30, 2014, the Company s annual report of Form 10-K for the fiscal year ended June 30, 2013, and the Company s quarterly report on Form 10-Q for the fiscal quarter ended March 31, 2014;
9. reviewed certain other publicly available business, operating and financial information of the Company; and
10. made such other studies and inquiries, reviewed such other data, and considered such other factors as we have deemed, in our sole judgment, to be necessary, appropriate or relevant to render the opinion set forth herein.
In preparing our opinion, we have, with your consent, assumed and relied upon the accuracy and completeness of all financial and other information supplied or otherwise made available to us by the Company and all publicly-available financial and other information regarding the Company reviewed by us, and have not independently verified any such information or assumed any responsibility or liability therefor. With regard to all of the foregoing information, we have relied upon the assurances of the senior management team of the Company that all such information is complete and accurate in all material respects and that they are unaware of any facts or circumstances that would make such

information incomplete or misleading in any material respect. We have not been requested to conduct and have not conducted a physical inspection of the properties or facilities of the Company, nor have we conducted any valuation or appraisal of any assets of the Company, nor have any such valuations or appraisals been provided to us. We have not evaluated the solvency of the Company under any state, federal or

other laws relating to bankruptcy, insolvency or similar matters, and have undertaken no

Table of Contents

independent analysis of any pending or threatened litigation, regulatory action, possible unasserted claims or other contingent liabilities, to which the Company is a party or may be subject. At the direction of the Company and with its consent, our opinion makes no assumption concerning, and therefore does not consider, the possible assertion of claims, outcomes or damages arising out of any such matters.

With respect to the Forecasts provided to us by the Company, we have, with your consent, assumed that such Forecasts were prepared in good faith on reasonable bases reflecting management s current best estimates and judgments of the Company s future financial performance were it not to be sold to Parent. We have also assumed, with your consent, that the financial results reflected in such Forecasts would be realized in the amounts and at the times projected, and we assume no responsibility for and express no view as to such Forecasts or the assumptions on which they are based. Further, without limiting the foregoing, we have, with your consent, assumed, without independent verification, that the historical and projected financial information provided to us by the Company accurately reflects the historical and projected operations of the Company, and that there has been no material change in the assets, financial condition, business or prospects of the Company since the respective dates of the most recent financial statements made available to us.

Our opinion is based on market, economic, financial and other circumstances and conditions as they exist as of the date of this letter. Our opinion can be evaluated only as of the date of this letter and any material change in such circumstances and conditions would require a reevaluation of this opinion, which we are under no obligation to undertake. We assume no responsibility to update or revise our opinion based upon events or circumstances occurring after the date hereof.

This letter does not constitute a recommendation to the Board of Directors of the Company or any other person with respect to the Transaction, and does not address the relative merits of the Transaction over any other alternative transactions which may be available to the Company. We express no opinion as to the underlying business decision of the Company to effect the Transaction, the structure, or accounting treatment or taxation consequences of the Transaction or the availability or advisability of any alternatives to the Transaction. We express no opinion with respect to any other reasons, legal, business, or otherwise, that may support the decision of the Board of Directors of the Company to approve or cause the Company to enter into the Merger Agreement or consummate the Transaction. No opinion is expressed with respect to the fairness (financial or otherwise) of the amount, nature or any other aspect of any compensation payable to any of the officers, directors or employees of the Company, or class of such persons, whether independently or relative to the Consideration, including whether such compensation is reasonable in the context of the Transaction, and we also express no opinion as to the price at which the Company Common Stock will trade upon announcement of the Transaction or at any future time. We have not made any independent investigation of any legal, accounting or tax matters affecting the Company, and we have assumed the correctness of all legal, accounting and tax advice given to the Company and its Board of Directors.

Table of Contents

We also have assumed, with your consent, that in the course of obtaining necessary regulatory and third party approvals and consents for the Transaction, no modification, delay, limitation, restriction or condition will be imposed that will have an adverse effect on the Company or the contemplated benefits of the Transaction and that the Transaction will be consummated in accordance with the terms of the Merger Agreement, without waiver, modification or amendment of any material term, condition or agreement therein. We also have assumed that all of the representations and warranties of the parties set forth in the Merger Agreement and all related agreements and documents are true and correct, that each party to the Merger Agreement and such other agreements will fully and timely perform all of the covenants and agreements required to be performed by such party, and that the final form of the Merger Agreement reviewed by us will not vary in any regard that is material to our analysis from the Draft Agreement, and that the Transaction will be consummated in accordance with the terms thereof, including that, in all respects material to our analysis, the representations and warranties made by the parties thereto are accurate and complete.

It is understood that this letter is solely for the information of the Company s Board of Directors in evaluating the Transaction and does not confer rights or remedies upon the stockholders of the Company, any creditor of the Company or any other person, and we express no opinion or recommendation as to how any stockholder of the Company should vote or act in connection with the Transaction. Furthermore, this letter should not be construed as creating any fiduciary duty on our part to the Company, the Company s Board of Directors or any other party. This opinion is not to be reproduced, summarized, described or referred to or given to any other person or otherwise made public or used for any other purpose, or published or referred to at any time, in whole or in part, without our prior written consent, except that this opinion may be included in its entirety, if legally required, in any proxy statement filed by the Company with the Securities and Exchange Commission. The issuance of this opinion was approved by our Fairness Opinion Review Committee.

Inverness Advisors, a division of KEMA Partners LLC (Inverness), as part of its investment banking services, is regularly engaged in the valuation of businesses and securities in connection with mergers, acquisitions, private placements and valuations for corporate and other purposes. Inverness and its affiliates, in the ordinary course of business, provide and in the future may continue to provide investment banking services to the Company and may receive fees for the rendering of such services. In addition, in the ordinary course of our businesses, we and our affiliates may actively trade the debt and equity securities of the Company for our own account or for the accounts of customers and, accordingly, we may at any time hold long or short positions in such securities.

Inverness has been engaged by the Company as its financial advisor pursuant to the engagement and indemnity agreement dated January 10, 2014, by and between Inverness and the Company (the Inverness Engagement Letter). In connection with the Transaction, Inverness will receive a fee for the rendering of this opinion and certain additional fees for our services in connection with the Transaction, a significant portion of which is contingent upon the consummation of the Transaction. In addition, the Company has agreed to reimburse our expenses and to indemnify us against and exculpate us from

Table of Contents

certain liabilities that may arise out of our engagement, all as more fully described in the Inverness Engagement Letter.

Based on and subject to the foregoing, and in reliance thereon, we are of the opinion that, as of the date hereof, the Consideration to be received by the holders of Company Common Stock pursuant to the Merger Agreement is fair, from a financial point of view, to such holders.

Very truly yours,

/s/ Inverness Advisors
INVERNESS ADVISORS

C-5

m	. 1		c			
Tal	hl	e	ot	on	itei	nts

ANNEX D

DISSENTER S RIGHTS STATUTE

RIGHTS OF DISSENTING OWNERS

NRS 92A.300 Definitions. As used in NRS 92A.300 to 92A.500, inclusive, unless the context otherwise requires, the words and terms defined in NRS 92A.305 to 92A.335, inclusive, have the meanings ascribed to them in those sections.

NRS 92A.305 Beneficial stockholder definedBeneficial stockholder means a person who is a beneficial owner of shares held in a voting trust or by a nominee as the stockholder of record.

NRS 92A.310 Corporate action definedCorporate action means the action of a domestic corporation.

NRS 92A.315 Dissenter definedDissenter means a stockholder who is entitled to dissent from a domestic corporation s action under NRS 92A.380 and who exercises that right when and in the manner required by NRS 92A.400 to 92A.480, inclusive.

NRS 92A.320 Fair value definedFair value, with respect to a dissenter s shares, means the value of the shares determined:

- 1. Immediately before the effectuation of the corporate action to which the dissenter objects, excluding any appreciation or depreciation in anticipation of the corporate action unless exclusion would be inequitable;
- 2. Using customary and current valuation concepts and techniques generally employed for similar businesses in the context of the transaction requiring appraisal; and
- 3. Without discounting for lack of marketability or minority status.

NRS 92A.325 Stockholder definedStockholder means a stockholder of record or a beneficial stockholder of a domestic corporation.

NRS 92A.330 Stockholder of record definedStockholder of record means the person in whose name shares are registered in the records of a domestic corporation or the beneficial owner of shares to the extent of the rights granted by a nominee s certificate on file with the domestic corporation.

NRS 92A.335 Subject corporation definedSubject corporation means the domestic corporation which is the issuer of the shares held by a dissenter before the corporate action creating the dissenter s rights becomes effective or the surviving or acquiring entity of that issuer after the corporate action becomes effective.

NRS 92A.340 Computation of interest. Interest payable pursuant to NRS 92A.300 to 92A.500, inclusive, must be computed from the effective date of the action until the date of payment, at the rate of interest most recently established pursuant to NRS 99.040.

NRS 92A.350 Rights of dissenting partner of domestic limited partnership. A partnership agreement of a domestic limited partnership or, unless otherwise provided in the partnership agreement, an agreement of merger or exchange, may provide that contractual rights with respect to the partnership interest of a dissenting general or limited partner of a domestic limited partnership are available for any class or group of partnership interests in connection with any merger or exchange in which the domestic limited partnership is a constituent entity.

NRS 92A.360 Rights of dissenting member of domestic limited-liability company. The articles of organization or operating agreement of a domestic limited-liability company or, unless otherwise provided in the articles of organization or operating agreement, an agreement of merger or exchange, may provide that contractual rights with respect to the interest of a dissenting member are available in connection with any merger or exchange in which the domestic limited-liability company is a constituent entity.

Table of Contents

NIDC 03 4 270	Dialida af		1	a f Jamaan4!a		4
NK5 92A.5/U	KIQNIS OI	aissenung	member e	oi aomesuc	nonprofit cor	porauon.

1. Except as otherwise provided in subsection 2, and unless otherwise provided in the articles or bylaws, any member of any constituent domestic nonprofit corporation who voted against the merger may, without prior notice, but within 30 days after the effective date of the merger, resign from membership and is thereby excused from all contractual obligations to the constituent or surviving corporations which did not occur before the member s resignation and is thereby entitled to those rights, if any, which would have existed if there had been no merger and the membership had been terminated or the member had been expelled.
2. Unless otherwise provided in its articles of incorporation or bylaws, no member of a domestic nonprofit corporation, including, but not limited to, a cooperative corporation, which supplies services described in chapter 704 of NRS to its members only, and no person who is a member of a domestic nonprofit corporation as a condition of or by reason of the ownership of an interest in real property, may resign and dissent pursuant to subsection 1.
NRS 92A.380 Right of stockholder to dissent from certain corporate actions and to obtain payment for shares.
1. Except as otherwise provided in NRS 92A.370 and 92A.390 and subject to the limitation in paragraph (f), any stockholder is entitled to dissent from, and obtain payment of the fair value of the stockholder s shares in the event of any of the following corporate actions:
(a) Consummation of a plan of merger to which the domestic corporation is a constituent entity:
(1) If approval by the stockholders is required for the merger by NRS 92A.120 to 92A.160, inclusive, or the articles of incorporation, regardless of whether the stockholder is entitled to vote on the plan of merger; or
(2) If the domestic corporation is a subsidiary and is merged with its parent pursuant to NRS 92A.180.
(b) Consummation of a plan of conversion to which the domestic corporation is a constituent entity as the corporation whose subject owner s interests will be converted.
(c) Consummation of a plan of exchange to which the domestic corporation is a constituent entity as the corporation whose subject owner s interests will be acquired, if the stockholder s shares are to be acquired in the plan of exchange.

(d) Any corporate action taken pursuant to a vote of the stockholders to the extent that the articles of incorporation, bylaws or a resolution of the board of directors provides that voting or nonvoting stockholders are entitled to dissent and obtain payment for their shares.
(e) Accordance of full voting rights to control shares, as defined in NRS 78.3784, only to the extent provided for pursuant to NRS 78.3793.
(f) Any corporate action not described in this subsection that will result in the stockholder receiving money or scrip instead of a fraction of a share except where the stockholder would not be entitled to receive such payment pursuant to NRS 78.205, 78.2055 or 78.207. A dissent pursuant to this paragraph applies only to the fraction of a share, and the stockholder is entitled only to obtain payment of the fair value of the fraction of a share.
2. A stockholder who is entitled to dissent and obtain payment pursuant to NRS 92A.300 to 92A.500, inclusive, may not challenge the corporate action creating the entitlement unless the action is unlawful or fraudulent with respect to the stockholder or the domestic corporation.
3. Subject to the limitations in this subsection, from and after the effective date of any corporate action described in subsection 1, no stockholder who has exercised the right to dissent pursuant to NRS 92A.300 to 92A.500, inclusive, is entitled to vote his or her shares for any purpose or to receive payment of dividends or any other distributions on shares. This subsection does not apply to dividends or other distributions payable to stockholders on a date before the effective date of any corporate action from which the stockholder has dissented. If a stockholder exercises the right to dissent with respect to a corporate action described in paragraph (f) of subsection 1, the restrictions of this subsection apply only to the shares to be converted into a fraction of a share and the dividends and distributions to those shares.
NRS 92A.390 Limitations on right of dissent: Stockholders of certain classes or series; action of stockholders not required for plan of merger.
1. There is no right of dissent with respect to a plan of merger, conversion or exchange in favor of stockholders of any class or series which is:
(a) A covered security under section 18(b)(1)(A) or (B) of the Securities Act of 1933, 15 U.S.C. § 77r(b)(1)(A) or (B), as amended;
D-2

Table of Contents

(b) Traded in an organized market and has at least 2,000 stockholders and a market value of at least \$20,000,000, exclusive of the value of such shares held by the corporation subsidiaries, senior executives, directors and beneficial stockholders owning more than 10 percent of such shares; or
(c) Issued by an open end management investment company registered with the Securities and Exchange Commission under the Investment Company Act of 1940, 15 U.S.C. §§ 80a-1 et seq., as amended, and which may be redeemed at the option of the holder at net asset value, unless the articles of incorporation of the corporation issuing the class or series or the resolution of the board of directors approving the plan of merger, conversion or exchange expressly provide otherwise.
2. The applicability of subsection 1 must be determined as of:
(a) The record date fixed to determine the stockholders entitled to receive notice of and to vote at the meeting of stockholders to act upon the corporate action requiring dissenter s rights; or
(b) The day before the effective date of such corporate action if there is no meeting of stockholders.
3. Subsection 1 is not applicable and dissenter s rights are available pursuant to NRS 92A.380 for the holders of any class or series of shares who are required by the terms of the corporate action requiring dissenter s rights to accept for such shares anything other than cash or shares of any class or any series of shares of any corporation, or any other proprietary interest of any other entity, that satisfies the standards set forth in subsection 1 at the time the corporate action becomes effective.
4. There is no right of dissent for any holders of stock of the surviving domestic corporation if the plan of merger does not require action of the stockholders of the surviving domestic corporation under NRS 92A.130.
5. There is no right of dissent for any holders of stock of the parent domestic corporation if the plan of merger does not require action of the stockholders of the parent domestic corporation under NRS 92A.180.
NRS 92A.400 Limitations on right of dissent: Assertion as to portions only to shares registered to stockholder; assertion by beneficial stockholder.
1. A stockholder of record may assert dissenter s rights as to fewer than all of the shares registered in his or her name only if the stockholder of

record dissents with respect to all shares of the class or series beneficially owned by any one person and notifies the subject corporation in writing of the name and address of each person on whose behalf the stockholder of record asserts dissenter s rights. The rights of a partial dissenter under this subsection are determined as if the shares as to which the partial dissenter dissents and his or her other shares were registered

in the names of different stockholders.

2. A beneficial stockholder may assert dissenter s rights as to shares held on his or her behalf only if the beneficial stockholder:
(a) Submits to the subject corporation the written consent of the stockholder of record to the dissent not later than the time the beneficial stockholder asserts dissenter s rights; and
(b) Does so with respect to all shares of which he or she is the beneficial stockholder or over which he or she has power to direct the vote.
NRS 92A.410 Notification of stockholders regarding right of dissent.
1. If a proposed corporate action creating dissenter s rights is submitted to a vote at a stockholders meeting, the notice of the meeting must state that stockholders are, are not or may be entitled to assert dissenter s rights under NRS 92A.300 to 92A.500, inclusive. If the domestic corporation concludes that dissenter s rights are or may be available, a copy of NRS 92A.300 to 92A.500, inclusive, must accompany the meeting notice sent to those record stockholders entitled to exercise dissenter s rights.
2. If the corporate action creating dissenter s rights is taken by written consent of the stockholders or without a vote of the stockholders, the domestic corporation shall notify in writing all stockholders entitled to assert dissenter s rights that the action was taken and send them the dissenter s notice described in NRS 92A.430.
NRS 92A.420 Prerequisites to demand for payment for shares.
1. If a proposed corporate action creating dissenter s rights is submitted to a vote at a stockholders meeting, a stockholder who wishes to assert dissenter s rights with respect to any class or series of shares:
(a) Must deliver to the subject corporation, before the vote is taken, written notice of the stockholder s intent to demand payment for his or her shares if the proposed action is effectuated; and
(b) Must not vote, or cause or permit to be voted, any of his or her shares of such class or series in favor of the proposed action.
D-3

Table of Contents

2. If a proposed corporate action creating dissenter s rights is taken by written consent of the stockholders, a stockholder who wishes to assert dissenter s rights with respect to any class or series of shares must not consent to or approve the proposed corporate action with respect to such class or series.
3. A stockholder who does not satisfy the requirements of subsection 1 or 2 and NRS 92A.400 is not entitled to payment for his or her shares under this chapter.
NRS 92A.430 Dissenter s notice: Delivery to stockholders entitled to assert rights; contents.
1. The subject corporation shall deliver a written dissenter s notice to all stockholders of record entitled to assert dissenter s rights in whole or in part, and any beneficial stockholder who has previously asserted dissenter s rights pursuant to NRS 92A.400.
2. The dissenter s notice must be sent no later than 10 days after the effective date of the corporate action specified in NRS 92A.380, and must:
(a) State where the demand for payment must be sent and where and when certificates, if any, for shares must be deposited;
(b) Inform the holders of shares not represented by certificates to what extent the transfer of the shares will be restricted after the demand for payment is received;
(c) Supply a form for demanding payment that includes the date of the first announcement to the news media or to the stockholders of the terms of the proposed action and requires that the person asserting dissenter s rights certify whether or not the person acquired beneficial ownership of the shares before that date;
(d) Set a date by which the subject corporation must receive the demand for payment, which may not be less than 30 nor more than 60 days after the date the notice is delivered and state that the stockholder shall be deemed to have waived the right to demand payment with respect to the shares unless the form is received by the subject corporation by such specified date; and
(e) Be accompanied by a copy of NRS 92A.300 to 92A.500, inclusive.
NRS 92A.440 Demand for payment and deposit of certificates; loss of rights of stockholder; withdrawal from appraisal process.

1. A stockholder who receives a dissenter s notice pursuant to NRS 92A.430 and who wishes to exercise dissenter s rights must:
(a) Demand payment;
(b) Certify whether the stockholder or the beneficial owner on whose behalf he or she is dissenting, as the case may be, acquired beneficial ownership of the shares before the date required to be set forth in the dissenter s notice for this certification; and
(c) Deposit the stockholder s certificates, if any, in accordance with the terms of the notice.
2. If a stockholder fails to make the certification required by paragraph (b) of subsection 1, the subject corporation may elect to treat the stockholder s shares as after-acquired shares under NRS 92A.470.
3. Once a stockholder deposits that stockholder s certificates or, in the case of uncertified shares makes demand for payment, that stockholder loses all rights as a stockholder, unless the stockholder withdraws pursuant to subsection 4.
4. A stockholder who has complied with subsection 1 may nevertheless decline to exercise dissenter s rights and withdraw from the appraisal process by so notifying the subject corporation in writing by the date set forth in the dissenter s notice pursuant to NRS 92A.430. A stockholder who fails to so withdraw from the appraisal process may not thereafter withdraw without the subject corporation s written consent.
5. The stockholder who does not demand payment or deposit his or her certificates where required, each by the date set forth in the dissenter s notice, is not entitled to payment for his or her shares under this chapter.
NRS 92A.450 Uncertificated shares: Authority to restrict transfer after demand for payment. The subject corporation may restrict the transfer of shares not represented by a certificate from the date the demand for their payment is received.
NRS 92A.460 Payment for shares: General requirements.
1. Except as otherwise provided in NRS 92A.470, within 30 days after receipt of a demand for payment pursuant to NRS 92A.440, the subject corporation shall pay in cash to each dissenter who complied with NRS 92A.440 the amount the subject corporation estimates to be the fair value of the dissenter s shares, plus accrued interest. The obligation of the subject corporation under this subsection may be enforced by the district court:
D-4

Table of Contents

(a) Of the county where the subject corporation s principal office is located;
(b) If the subject corporation s principal office is not located in this State, in the county in which the corporation s registered office is located; or
(c) At the election of any dissenter residing or having its principal or registered office in this State, of the county where the dissenter resides or has its principal or registered office.
Ê The court shall dispose of the complaint promptly.
2. The payment must be accompanied by:
(a) The subject corporation s balance sheet as of the end of a fiscal year ending not more than 16 months before the date of payment, a statement of income for that year, a statement of changes in the stockholders equity for that year or, where such financial statements are not reasonably available, then such reasonably equivalent financial information and the latest available quarterly financial statements, if any;
(b) A statement of the subject corporation s estimate of the fair value of the shares; and
(c) A statement of the dissenter s rights to demand payment under NRS 92A.480 and that if any such stockholder does not do so within the period specified, such stockholder shall be deemed to have accepted such payment in full satisfaction of the corporation s obligations under this chapter.
NRS 92A.470 Withholding payment for shares acquired on or after date of dissenter s notice: General requirements.
1. A subject corporation may elect to withhold payment from a dissenter unless the dissenter was the beneficial owner of the shares before the date set forth in the dissenter s notice as the first date of any announcement to the news media or to the stockholders of the terms of the proposed action.
2. To the extent the subject corporation elects to withhold payment, within 30 days after receipt of a demand for payment pursuant to NRS 92A.440, the subject corporation shall notify the dissenters described in subsection 1:
(a) Of the information required by paragraph (a) of subsection 2 of NRS 92A.460;

(b) Of the subject corporation s estimate of fair value pursuant to paragraph (b) of subsection 2 of NRS 92A.460;
(c) That they may accept the subject corporation s estimate of fair value, plus interest, in full satisfaction of their demands or demand appraisal under NRS 92A.480;
(d) That those stockholders who wish to accept such an offer must so notify the subject corporation of their acceptance of the offer within 30 days after receipt of such offer; and
(e) That those stockholders who do not satisfy the requirements for demanding appraisal under NRS 92A.480 shall be deemed to have accepted the subject corporation s offer.
3. Within 10 days after receiving the stockholder s acceptance pursuant to subsection 2, the subject corporation shall pay in cash the amount offered under paragraph (b) of subsection 2 to each stockholder who agreed to accept the subject corporation s offer in full satisfaction of the stockholder s demand.
4. Within 40 days after sending the notice described in subsection 2, the subject corporation shall pay in cash the amount offered under paragraph (b) of subsection 2 to each stockholder described in paragraph (e) of subsection 2.
NRS 92A.480 Dissenter s estimate of fair value: Notification of subject corporation; demand for payment of estimate.
1. A dissenter paid pursuant to NRS 92A.460 who is dissatisfied with the amount of the payment may notify the subject corporation in writing of the dissenter s own estimate of the fair value of his or her shares and the amount of interest due, and demand payment of such estimate, less any payment pursuant to NRS 92A.460. A dissenter offered payment pursuant to NRS 92A.470 who is dissatisfied with the offer may reject the offer pursuant to NRS 92A.470 and demand payment of the fair value of his or her shares and interest due.
2. A dissenter waives the right to demand payment pursuant to this section unless the dissenter notifies the subject corporation of his or her demand to be paid the dissenter s stated estimate of fair value plus interest under subsection 1 in writing within 30 days after receiving the subject corporation s payment or offer of payment under NRS 92A.460 or 92A.470 and is entitled only to the payment made or offered.
NRS 92A.490 Legal proceeding to determine fair value: Duties of subject corporation; powers of court; rights of dissenter.
1. If a demand for payment pursuant to NRS 92A.480 remains unsettled, the subject corporation shall commence a proceeding within 60 days after receiving the demand and petition the court to determine the fair value

of the shares and accrued interest. If the subject corporation does not commence the proceeding within the 60-day period, it shall pay each

Table of Contents

dissenter whose demand remains unsettled the amount demanded by each dissenter pursuant to NRS 92A.480 plus interest.
2. A subject corporation shall commence the proceeding in the district court of the county where its principal office is located in this State. If the principal office of the subject corporation is not located in this State, the right to dissent arose from a merger, conversion or exchange and the principal office of the surviving entity, resulting entity or the entity whose shares were acquired, whichever is applicable, is located in this State, it shall commence the proceeding in the county where the principal office of the surviving entity, resulting entity or the entity whose shares were acquired is located. In all other cases, if the principal office of the subject corporation is not located in this State, the subject corporation shall commence the proceeding in the district court in the county in which the corporation s registered office is located.
3. The subject corporation shall make all dissenters, whether or not residents of Nevada, whose demands remain unsettled, parties to the proceeding as in an action against their shares. All parties must be served with a copy of the petition. Nonresidents may be served by registered or certified mail or by publication as provided by law.
4. The jurisdiction of the court in which the proceeding is commenced under subsection 2 is plenary and exclusive. The court may appoint one of more persons as appraisers to receive evidence and recommend a decision on the question of fair value. The appraisers have the powers described in the order appointing them, or any amendment thereto. The dissenters are entitled to the same discovery rights as parties in other civil proceedings.
5. Each dissenter who is made a party to the proceeding is entitled to a judgment:

(a) For the amount, if any, by which the court finds the fair value of the dissenter s shares, plus interest, exceeds the amount paid by the subject corporation; or

(b) For the fair value, plus accrued interest, of the dissenter s after-acquired shares for which the subject corporation elected to withhold payment pursuant to NRS 92A.470.

NRS 92A.500 Assessment of costs and fees in certain legal proceedings.

1. The court in a proceeding to determine fair value shall determine all of the costs of the proceeding, including the reasonable compensation and expenses of any appraisers appointed by the court. The court shall assess the costs against the subject corporation, except that the court may assess costs against all or some of the dissenters, in amounts the court finds equitable, to the extent the court finds the dissenters acted arbitrarily, vexatiously or not in good faith in demanding payment.

2. The court may also assess the fees and expenses of the counsel and experts for the respective parties, in amounts the court finds equitable:
(a) Against the subject corporation and in favor of all dissenters if the court finds the subject corporation did not substantially comply with the requirements of NRS 92A.300 to 92A.500, inclusive; or
(b) Against either the subject corporation or a dissenter in favor of any other party, if the court finds that the party against whom the fees and expenses are assessed acted arbitrarily, vexatiously or not in good faith with respect to the rights provided by NRS 92A.300 to 92A.500, inclusive.
3. If the court finds that the services of counsel for any dissenter were of substantial benefit to other dissenters similarly situated, and that the fees for those services should not be assessed against the subject corporation, the court may award to those counsel reasonable fees to be paid out of the amounts awarded to the dissenters who were benefited.
4. In a proceeding commenced pursuant to NRS 92A.460, the court may assess the costs against the subject corporation, except that the court may assess costs against all or some of the dissenters who are parties to the proceeding, in amounts the court finds equitable, to the extent the court finds that such parties did not act in good faith in instituting the proceeding.
5. To the extent the subject corporation fails to make a required payment pursuant to NRS 92A.460, 92A.470 or 92A.480, the dissenter may bring a cause of action directly for the amount owed and, to the extent the dissenter prevails, is entitled to recover all expenses of the suit.
6. This section does not preclude any party in a proceeding commenced pursuant to NRS 92A.460 or 92A.490 from applying the provisions of N.R.C.P. 68 or NRS 17.115.
D-6